UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, DC 20549

FORM	10-Q
-------------	------

(Mark ⊠		ON 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF
	FOR THE QUARTERLY PERIOD ENDED MARCH 31, 2007	
	TRANSITION REPORT PURSUANT TO SECTION 1934	ON 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF
	FOR THE TRANSITION PERIOD FROM TO	
	Commission	file number: 000-50789
	Digirad	Corporation
		gistrant as specified in its charter)
	Delaware (State or Other Jurisdiction of Incorporation or Organization)	33-0145723 (I.R.S. Employer Identification No.)
	13950 Stowe Drive, Poway, CA (Address of Principal Executive Offices)	92064 (Zip Code)
	· ·	358) 726-1600 one Number, Including Area Code)
the pre		ired to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during required to file such reports), and (2) has been subject to such filing requirements for
Indicat Act.)	e by check mark whether registrant is a large accelerated filer, an acc	celerated filer, or a non-accelerated filer (as defined in Rule 12b-2 of the Exchange
	Large accelerated filer 🛚 Acc	celerated filer $oxtimes$ Non-accelerated filer $oxtimes$
Indicat	e by check mark whether registrant is a shell company (as defined in	Rule 12b-2 of the Exchange Act). Yes No
As of A	April 23, 2007, the registrant had 18,817,951 shares of Common Stoc	ck (\$0.0001 par value) outstanding.

DIGIRAD CORPORATION TABLE OF CONTENTS

PART I, FINANCIAL INFORMATION	3
Item 1. Financial Statements	3
Consolidated Balance Sheets as of March 31, 2007 (Unaudited) and December 31, 2006	3
Unaudited Consolidated Statements of Operations for the three months ended March 31, 2007 and 2006	4
Unaudited Consolidated Statements of Cash Flows for the three months ended March 31, 2007 and 2006	5
Notes to Unaudited Condensed Consolidated Financial Statements	6
Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations	10
Item 3. Quantitative and Qualitative Disclosure about Market Risk	14
Item 4. Controls and Procedures	15
PART II. OTHER INFORMATION	15
<u>Item 1. Legal Proceedings</u>	15
Item 1A. Risk Factors	15
<u>Item 2. Unregistered Sales of Equity Securities and Use of Proceeds</u>	25
Item 3. Defaults Upon Senior Securities	25
<u>Item 4. Submission of Matters to a Vote of Security Holders</u>	25
<u>Item 5. Other Information</u>	25
<u>Item 6. Exhibits</u>	25
<u>SIGNATURES</u>	27
EXHIBIT 10.2	
<u>EXHIBIT 31.1</u>	
EXHIBIT 31.2	
<u>EXHIBIT 32.1</u>	
EXHIBIT 32.2	

PART I. FINANCIAL INFORMATION

ITEM 1. FINANCIAL STATEMENTS

Digirad Corporation Consolidated Balance Sheets (In thousands, except par value amounts)

	March 31, 2007	December 31, 2006
Assets	(Unaudited)	
Current assets:		
Cash and cash equivalents	\$ 9,711	\$ 10,070
Securities available-for-sale	30,054	34,256
Accounts receivable, net	8,836	7,534
Inventories, net	6,801	5,860
Other current assets	1,505	1,499
Total current assets	56,907	59,219
Property and equipment, net	10,689	9,570
Intangibles, net	422	428
Restricted cash	60	60
Total assets	\$ 68,078	\$ 69,277
Liabilities and stockholders' equity		
Current liabilities:		
Accounts payable	\$ 2,029	\$ 2,643
Accrued compensation	2,497	3,650
Accrued warranty	993	788
Other accrued liabilities	3,351	3,306
Deferred revenue	2,721	2,775
Current portion of long-term debt	294	269
Total current liabilities	11,885	13,431
Long-term debt, net of current portion	41	99
Deferred rent	285	302
Commitments and contingencies		
Stockholders' equity:		
Preferred stock, \$0.0001 par value: 10,000 shares authorized at March 31, 2007 and December 31, 2006; no shares issued or		
outstanding at March 31, 2007 and December 31, 2006	_	_
Common stock, \$0.0001 par value: 80,000 shares authorized at March 31, 2007 and December 31, 2006; 18,818 and 18,795		
shares issued and outstanding at March 31, 2007 and December 31, 2006, respectively	2	2
Additional paid-in capital	151,817	151,539
Accumulated other comprehensive loss	(20)	(91)
Accumulated deficit	(95,932)	(96,005)
Total stockholders' equity	55,867	55,445
Total liabilities and stockholders' equity	\$ 68,078	\$ 69,277

See accompanying notes.

Digirad Corporation Consolidated Statements of Operations (In thousands, except per share amounts) (Unaudited)

		ended March 31,
Revenues:		2006
DIS	\$ 12,197	\$ 13,217
Product	5,341	5,738
Total revenues	17,538	18,955
Cost of revenues:	,	
DIS	8,938	10,432
Product	3,158	4,130
Total cost of revenues	12,096	14,562
Gross profit	5,442	4,393
Operating expenses:		
Research and development	782	1,096
Sales and marketing	2,098	2,459
General and administrative	2,978	4,138
Total operating expenses	5,858	7,693
Loss from operations	(416)	(3,300)
Other income (expense):		
Interest income	475	522
Interest expense	(11)	(26)
Other	26	
Total other income	490	496
Net income (loss)	\$ 74	\$ (2,804)
Net income (loss) per common share – basic and diluted	\$ 0.00	\$ (0.15)
Weighted average shares outstanding – basic	18,815	18,710
Weighted average shares outstanding – diluted	19,200	18,710

See accompanying notes.

Digirad Corporation Consolidated Statements of Cash Flows (In thousands) (Unaudited)

		ended March 31,
Operating activities	2007	2006
Net income (loss)	\$ 74	\$ (2,804)
Adjustments to reconcile net income (loss) to net cash used by operating activities:	Ψ / 1	ψ (2,004)
Depreciation	855	1,335
Loss on disposal of assets	19	
Amortization of premium on securities available-for-sale	22	56
Amortization and write-off of intangibles	6	14
Stock-based compensation	267	476
Changes in operating assets and liabilities:		
Accounts receivable	(1,302)	(894)
Inventories	(941)	999
Other assets	(6)	130
Accounts payable	(614)	413
Accrued compensation	(1,153)	(160)
Accrued warranty, deferred rent and other accrued liabilities	233	(560)
Deferred revenue	(54)	11
Net cash used by operating activities	(2,594)	(984)
Investing activities		
Purchases of securities available-for-sale	(2,750)	(4,749)
Maturities of securities available-for-sale	7,000	8,200
Purchases of property and equipment	(1,993)	(1,353)
Patents and other assets		(65)
Net cash provided by investing activities	2,257	2,033
Financing activities		
Issuances of common stock	11	18
Proceeds from capital lease financing	37	_
Repayment of obligations under capital leases	(70)	(243)
Net cash used by financing activities	(22)	(225)
Net decrease (increase) in cash and cash equivalents	(359)	824
Cash and cash equivalents at beginning of period	10,070	16,303
Cash and cash equivalents at end of period	\$ 9,711	\$ 17,127

See accompanying notes.

DIGIRAD CORPORATION NOTES TO UNAUDITED CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

(in thousands, except per share amounts)

1. Interim Financial Information

Organization and Business

Digirad Corporation ("Digirad"), a Delaware corporation, is a provider of cardiovascular imaging services and solid-state nuclear medicine imaging products to physician offices, hospitals and imaging centers. Digirad has two reportable segments, Digirad Imaging Solutions ("DIS") and Product. Through DIS, we provide in-office services to physicians, offering certified personnel, required licensure, an imaging system and other support and supplies for the performance of nuclear imaging procedures under the supervision of our physician customers. DIS physician customers enter into annual lease contracts for imaging services delivered on a per-day basis. Our Product segment sells solid-state gamma cameras and provides camera service and maintenance contracts to physician offices, hospitals, and imaging centers primarily in the United States.

Basis of Presentation

We have prepared the accompanying unaudited consolidated financial statements in accordance with accounting principles generally accepted in the United States of America for interim financial information and with the instructions to Form 10-Q and Article 10 of Regulation S-X. Accordingly, they do not include all of the information and disclosures required by generally accepted accounting principles for complete financial statements. In the opinion of our management, all adjustments (consisting of normal recurring accruals) considered necessary for a fair presentation have been included. Intercompany accounts have been eliminated in consolidation. Operating results for the three months ended March 31, 2007 are not necessarily indicative of the results that may be expected for the entire year. For further information see our financial statements and related disclosures thereto for the year ended December 31, 2006 in our Annual Report on Form 10-K filed with the Securities and Exchange Commission.

Net Income (Loss) Per Share

We calculate net income (loss) per share in accordance with SFAS No. 128 ("SFAS 128"), *Earnings Per Share*. SFAS 128 requires presentation of "basic" earnings per share and "diluted" earnings per share. Basic earnings per share ("EPS") is calculated by dividing the net income or loss by the weighted average number of common shares outstanding for the period, without consideration for common stock equivalents. Diluted EPS is computed by dividing the net income by the weighted average number of common shares outstanding for the period and the weighted average number of dilutive common stock equivalents such as options and warrants. Options and warrants are only included in the calculation of diluted earnings per share when their effect is dilutive.

The weighted average shares (share data in thousands) used to calculate EPS was 18,815 and 18,710 for the three months ended March 31, 2007 and 2006. The difference between the calculation of basic and diluted EPS for the three months ended March 31, 2007 is attributable to outstanding stock options. Stock options had the effect of increasing the number of shares used in the calculation (by application of the treasury stock method) by 385. Stock options of 1,354 were not included in the calculation of diluted earnings per share for the three months ended March 31, 2007 as the effect of exercising these options would have been anti-dilutive.

2. Financial Statement Details

Inventories consist of the following (in thousands):

	March 31, 	Dec	ember 31, 2006
Raw materials	\$ 2,865	\$	2,985
Work-in-progress	3,825		3,316
Finished goods	1,008		471
	7,698		6,772
Less reserves for excess and obsolete inventories	(897)		(912)
	\$ 6,801	\$	5,860

Property and equipment consist of the following (in thousands):

	March 31, 2007	December 31, 2006
Machinery and equipment	\$ 22,121	\$ 21,276
Computers and software	3,686	3,446
Leasehold improvements	762	749
Furniture and fixtures	158	158
	26,727	25,629
Less accumulated depreciation and amortization	(16,038)	(16,059)
	\$ 10,689	\$ 9,570

Other accrued liabilities consist of the following (in thousands):

	rch 31, 2007	ember 31, 2006
Radiopharmaceuticals and consumable medical supplies	\$ 820	\$ 579
Professional fees	407	495
Outside services and consulting	480	454
Customer deposits	300	355
Facilities and related costs	246	279
Travel expenses	259	244
Sales and property taxes payable	165	236
Other accrued liabilities	 674	664
	\$ 3,351	\$ 3,306

3. Warranty

We provide a warranty on certain of our products and accrue the estimated cost at the time revenue is recorded. Warranty expense is charged to product cost of revenues. The majority of all warranty periods are 12 months before customer-sponsored maintenance begins. Warranty reserves are established based on historical experience with failure rates and repair costs and the number of gamma cameras covered by warranty. Warranty reserves are depleted as gamma cameras are repaired. The costs consist principally of materials, personnel, overhead and transportation. We review warranty reserves quarterly and, if necessary, make adjustments.

The activities in our warranty reserve are as follows (in thousands):

	Thre	Three months ended March 31,		
	2007		2006	
Balance at beginning of period	\$	788 \$	825	
Charges to cost of revenues		523	260	
Applied to liability	(318)	(306)	
Balance at end of period	\$	993 \$	779	

4. Comprehensive Income (Loss)

Comprehensive income (loss) consists of the following components (in thousands):

	Thre	Three months ended March 31,	
	2007	2007	
Net income (loss), as reported	\$	74 \$	\$ (2,804)
Unrealized gains (losses) on marketable securities		71	(43)
Comprehensive income (loss)	\$ 1	45 <u>\$</u>	\$ (2,847)

5. Stock-Based Compensation

We have one stock option plan under which stock options are granted to our employees and directors. Stock options granted under this plan generally have a term of ten years from the date of grant and vest over four years. Prior to June 2004,

we were authorized to issue options under various other option plans and programs, however, no additional awards may now be made under such plans. For purposes of accounting for stock-based compensation, the fair value of each option award is estimated on the date of grant using the Black-Scholes-Merton option pricing formula. There were no significant modifications to our share-based employee payment plans during the periods presented that resulted in any incremental compensation cost. As of March 31, 2007, \$2.2 million of total unrecognized compensation cost related to unvested share-based compensation arrangements granted under our various plans is expected to be recognized over a weighted-average period of 2.0 years.

Following is a summary of stock-based compensation costs by income statement classification (in thousands):

	,	Three months ended March 31,		
		2007		006
Cost of DIS revenue	\$	25	\$	20
Cost of product revenue		26		18
Research and development		23		42
Sales and marketing		50		75
General and administrative		150		316
Total stock compensation	\$	274	\$	471

6. Segments

Our reporting segments have been determined based on the nature of the products and/or services offered to customers or the nature of their function in the organization. We evaluate performance based on the operating income contributed by each segment. The accounting policies of the reportable segments are the same as those described in the summary of significant accounting policies.

Segment results are as follows (in thousands):

		ree months e	nded I	March 31, 2006
Gross profit by segment:				
DIS	\$	3,259	\$	2,785
Product		2,183		1,608
Consolidated gross profit	\$	5,442	\$	4,393
Income (loss) from operations by segment:				
DIS	\$	123	\$	(1,612)
Product		(539)		(1,688)
Consolidated loss from operations	\$	(416)	\$	(3,300)
Depreciation, and amortization of intangible assets by segment:				
DIS	\$	628	\$	1,038
Product		233		311
Consolidated depreciation and amortization	\$	861	\$	1,349
		As of March 31, 2007 2006		
Identifiable assets by segment:	_		_	
DIS	\$	16,340	\$	14,764
Product		51,738		56,848
Consolidated assets	\$	68,078	\$	71,612

Although we have historically sold a small number of imaging systems internationally, foreign sales are not a significant source of our revenues for the periods presented above.

7. Commitments and Contingencies

Compliance with Laws and Regulations

We are, directly or indirectly through our clients, subject to extensive regulation by the federal government, the states and foreign countries in which we conduct business. The healthcare laws applicable to us are complex and are subject to variable interpretations. We have established a compliance program to identify any compliance issues, correct any identified issues and assist us in remaining in compliance with the applicable healthcare laws, and have instituted other safeguards intended to help prevent any violations of the laws and to remedy any situations that could give rise to violations. We can provide no assurance that these measures will be successful in preventing compliance violations and the resulting fines, penalties or damages.

Legal Matters

In the normal course of business, we have been and will likely continue to be subject to litigation or administrative proceedings incidental to our business, such as claims related to customer disputes, employment practices, wage and hour disputes, product liability, professional liability, commercial disputes, licensure restrictions or denials, and warranty or patent infringement. Responding to litigation or administrative proceedings, regardless of whether they have merit, can be expensive and disruptive to normal business operations. As litigation and the administrative proceedings are inherently uncertain, we cannot predict the outcome of such matters. We can provide no assurance that the ultimate outcome, either individually or in the aggregate, will not have a material adverse effect on our business or financial results.

8. Income Taxes

On January 1, 2007, we adopted the provisions of Financial Accounting Standards Board Interpretation No. 48, "Accounting for Uncertainty in Income Taxes" ("FIN 48"), which clarifies the accounting for uncertainty in income taxes recognized in a company's financial statements in accordance with Statement of Financial Accounting Standards No. 109, "Accounting for Income Taxes." The adoption of FIN 48 did not impact our consolidated financial condition, results of operations or cash flows. At January 1, 2007, we had net deferred tax assets of \$36.8 million. These deferred tax assets are primarily composed of federal and state tax net operating loss ("NOL") carryforwards and federal and state research and development ("R&D") credit carryforwards. Due to uncertainties surrounding our ability to generate future taxable income to realize these assets, a full valuation has been established to offset our net deferred tax asset. Additionally, the future utilization of our NOL and R&D credit carryforwards to offset future taxable income may be subject to a substantial annual limitation as a result of ownership changes that may have occurred previously or that could occur in the future. Until we have determined whether such an ownership change has occurred, and until the amount of any limitation becomes known, no amounts are being presented as an uncertain tax position in accordance with FIN 48.

Management believes that the amount subject to limitation could be significant. Any carryforwards that will expire prior to utilization as a result of such limitations will be removed from deferred tax assets with a corresponding reduction of the valuation allowance. Due to the existence of the valuation allowance, future changes in our unrecognized tax benefits will not impact our effective tax rate.

We file income tax returns in the U.S. and in various state jurisdictions. We are no longer subject to income tax examination by tax authorities for years prior to 2003, however, our net operating loss and research and development carryforwards arising prior to that year are subject to adjustment. Our policy is to recognize interest and penalties related to income tax matters as a component of income tax expense.

9. Subsequent Event

On May 1, 2007, we acquired substantially all of the assets of Ultrascan, Inc. ("Ultrascan"), a leading provider of mobile ultrasound and fixed-site nuclear medicine services primarily in Georgia, in exchange for cash consideration of \$7.25 million and the assumption of debt obligations totaling \$1.5 million. The purchase price is subject to a working capital adjustment after the closing of the transaction. Additional consideration, payable in cash and common stock, of up to \$3.85 million may be payable to the seller or its designees in the event that Ultrascan achieves certain financial milestones over the next four years.

ITEM 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

The following discussion and analysis should be read in conjunction with our financial statements and notes thereto included in this report on Form 10-Q, and the audited financial statements and notes thereto as of and for the year ended December 31, 2006 included in our Annual Report on Form 10-K filed with the Securities and Exchange Commission on February 20, 2007. Operating results are not necessarily indicative of results that may occur in future periods.

This report includes various forward-looking statements that are subject to risks and uncertainties, many of which are beyond our control. Our actual results could differ materially from those anticipated in these forward looking statements as a result of various factors, including those set forth below under the caption "Risk Factors." Forward-looking statements discuss matters that are not historical facts. Forward-looking statements include, but are not limited to, discussions regarding our operating strategy, growth strategy, product development, cost savings initiatives, industry, economic and market conditions, financial condition, liquidity and capital resources and results of operations. In this report, for example, we make forward-looking statements regarding our expectations about revenue growth in DIS as we expand into new markets and continue to improve utilization rates within existing markets, growth in our ultrasound services, our efforts to obtain accreditation for our DIS hubs, the effects of changes to the Stark Law, the size and impact of potential Medicare and other third party payor reimbursement decreases, the roll-out of our multi-headed Cardius XPO camera into DIS and the value it represents to our customers, persistent pricing pressures affecting our sales of multi-headed gamma cameras, and continuing investments in research and development initiatives. Such statements include, but are not limited to, statements preceded by, followed by or that otherwise include the words "believes," "expects," "anticipates," "intends," "estimates," "projects," "can," "could," "may," "will," "would," or similar expressions. For those statements, we claim the protection of the safe harbor for forward-looking statements contained in the Private Securities Litigation Reform Act of 1995. You should not unduly rely on these forward-looking statements, which speak only as of the date on which they were made. We undertake no obligation to update publicly or revise any forward-looking statements, whether as a r

Overview

We are a leading provider of cardiovascular imaging services and solid-state nuclear medicine imaging products to physician offices, hospitals and imaging centers. We designed and commercialized the first solid-state nuclear gamma camera for the detection of cardiovascular disease and other medical conditions. Our imaging systems are mobile as well as fixed and provide enhanced operability, improved patient comfort and, in the case of our triple-headed Cardius®-3 XPO system, shorter image acquisition time, when compared to traditional vacuum tube cameras. The cameras and accompanying equipment fit easily into floor spaces as small as seven feet by eight feet and facilitate the delivery of nuclear medicine procedures in a physician's office, an outpatient hospital setting or within multiple departments of a hospital.

First Quarter 2007 Financial Highlights

Our consolidated revenues were \$17.5 million during the three months ended March 31, 2007 ("2007"), which represented a decrease of 7.5% over the comparable prior year period ("2006") due to the decrease in revenue in both segments of our business. In DIS, revenue decreased 7.7% to \$12.2 million, primarily due to DIS having phased out the delivery of stress agents. In the product business, revenue decreased 6.9% to \$5.3 million. Consolidated net income for the quarter was \$74,000, compared to a net loss of \$2.8 million during the same period in 2006.

DIS began to phase out providing stress agents used in some imaging procedures in June 2006. As a result, we recognized no stress agent revenue in 2007 compared to \$1.1 million recognized in 2006. As of March 31, 2007, our DIS segment operated in 22 states and the District of Columbia. DIS service days were unchanged at 3,461 in 2007 compared to 2006. Revenue per day was \$3,524 for 2007 compared to \$3,489 for 2006 (excluding the stress agent revenue we received in 2006). Our DIS gross margins in 2007 increased to 26.7% compared to 21.1% in 2006, due primarily to lower pharmaceutical costs and a reduction in depreciation expense.

As of March 31, 2007, DIS operated 82 mobile cameras compared to 80 cameras during the same period in 2006. In connection with our plan to upgrade our DIS camera fleet over the next few years, we placed 8 incremental triple-headed cameras into our DIS business in the first quarter of 2007, bringing the total number of such cameras in the fleet to 27. We continue to obtain additional hub accreditation to respond to the reimbursement requirements of some third party payors. As of March 31, 2007, we had obtained accreditation from the Intersocietal Commission for the Accreditation of Nuclear Medicine Laboratories for 28 of our DIS hub locations.

Our product business delivered 19 gamma cameras in the first quarter of 2007, an increase of one system compared to 2006. Product revenue decreased over the prior year period by \$0.4 million, primarily due to a change in the mix of cameras

sold and reduced average selling prices due to continuing competitive pricing pressures. Product gross margins improved to 40.9% in 2007 compared to 28.0% during 2006, due primarily to reduced labor and overhead expenses and improved manufacturing yields. During the first quarter of 2007, we expanded our product portfolio by adding the XPO features already available on our Cardius-3 camera to the Cardius-1 and Cardius-2 cameras. Our new Cardius XPO camera series allows physicians to choose among single-, dual- and triple-head cameras to accommodate their practices' speed and throughput needs, or to upgrade a single head camera to a dual or triple head configuration as their practice grows and changes. These cameras can image patients weighing up to 500 pounds and include our latest image acquisition and processing software.

Our Market

The target market for our products and services is comprised of approximately 26,000 cardiologists, the larger practices among the 130,000 internists and family practitioners, and hospitals in the United States that perform or could perform nuclear cardiac procedures. We estimate that there are approximately 8,000 internist practices with more than four internists. As of March 31, 2007, we have provided imaging services through DIS to approximately 600 physicians and physician groups, the majority of which are cardiologists, and have sold 477 cameras through our product segment.

According to IMV Medical Information, U.S. procedure volume in nuclear medicine (excluding PET studies) grew by 15% between 2002 and 2005 to an estimated 19.7 million nuclear imaging procedures in 2005, of which some 11.2 million were cardiovascular-specific procedures. The National Electrical Manufacturers Association, or NEMA, estimates that sales of general nuclear imaging equipment declined approximately 10% during 2006.

We believe our market has been negatively affected by declining reimbursement, significant pricing pressures in the product business, and continuing efforts by some third party payors to reduce health care expenditures by requiring physicians to obtain specific accreditations or certifications, or disallowing reimbursement if imaging is performed with mobile or leased cameras. A number of companies have recently begun to market mobile nuclear imaging cameras for cardiac applications, and we believe competition from local or regional companies providing mobile imaging services has increased. We expect each of these trends to continue.

Revenue Sources

We generated revenues within two primary operating segments: our DIS business and our product sales business. Through DIS, we offer a comprehensive mobile imaging services leasing program as an alternative to purchasing a gamma camera or ultrasound machine for physicians who wish to perform nuclear imaging, echocardiography, vascular ultrasound or any combination of these procedures in their offices by leasing the imaging system, certified personnel and other support required to perform imaging in the physician office. The flexibility of our products and our DIS leasing service allows physicians more control over the diagnosis and treatment of their patients in their offices and to retain revenue from procedures they would otherwise refer elsewhere. We also offer DigiTech leases to customers who own one of our nuclear gamma cameras or ultrasound machines but wish us to provide staffing and other support services. DIS leasing services are primarily provided to cardiologists and internists. Physicians enter into annual contracts for imaging services delivered on a per-day basis. Our typical lease contracts provide service coverage ranging from once per month to three times per week, adjusted for holidays and vacations. We experience some seasonality in our DIS business related to summer slowdowns (principally relating to vacations), holidays and inclement weather. Historically, the DIS results have been most negatively affected by seasonality in the third quarter.

Our product revenue results primarily from selling solid-state gamma cameras and other ancillary items, and from our camera maintenance contracts. We sell our imaging systems to physician offices, hospitals, and imaging centers primarily in the United States, although we have sold a small number of imaging systems internationally. We do not anticipate that the international market will be a significant source of revenues in the foreseeable future.

Trends and Drivers

We believe this quarter's improved net income and gross margins in both our business segments are the results of our efforts to enhance operating efficiencies, lower material costs and reduce operating expenses. We expect these trends to continue, and to reap further labor cost reductions and camera reliability improvements as we continue to upgrade our DIS mobile camera fleet. We have also commenced the delivery of mobile ultrasound imaging services, a first step in diversifying our mobile service model.

The decrease in DIS revenue is primarily attributable to the decision to stop the delivery of stress agents, which contributed approximately \$1.1 million to the results in the same quarter of 2006. The increase in the number of physicians who either bought their own cameras or switched to other mobile imaging service providers during the second half of 2006 continued during the first quarter of 2007. We attribute this trend to increasing competition from local and regional companies offering mobile imaging products or services.

We also note that there were Medicare reimbursement cuts of approximately 8.5% for 2007 as compared to 2006 for the imaging procedures most often performed by our physician customers.

In the product business, industry sources predict that the rate of decline of sales of cardiac-specific nuclear cameras will slow from an estimated 24% in 2006 to approximately 5% in 2007, and that purchases of multi-head cameras will far outpace those of single-head cameras. We believe pricing pressures persist in the cardiac-specific camera market, evidenced by the declining average selling price of nuclear cameras. We will continue to invest in research and development to improve the image quality, speed, reliability, cost structure and overall performance of our multi-headed cameras and software to counteract these pricing pressures. The hospital market has expressed a renewed interest in our general purpose single-head camera, and we expect to continue to sell more cameras into this market in 2007.

Results of Operations

The following table sets forth our results from operations, expressed as percentages of revenues for the three months ended March 31, 2007 and 2006:

	Three months ende	Three months ended March 31,	
	2007	2006	
Revenues:			
DIS	69.5%	69.7%	
Product	30.5	30.3	
Total revenues	100.0	100.0	
Total cost of revenues	69.0	76.8	
Gross profit	31.0	23.2	
Operating expenses:			
Research and development	4.5	5.8	
Sales and marketing	12.0	13.0	
General and administrative	16.9	21.8	
Total operating expenses	33.4	40.6	
Loss from operations	(2.4)	(17.4)	
Other income (expense)	2.8	2.6	
Net income (loss)	0.4%	(14.8)%	

Comparison of Three Months Ended March 31, 2007 and 2006

Revenues

Consolidated. Consolidated revenue was \$17.5 million for 2007, which represents a decrease of \$1.4 million, or 7.5% over 2006, primarily as a result of the discontinuation of the sale of stress agents by DIS in June 2006, and a decline in product segment revenue associated with our selling a higher number of used and lower-priced cameras at a reduced average selling price. DIS and product revenue accounted for 69.5% and 30.5%, respectively, of total revenues for 2007, compared to 69.7% and 30.3%, respectively, for 2006. We expect DIS revenue to continue to represent the larger percentage of our consolidated revenue in future periods.

DIS. Our DIS revenue decreased to \$12.2 million for 2007, which represents a decrease of \$1.0 million, or 7.7%, over the prior year quarter. This decrease was primarily the result of our decision to discontinue the sale of stress agents in June 2006 (stress agent revenue was \$1.1 million for 2006). We seek to increase our DIS revenue by broadening our services to include ultrasound imaging, penetrating existing markets and expanding into new markets. Any growth will fluctuate, however, based on seasonality stemming from physician vacations, holidays, and incleme+6nt weather and from the start up time required as we enter new geographic areas.

Product. Our product revenue decreased to \$5.3 million for 2007, representing a decrease of \$0.4 million, or 6.9%, over the prior year quarter. The decrease in product revenue is attributable to a decline in the average selling prices for our gamma cameras and selling more used cameras in 2007 than in 2006.

Gross Profit

Consolidated. Consolidated gross profit was \$5.4 million for 2007, representing an increase of \$1.0 million or 23.9%, compared to the prior year quarter. The increase in consolidated gross profit is principally the result of our efforts to improve operational efficiencies and lower material costs as well as to improve product reliability and serviceability. Consolidated gross profit as a percentage of revenue increased to 31.0% for 2007 from 23.2% for 2006.

DIS. Cost of DIS revenue consists primarily of labor, radiopharmaceuticals, equipment depreciation, and other costs associated with the provision of services. Cost of DIS revenue decreased to \$8.9 million for 2007, representing a decrease of \$1.5 million, or 14.3%, over the prior year quarter, primarily a result of the decrease in revenue. DIS gross profit increased to \$3.3 million for 2007, which represents an increase of \$0.5 million, or 17.0%. DIS gross profit as a percentage of revenue increased to 26.7% for 2007 from 21.1% for 2006, primarily as a result of a reduction in radiopharmaceutical costs and a reduction in depreciation expense.

Product. Cost of goods sold primarily consists of materials, labor and overhead costs associated with the manufacturing and warranty of our products. Warranty costs are charged to cost of goods sold in the period our cameras are sold and are based on our historical experience with failure rates and repair costs. Cost of goods sold was \$3.2 million for 2007, representing a decrease of \$1.0 million, or 23.5%, compared to the prior year quarter. Product gross profit increased to \$2.2 million for 2007 as compared to \$1.6 million for 2006. Product gross profit as a percentage of revenue increased to 40.9% for 2007 from 28.0% for 2006. Product margin improvement is due to a reduction in material costs and an improvement in operating efficiency.

Operating Expenses

Research and Development. Research and development expenses consist primarily of costs associated with the design, development, testing, and enhancement of our products. The primary costs are salaries and fringe benefits, development material costs, facility and overhead costs, consulting fees, and non-recurring engineering costs. Research and development expenses were \$0.8 million for 2007, which represents a decrease of \$0.3 million, or 28.6%, over the prior year quarter. This was primarily attributable to a reduction in the number of research personnel and decreased spending on indirect materials associated with new product development. Research and development expenses were 4.5% of total revenue for 2007 compared to 5.8% for 2006. In the future, we expect to maintain our investment in research and development as we innovate and seek to improve our existing technology.

Sales and Marketing. Sales and marketing expenses consist primarily of salaries, commissions, bonuses, recruiting costs, travel, marketing and collateral materials and trade show costs. Sales and marketing expenses were \$2.1 million for 2007, representing a decrease of \$0.4 million or 14.7%, over the prior year quarter, primarily as a result of a reduction in personnel and travel costs and trade show expenses. Sales and marketing expenses were 12.0% of total revenue for 2007 compared to 13.0% for 2006. We expect to increase our sales and marketing efforts as we expand into new territories and launch a marketing program for echocardiography and vascular ultrasound.

General and Administrative. General and administrative expenses consist primarily of salaries and other related costs for finance, accounting, human resources and other personnel, and legal and other professional fees and insurance. General and administrative expenses were \$3.0 million for 2007, representing a decrease of \$1.2 million or 28.0%, over the prior year quarter. In 2006, the cost of certain personnel was shifted from general and administrative expenses to cost of goods sold to reflect the change in the nature of their job responsibilities. Additionally, we experienced a \$166,000 reduction in stock based compensation costs as well as lower legal costs, and our cost reduction efforts have resulted in a lower level of spending associated with recruiting, consulting fees, and other outside services. General and administrative expenses were 16.9% of total revenue for 2007 compared to 21.8% for 2006.

Other Income (Expense)

Other income (expense) consists primarily of interest income, net of interest and other expenses. Other income was essentially unchanged for both periods presented.

Net Income (Loss)

Our net income was \$0.1 million for 2007 compared to a net loss of \$2.8 million for 2006, primarily as a result of the factors described above.

Liquidity and Capital Resources

We require capital principally for debt service, capital expenditures and working capital to finance accounts receivable and inventory. Our working capital requirements vary from period to period depending on manufacturing volumes, the timing of deliveries and the payment cycles of our customers. Our capital expenditures consist primarily of DIS nuclear cameras, ultrasound machines, vans, and computer hardware and software. As of March 31, 2007, we had cash, cash equivalents and investments of \$39.8 million. We currently invest our cash reserves in money market funds, high-grade auction rate securities and U.S. government or corporate debt securities. Based upon our current level of expenditures, we believe our current working capital together with cash flows from operating activities, will be adequate to meet our anticipated cash requirements for working capital, debt service and capital expenditures for at least the next 12 months.

Net cash used in operations totaled \$2.6 million for the three months ended March 31, 2007, despite positive cash flow from net income and non-cash charges (including depreciation, amortization and stock-based compensation). The cash used in operations is due to increases in both receivables and inventory and the reduction of accounts payable and accrued compensation. The increase in receivables reflects the increase in revenue activity at the end of the first quarter as compared to the previous quarter, while the increase in inventory is primarily associated with an increase in finished goods. The decrease in accounts payable reflects routine fluctuation in vendor activity and the decrease in accrued compensation is primarily associated with the payment of year end bonuses to our employees. Net cash provided by investing activities amounted to \$2.3 million for the three months ended March 31, 2007, and is comprised of the cash flow from net maturities of securities available-for-sale of \$4.3 million partially offset by \$2.0 million of capital expenditures primarily associated with our DIS operations. Net cash used by financing activities amounted to approximately \$22,000 for the three months ended March 31, 2007, and represents the repayment of capital lease obligations, net of proceeds arising from the exercise of stock options.

On May 1, 2007, we acquired substantially all of the assets of Ultrascan, Inc. ("Ultrascan") in exchange for cash consideration of \$7.25 million and the assumption of debt obligations totaling \$1.5 million. We repaid the debt obligations at the time of the acquisition. Additional consideration, payable in cash and common stock, of up to \$3.85 million may be payable to the seller or its designees in the event that Ultrascan achieves certain financial milestones over the next four years.

Critical Accounting Policies

Management's discussion and analysis of our financial condition and results of operations are based upon our consolidated financial statements which are prepared in accordance with accounting principles that are generally accepted in the United States. The preparation of these financial statements requires us to make estimates and judgments that affect the reported amounts of assets and liabilities, related disclosure of contingent assets and liabilities at the date of the financial statements, and the reported amounts of revenues and expenses during the reporting period. We continually evaluate our estimates and judgments, the most critical of which are those related to revenue recognition and inventory valuation. We base our estimates and judgments on historical experience and other factors that we believe to be reasonable under the circumstances. Materially different results can occur as circumstances change and additional information becomes known.

There were no significant changes during the quarter ended March 31, 2007 to the items that we disclosed as our critical accounting policies and estimates in Management's Discussion and Analysis of Financial Condition and Results of Operations in the Company's Annual Report on Form 10-K for the year ended December 31, 2006.

New accounting requirement. Effective January 1, 2007, we adopted Financial Accounting Standards Board, or FASB, Interpretation No. 48, "Accounting for Uncertainty in Income Taxes – an interpretation of SFAS No. 109", or FIN No. 48, which prescribes a recognition threshold and measurement process for recording in the financial statements uncertain tax positions taken or expected to be taken in a tax return. Additionally, FIN No. 48 provides guidance on the derecognition, classification, accounting in interim periods and disclosure requirements for uncertain tax positions. The adoption of FIN 48 did not impact our consolidated financial condition, results of operations or cash flows. We do not anticipate that the adoption of FIN No. 48 will have a material effect on our effective tax rate in future periods.

Corporate Information

As of March 31, 2007, we hold trademark registrations in the United States for the following marks: 2020tc Imager®, CardiusSST®, Digirad®, Digirad Logo®, Digirad Imaging Solutions®, FlexImaging®, Cardius®, SPECTour®, DigiServ®, DigiTech®, SPECTpak Plus® and Solidium®. We have trademark applications pending in the United States for the following marks: SeeQuanta™, AcqSmart™, Stasys™, Cardius X-Act™, and TruAcq CountBased Imaging™. We have obtained and sought trademark protection for some of these listed marks in the European Community and Japan.

ITEM 3. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

Our exposure to market risk due to changes in interest rates relates primarily to the increase or decrease in the amount of interest income we can earn on our investment portfolio. Our risk associated with fluctuating interest rates is limited to our investments in interest rate sensitive financial instruments. Under our current policies, we do not use interest rate derivative instruments to manage exposure to interest rate changes. We attempt to increase the safety and preservation of our invested principal funds by limiting default risk, market risk and reinvestment risk. We mitigate default risk by investing in investment grade securities. A hypothetical 100 basis point adverse move in interest rates along the entire interest rate yield curve would not materially affect the fair value of our interest sensitive financial instruments. Changes in interest rates over time will increase or decrease our interest income.

ITEM 4. CONTROLS AND PROCEDURES

Disclosure Controls and Procedures

We maintain disclosure controls and procedures and internal controls that are designed to provide reasonable assurance that information required to be disclosed in our Exchange Act reports is recorded, processed, summarized and reported within the time periods specified in the Securities and Exchange Commission's rules and forms and that such information is accumulated and communicated to our management, including our Chief Executive Officer and Chief Financial Officer, as appropriate, to allow timely decisions regarding required disclosure. In designing and evaluating the disclosure controls and procedures and internal controls, management recognizes that any controls and procedures, no matter how well designed and operated, can provide only reasonable, and not absolute, assurance of achieving the desired control objectives. In reaching a reasonable level of assurance, management necessarily was required to apply its judgment in evaluating the cost benefit relationship of possible controls and procedures and internal controls.

As required by the Securities and Exchange Commission Rule 13a-15(e) and Rule 15d-15(e), we carried out an evaluation, under the supervision of and with the participation of our management, including our Chief Executive Officer and Chief Financial Officer, of the effectiveness of the design and operation of our disclosure controls and procedures as of the end of the period covered by this report. Based on the foregoing, our Chief Executive Officer and Chief Financial Officer concluded that our disclosure controls and procedures were effective at the reasonable assurance level.

Changes in Internal Control over Financial Reporting.

In addition, an evaluation was performed under the supervision of and with the participation of our management, including our Chief Executive Officer and Chief Financial Officer, of any change in our internal controls over financial reporting that has occurred during our last fiscal quarter that has materially affected, or is reasonably likely to affect materially, our internal controls over financial reporting. There has been no change in our internal controls over financial reporting during our most recent fiscal quarter that has materially affected, or is reasonably likely to materially affect, our internal controls over financial reporting.

PART II. OTHER INFORMATION

ITEM 1. LEGAL PROCEEDINGS

In the normal course of business, we have been and will likely continue to be subject to litigation or administrative proceedings incidental to our business, such as claims related to customer disputes, employment practices, wage and hour disputes, product liability, professional liability, commercial disputes, licensure restrictions or denials, and warranty or patent infringement. Responding to litigation or administrative proceedings, regardless of whether they have merit, can be expensive and disruptive to normal business operations. As litigation and the administrative proceedings are inherently uncertain, we cannot predict the outcome of such matters. We can provide no assurance that the ultimate outcome, either individually or in the aggregate, will not have a material adverse effect on our business or financial results.

ITEM 1A. RISK FACTORS

Risks Related to Our Business and Industry

Our industry is highly competitive.

The nuclear imaging industry is highly competitive, subject to rapid change and significantly affected by new product introductions and market activities of other industry participants. Our primary competitors with respect to nuclear imaging systems include Philips Medical Systems, General Electric Healthcare, Siemens Medical Systems and Toshiba Medical Systems. All of these competitors offer a full line of imaging cameras for each diagnostic imaging technology, including x-ray, magnetic resonance imaging, computerized tomography, ultrasound and nuclear medicine, as well as hybrid modalities that combine, for example, the technologies of positron emission tomography, or PET, with computed tomography, or CT. Many of our competitors and potential competitors enjoy significant advantages over us, including significantly greater name recognition and financial, technical, service and marketing resources; established relationships with healthcare professionals, customers and third-party payors; established distribution networks; technical features our current products do not possess; multiple product lines and the ability to offer rebates or bundle products to offer discounts or incentives; and greater resources for product development, sales and marketing.

Certain medical device companies are developing solid-state cameras that may compete with our product offerings. A privately-held company, Gamma Medica, is marketing a solid-state gamma camera for breast imaging. A second company, Spectrum Dynamics, has demonstrated a proof-of-concept solid-state gamma camera that we believe it may market in the cardiac segment, and a third company, Spectrica, is marketing a mobile cardiac camera based on vacuum tube technology. We anticipate that additional companies will dedicate significant resources to developing competing products and services that may demonstrate better image quality, ease of use or mobility than our imaging systems. If we are unable to compete effectively against our existing and future competitors, our sales will decline and our business will be harmed.

In providing DIS lease services, we compete against small businesses employing traditional vacuum tube cameras that cannot be moved in and out of physician offices. We also compete against physicians and companies that use Digirad cameras in local, regional and national mobile imaging businesses, some of which have the advantage of a lower cost structure, and competition from these mobile operations is likely to increase. In addition, we compete against imaging centers that install nuclear gamma cameras and make them available to referring physicians in their geographic vicinity. If these competitors are able to win significant portions of our business, our sales will decline and our business will be harmed.

The competitive nature of the nuclear imaging industry has affected the volume of both our camera sales and our leasing services, and the pricing of our gamma cameras. We anticipate that these pressures will continue.

If the market for nuclear imaging cameras continues to decrease, or if we are not successful in expanding our market share or product and service offerings, our revenues will decline and our business will be harmed.

The National Electrical Manufacturers Association, or NEMA, estimates that sales of nuclear imaging equipment, excluding maintenance revenue, declined approximately 10% in 2006. We believe this decline may be attributable to concerns about reimbursement changes and the increasing adoption of alternative imaging modalities, such as magnetic resonance imaging, computerized tomography, positron emission tomography, and hybrids among these modalities. We believe our market has been negatively affected by declining reimbursement, significant pricing pressures in the product business, increased competition in the mobile imaging business and increasing reimbursement restrictions. We expect each of these trends to continue.

If these trends continue and we are unable to offset their effects on our business by expanding our market share or successfully introducing alternative products and services, our business will be significantly harmed.

Changes in coverage and reimbursement policies of third-party payors may adversely impact our ability to market and sell our products and services.

Private third-party payors continue to act to contain or reduce healthcare costs through various means, including the movement to managed care systems where healthcare providers contract to provide comprehensive healthcare for a fixed fee per patient. A number of third party payors in geographic locations currently served by us issued guidelines preventing our physician customers from obtaining reimbursement for procedures they perform unless they own or lease our cameras on a full time basis. These and other payors are also requiring physicians to be accredited by either the Intersocietal Commission for the Accreditation of Nuclear Medicine Laboratories (ICANL) or by the American College of Radiology, and to meet certain other privileging standards. Some of these privileging standards also exclude physicians who are not radiologists or cardiologists from obtaining reimbursements for nuclear imaging procedures. An increasing number of our DIS customers are non-cardiologists who would not currently meet the certifications required by payors in certain geographic areas. We have obtained accreditation from the Intersocietal Commission for the Accreditation of Nuclear Medicine Laboratories for 28 of our hub locations to address certification requirements, but we cannot assure you that we will be successful in obtaining additional certifications, or that obtaining them will satisfy the requirements of these payors. We also cannot assure you that these third party payor guidelines will be changed, or that they will not be adopted by other third party payors, including Medicaid and Medicare. These continued efforts to restrict reimbursement have resulted in the denial of reimbursement in some instances. An increase in such denials will negatively affect our DIS business and product sales.

If we acquire new or complementary businesses, products or technologies instead of developing them ourselves, we may be unable to complete those acquisitions or to successfully integrate them in a cost-effective and non-disruptive manner.

Our success depends on our ability to continually enhance and broaden our product and service offerings in response to changing customer demands, competitive pressures and technologies. We may choose to pursue collaborations or acquisitions instead of developing businesses, products or technologies ourselves. We cannot assure you, however, that we would be able to successfully complete any acquisition or that we would be able to successfully integrate any acquired business, product or technology into our company in a cost-effective and non-disruptive manner. Furthermore, there is no certainty that we would be able to attract, hire or retain key employees associated with any acquired businesses, products or technologies.

Integrating any acquired businesses, products or technologies could be expensive and time consuming, disrupt our ongoing business and divert the attention and resources of our management. If we are unable to integrate any acquired businesses, products or technologies effectively, our business will likely suffer. Additionally, any amortization of assets or charges resulting from the costs of acquisitions could negatively impact our operating results.

Failure to attract qualified managers, engineers and imaging technologists, or high employee attrition rates, could limit our growth and adversely affect our business.

Our success is dependent on the efforts of our key executives and technical, sales and managerial personnel and our ability to retain them. Losing any one or more of these individuals could place a significant strain on our remaining management team and we may have difficulty replacing any of these individuals. Our future growth and ability to generate profits will depend in part upon our ability to identify, hire and retain nuclear imaging technologists, certified cardiographic technicians, nurses, radiation safety officers, engineers, management, sales personnel and other highly skilled personnel.

Hiring qualified management and technical personnel will be difficult due to the limited number of qualified candidates. Competition for these types of employees, particularly nuclear imaging technologists and engineers, is intense in the medical imaging field. Failure to attract, hire and retain key personnel could have an adverse effect on our business, financial condition and results of operations. For the quarter ended March 31, 2007, we experienced an 11.1% rate of employee turnover for the combined service and product segments. If we are unable to improve upon this metric, our business and financial condition may continue to be adversely affected.

Our imaging systems and DIS services may become obsolete, and we may not be able to timely develop new products, product enhancements or services that will be accepted by the market.

Our nuclear imaging systems and DIS services may become obsolete or unmarketable if other products or services utilizing new technologies or the development of hybrid imaging modalities are introduced by our competitors or new industry standards emerge. We cannot assure you that any future products and enhancements will be accepted by the market. To be successful, we will need to enhance our products or services and to design, develop and market new products that successfully respond to competitive developments, all of which may be expensive and time consuming.

The success of any new product offering or enhancement to an existing product will depend on several factors, including our ability to properly identify and anticipate physician and patient needs; develop new products or enhancements in a timely manner; obtain the necessary regulatory approvals or clearances for new products or product enhancements in a timely manner; provide adequate training to users of our products; price our products competitively; obtain required licensure; continue to offer cost-competitive products and services despite increasing reimbursement restrictions and pricing pressures; comply with changing or new regulatory requirements; and develop an effective marketing, sales and distribution network.

If we are unable to meet these requirements, our business, financial condition and results of operations will suffer. In addition, even if our customers acquire new products, services or product enhancements, the revenues from such sales may not be sufficient to offset the significant costs associated with developing and offering such products, services or enhancements to customers. In addition, any announcements of new products, services or enhancements may cause customers to decline or cancel their purchasing decisions in anticipation of replacements.

If we are unable to expand our DIS business, our growth rates could be significantly diminished and our business could be materially harmed.

We plan to grow our DIS business by expanding into new states, adding new hub locations in states in which we currently operate and increasing existing hub utilization by adding physician customers and routes. Our progress in expanding into new geographies has been slower than anticipated, our hub utilization and customer density in some geographies have decreased, and we cannot assure you that we will be able to sell our leasing services at the rates we anticipate, or that physicians or hospitals in these new markets will accept our imaging products or services. Our expansion into additional markets is subject to inherent risks, including those associated with compliance with applicable state laws and regulations. We may find the laws of states in which we do not currently operate preclude us from operating our DIS business, or require us to change the structure in which we operate our DIS business in such states. Our inability to expand into new markets for any of these reasons could diminish our prospects for growth and profitability.

Because our imaging systems and DIS services are not widely diversified, a decrease in sales of our products and leasing services could seriously harm our business.

Our current product and service offerings consist primarily of our line of gamma cameras, including our Cardius-1, Cardius-2, and Cardius-3 XPO Series, 2020tc Imager and SPECTpak PLUS camera systems, all designed for use in the nuclear imaging market segment. In addition, our DIS mobile imaging leasing service utilizes our own line of cameras and at present is focused nearly exclusively on nuclear cardiology. As such, our line of products and services is not as diversified as those of some of our competitors. If the sales of our products or leasing services decline, our business would be seriously harmed, and it would likely be difficult for us to sustain our business while seeking to develop new types of products or services or other markets for our existing products and services. In addition, because our technical know-how and intellectual property have limited applications, we may be unable to leverage these assets to diversify our products and services or to develop other products or sources of revenue outside of the nuclear imaging market.

If we experience problems with the technologies used in our imaging systems or if delivery of our DIS services is delayed, public perception of our products and service offerings could be harmed and we may lose customers and revenue.

We have experienced some reliability issues with our camera detector heads and other parts of our imaging systems, and some of the cameras in our DIS fleet are more than four years old. Although we have embarked on a program to upgrade our fleet over a three year horizon, as the period of use of our cameras increases, other significant defects may occur. Additionally, physicians rely on our DIS services to provide nuclear imaging procedures to their patients on the dates and at the times they have leased. Many factors could prevent us from delivering our DIS services on a timely basis, including equipment failures, unanticipated problems with our mobile Cardius-3XPO camera, weather and the availability of staffing, transportation and necessary supplies. If we are unable to provide physicians or hospitals our DIS services in a timely and efficient manner, our business would be harmed.

We are subject to the financial risks associated with providing services through our DIS business.

There are numerous risks associated with any leasing arrangement, including the possibility that physicians may fail to pay us because of general economic and business conditions, the availability of reimbursement or other reasons. In addition, a number of our DIS customers decide to purchase their own cameras, made by us or by one of our competitors, rather than continue to use our DIS leasing service. If purchases by DIS customers of cameras made by our competitors were to increase, our business and financial condition could be adversely affected. In addition, the number of customers who have canceled or failed to renew their lease contracts recently has increased, and our business may be significantly harmed if this trend continues.

Our manufacturing operations are highly dependent upon third-party suppliers, making us vulnerable to supply problems and price fluctuations, which could harm our business.

We rely on a limited number of third parties to manufacture and supply certain of the key components of our products. While many of the components used in our products are available from multiple sources, we obtain some components from single sources, and alternative sources for them may not be readily available. For example, key components of the detector heads and the processing and control software utilized in our gamma cameras are manufactured or supplied by a single source. If we were unable to obtain these components, our ability to build gamma cameras could be materially affected, and we could experience delays in the production of our gamma cameras for an extended period of time that could cause the loss of customers and revenue.

We have limited marketing, sales and distribution capabilities.

Our future revenue growth will depend in large part on our success in maintaining and expanding our marketing, sales and distribution channels, which is an expensive and time-consuming process. We are highly dependent upon the efforts of our sales force and third-party distributors to increase our revenue. We use three independent distributors in the United States and an independent, international sales distributor to market, sell and distribute our products and services. Our domestic third-party distributors are generally permitted to market, sell and distribute competing imaging products that are used or refurbished and meet specified age requirements. Our international distributors are prohibited from promoting or distributing any other gamma camera product, but not prohibited from offering competing services. We face intense competition for qualified sales employees and may be unable to hire, train, manage and retain such personnel, which could adversely affect our ability to maintain and expand our marketing, sales and distribution network. If we are unable to maintain and expand our direct and third-party marketing, sales and distribution networks, we may be unable to sell enough of our products and imaging services for our business to be profitable and our financial condition and results of operations will likely suffer.

We are exposed to risks relating to product liability, product recalls, property damage and personal injury and death for which insurance coverage is expensive, limited and potentially inadequate, and our business may be negatively affected by insufficiently insured claims and increased insurance costs.

Our operations entail risks relating to claims or litigation relating to product liability, warranty, product recalls, property damage, misdiagnosis, personal injury and death. We may incur significant liability in the event of any such litigation, regardless of the merit of the action. We currently maintain insurance that we believe is adequate with respect to the nature of the risks insured against, including product liability insurance, professional liability insurance, automobile insurance, property insurance, workers compensation insurance and general liability insurance. In many cases such insurance is expensive and difficult to obtain, and no assurance can be given that we will be able to maintain our current insurance or that we will be able to obtain or maintain comparable or additional insurance in the future on reasonable terms, if at all. If we are unable to obtain insurance, or if our insurance is inadequate to cover claims, our cash reserves and other assets could be jeopardized. Additionally, costs associated with maintaining our insurance could become prohibitively expensive, and our ability to become profitable could be diminished.

Our manufacturing operations and executive offices are located at a single facility that may be at risk from fire, earthquakes or other natural or manmade disasters or crises.

Our manufacturing operations and executive offices are located at a single facility in Poway, California, near known fire areas and earthquake fault zones. We have taken precautions to safeguard our facilities, including insurance and health and safety protocols. However, any future natural disaster, such as a fire or an earthquake, could cause substantial delays in our operations, damage to or destroy our manufacturing equipment or inventory, and cause us to incur additional expenses. The insurance we maintain against fires and other natural disasters may not be adequate to cover our losses in any particular case.

Additionally, California has experienced significant electrical power shortages and price volatility in recent years, and if our energy costs substantially increase or blackouts interrupt our power supply frequently or for more than a few days, we may have to reduce or temporarily discontinue our normal operations. Any such reduction or disruption of our operations at our facilities could harm our business.

Risks Related to Government Regulation

We must be licensed to handle and use hazardous materials and may be liable for contamination or other harm caused by hazardous materials that we use.

We use hazardous and radioactive materials in our research and development and manufacturing processes, as well as in the provision of our imaging services. We are subject to federal, state and local laws and regulations governing use, storage, handling and disposal of hazardous and radioactive materials and waste products, or hazardous materials. We are currently licensed to handle such hazardous materials in all states in which we operate, but there can be no assurances that we will be able to maintain those licenses in the future. In addition, we must become licensed to handle hazardous materials in all states into which we plan to expand. Obtaining and maintaining those additional hazardous materials licenses is an expensive and time consuming process. If we are unable to obtain and maintain the requisite licenses, we will not be able to expand into a state and our ability to grow and become profitable will be reduced. Additionally, we cannot completely eliminate the risk of contamination or injury resulting from hazardous materials and we may incur liability as a result of any such contamination or injury. In the event of an accident, we could be held liable for damages or penalized with fines and the liability and associated legal costs could exceed our resources.

We have also incurred and may continue to incur expenses related to compliance with environmental laws. Such future expenses or liability could have a significant negative impact on our business, financial condition and results of operations. Further, we cannot assure you that the cost of complying with these laws and regulations will not materially increase in the future.

We spend considerable time and money complying with federal and state laws, regulations, and other rules, and, if we are unable to comply with such laws, regulations and other rules, we could face substantial penalties.

We are directly, or indirectly through our clients, subject to extensive regulation by both the federal government and the states in which we conduct our business, including the federal Medicare and Medicaid anti-kickback laws; other Medicare laws, regulations, rules, manual provisions, and policies that prescribe the requirements for coverage and payment for services performed by us and our DIS customers; the federal False Claims statutes; the federal Health Insurance Portability and Accountability Act of 1996, or HIPAA; the Stark Law; the federal Food, Drug and Cosmetic Act; federal and state radioactive materials laws; state food and drug and pharmacy laws and regulations; state laws that prohibit the practice of medicine by non-physicians and fee-splitting arrangements between physicians and non-physicians; state scope-of-practice laws that require either specific licenses or certifications for our personnel or their direct supervision by the site physician to perform certain tasks in the absence of such licensures or certifications; federal laws, regulations, rules and policies that permit physicians to bill and receive payment for certain diagnostic tests under the Medicare Physician Fee Schedule only if certain conditions are satisfied, including the requirement that the physician either personally perform, or adequately supervise the performance of, the tests using equipment they own or lease, and that prohibit physicians from marking up the cost of tests they "purchase," rather than perform or supervise, for Medicare patients; and state law equivalents to any of the foregoing federal laws.

We maintain a compliance program to help identify and correct any compliance issues and remain in compliance with all applicable laws, to train employees, to audit and monitor the Company's operations, to provide for a compliance hotline, and to achieve other compliance goals. Like most companies with compliance programs, we occasionally discover compliance concerns. In such cases, we take responsive action, including, when necessary, corrective measures. There can be no assurance that the Company's responsive actions will insulate us from liability associated with any detected compliance concerns.

If our past or present operations are found to be in violation of any of the laws, regulations, rules or policies described above or the other laws or regulations to which we or our customers are subject, we may be subject to civil and criminal penalties, damages, fines, exclusion from federal or state health care programs, or the curtailment or restructuring of our operations. Similarly, if our customers are found to be non-compliant with applicable laws, they may be subject to sanctions, which could have a negative impact on us. If we are excluded from federal or state health care programs, our customers who participate in those programs could not do business with us. In addition, if we are required to obtain permits or licensure that we do not possess, we may become subject to substantial additional regulation or incur significant expense. Any penalties, damages, fines, curtailment or restructuring of our operations would adversely affect our ability to operate our business and our financial results. The risk of our being found in violation of these laws is increased because many of them have not been fully interpreted by the regulatory authorities or the courts, and their provisions are open to many interpretations and additional legal or regulatory change. Any action against us for violation of these laws, even if we successfully defend against it, could cause us to incur significant legal expenses, divert our management's attention from the operation of our business and damage our reputation.

Legislative or regulatory reform of the healthcare system may affect our ability to sell our products.

Federal and state lawmakers from time to time enact new legislation establishing significant changes in the healthcare system. Downward changes to Medicare reimbursement rates for items such as the procedures our physician clients perform or the drugs used in conjunction with them may adversely affect reimbursement to customers or potential customers that use or could use our cameras and services, and may therefore affect us. Effective January 2007, the "technical component" of Medicare reimbursement for nuclear imaging services performed in physicians' offices was capped at the lesser amount of either the Hospital Outpatient Prospective Payment System rates or the Medicare Part B Physician Fee Schedule rates. As a result of this and other reimbursement changes, the average Medicare reimbursement rate for the imaging procedures most commonly performed by our physician clients declined by 8.5% from 2006 to 2007. Other reimbursement reductions remain under consideration, and we cannot predict whether and to what extent implementation of these reductions will be delayed or whether and how the specific reimbursement rates applicable to procedures performed by our physician clients will be affected. If reimbursement reductions increase, sales of our gamma cameras would suffer and we may receive pressure from our customers to terminate or otherwise modify the lease arrangements for our DIS services. Under such circumstances, our business, financial condition and results of operations could be materially adversely affected. In addition, nuclear medicine is a "designated health service" under the federal anti-self-referral laws known as the Stark Law that a physician may not refer to an entity with which the physician or an immediate family member has a financial relationship, unless an exception applies. DIS' physician customers may be able to meet the "in-office ancillary services" exception to the Stark Law if they meet the definition of a "Group Practice" under Stark, personally supervise the individuals performing the nuclear imaging services and bill for them, and if the services are performed in the same building in which the physicians regularly practice medicine. If DIS' customers are unable or unwilling to comply with the Stark Law, utilization rates of our services and products will decline and our business will be harmed. The potential for adoption of healthcare reform proposals on a state-by-state basis could require us to develop state-specific marketing and sales approaches and adversely affect the results of operations.

Regulatory changes could have a negative impact on camera sales to and leases with hospitals desiring to use our cameras and services in their outpatient facilities.

In order for hospitals to receive certain payments for their outpatient facilities as hospital outpatient services, including services that utilize our products, these services must be furnished in a "provider-based" organization or facility, or be covered services furnished "under arrangement" with the hospital. Failure to meet these requirements may result in reduced payments to the hospitals for their services. The Medicare program has published and revised rules establishing criteria for classifying a facility as "provider-based" or a service as furnished "under arrangement." These rules require an analysis of the facts and circumstances surrounding the delivery by a hospital of a particular service, and hospitals that use our products or DIS services in their outpatient facilities will need to determine if they meet the applicable "provider-based" or "under arrangement" requirements. Hospitals that cannot obtain sufficient payments for these services may not purchase a camera from us or enter into arrangements with us for provision of services.

If we fail to comply with various licensure or certification laws, regulations or standards, we may be subject to civil, criminal and/or administrative penalties, which would adversely affect our operations.

All of the states in which we operate require that the imaging technicians operating our cameras be licensed or certified and such licensing and certification requirements are subject to change. Obtaining licenses may take significant time as we expand into additional states or if requirements change. Any lapse in the licensure or certification of our technicians could increase our costs and adversely affect our operations and financial results.

In addition, our DIS services model involves administering and furnishing radiopharmaceuticals and, until recently, pharmacological stress agents, which are regulated as drugs by state and federal agencies, including the FDA and state pharmacy boards. If a state regulatory authority determines that we have operated our business without required permits or licensure, we could be subject to civil, criminal and/or administrative penalties, including the curtailing or halting of our business. In addition, an inability to obtain required licenses or permits where we currently conduct business, or in states where we plan to expand, would require us to modify the business models we can utilize in the affected jurisdictions. In either case, we could incur substantial expense and could encounter substantial operational burdens.

Our products are subject to reporting requirements and recalls even after receiving FDA clearance or approval, which could harm our reputation, business and financial results.

We are subject to medical device reporting regulations that require us to report to the FDA or similar governmental bodies in other countries if our products cause or contribute to a death or serious injury or malfunction in a way that would be reasonably likely to contribute to death or serious injury. In addition, the FDA and similar governmental bodies in other countries have the authority to require the recall of our products in the event of material deficiencies or defects in design or manufacture that could cause adverse health consequences. A government-mandated or voluntary recall by us could occur as a result of component failures, manufacturing errors or design defects, including defects in labeling. Any recall would divert management attention and financial resources and harm our reputation with customers.

If we fail to obtain, or are significantly delayed in obtaining, FDA or foreign regulatory clearances or approvals for future products or product enhancements, or if we or our third party contractors fail to comply with FDA's Quality System Regulation, or the quality regulations of foreign governments, our ability to market and distribute our products will suffer.

Our products are subject to rigorous regulation by the FDA and numerous other federal, state and foreign governmental authorities. In the U.S., the FDA regulates virtually all aspects of a medical device's testing, manufacture, safety, labeling, storage, recordkeeping, reporting, promotion and distribution. Our failure to comply with those regulations could lead to the imposition of administrative or judicial sanctions, including Warning Letters, injunctions, suspensions or the loss of regulatory clearances or approvals, product recalls, termination of distribution, product seizures, injunctions, criminal sanctions or closure of our manufacturing facilities. The process of obtaining regulatory clearances or approvals to market a medical device can be costly and time consuming, and there can be no assurance that such clearances or approvals will be granted on a timely basis, if at all. Commercial distribution of a new medical device generally requires 510(k) clearance or an approved PMA. The FDA will clear marketing of a medical device through the 510(k) process if it is demonstrated that the new product is substantially equivalent to a legally marketed predicate device. The PMA approval process is more costly, lengthy and uncertain than the 510(k) clearance process, and must be supported by extensive data, including data from preclinical studies and human clinical trials. We cannot assure you that we will receive marketing clearance or PMA approval for any of our new products or product enhancements may not occur. While we have not been required to obtain PMA approval for any of our products, we may in the future have to undergo the lengthier, burdensome, and expensive PMA approval process. Further, pursuant to FDA regulations, we can only market our products for cleared or approved uses.

Our manufacturing processes and those of our third-party manufacturers are required to comply with the FDA's Quality System Regulation, which covers the design, testing, production processes, controls, quality assurance, labeling, packaging, storage and shipping of our devices. In addition, we must engage in extensive recordkeeping and reporting and must make available our manufacturing facility and records for periodic unscheduled inspections by federal, state and foreign agencies, including the FDA. Our failure or our third-party manufacturers' failure to pass a Quality System Regulation inspection or to comply with these and other applicable regulatory requirements could result in disruption of our operations and manufacturing delays, and a failure to take adequate corrective action could result in, among other things, Warning Letter(s), withdrawal of our medical device clearances, seizure or recall of our devices, or other civil or criminal enforcement actions.

Foreign governmental authorities that regulate the manufacture and sale of medical devices have become increasingly stringent and, to the extent we now or in the future market and sell our products in foreign countries, we may be subject to rigorous regulation by those foreign governmental authorities. In such circumstances, we would rely significantly on our foreign independent distributors to comply with the varying regulations, and any failures on their part could result in restrictions on the sale of our products in foreign countries.

Modifications to our products may require new 510(k) clearances or pre-market approvals, or may require us to cease marketing or recall the modified products until clearances are obtained.

Any modification to a 510(k)-cleared device that could significantly affect its safety or effectiveness, or that would constitute a major change in its intended use, requires a new 510(k) clearance or, possibly, approval of a PMA. The FDA requires every manufacturer to make this determination in the first instance, but the FDA may review any manufacturer's decision. The FDA may not agree with our decisions regarding whether new clearances or approvals are necessary. If the FDA requires us to seek 510(k) clearance or PMA approval for modification of a previously cleared product for which we have concluded that no clearances or approvals are necessary, we may be required to cease marketing or to recall the modified product until we obtain clearance or approval, and we may be subject to significant regulatory fines or penalties. Further, our products could be subject to recall if the FDA determines, for any reason, that our products are not safe or effective. Any recall or FDA requirement for which we seek additional approvals or clearances could result in delays, fines, costs associated with modification of a product, loss of revenue and potential operating restrictions imposed by the FDA, any of which would harm our business.

Risks Related to Our Financial Results

Our quarterly and annual financial results are difficult to predict and are likely to fluctuate significantly from period to period because our business prospects are uncertain and our DIS leasing services business is seasonal.

Our revenue and results of operations at any given time will be primarily based on the following factors, many of which we cannot control:

- physician, healthcare provider and patient acceptance of our products and services;
- demand for and pricing of our products and services;
- levels of and restrictions upon third-party payor reimbursement for our products and services;
- accreditation and credentialing requirements imposed by third-party payors on physicians and providers of mobile imaging services;
- our ability to retain our DIS customers;
- success and timing of new product offerings, acquisitions, licenses or other significant events by us or our competitors;
- · camera purchases by DIS customers;
- our ability to establish and maintain a productive manufacturing, marketing, sales and distribution force;
- the ability of our suppliers to provide us with an adequate supply of necessary components on a timely basis;
- · timing and magnitude of our expenditures;
- our ability to reduce our expenses quickly enough to respond to any declines in revenue;
- · regulatory approvals and legislative changes affecting the products we may offer or those of our competitors;
- · the effect of competing technological and market developments; and
- interruption in the manufacturing or distribution of our products and services.

Furthermore, we have experienced seasonality in the leasing services offered by DIS. While our physicians are obligated to pay us for all lease days to which they have committed, our contracts permit some flexibility in scheduling when services are to be performed. This accounts for some of the seasonality of our DIS revenues. We cannot predict with certainty the degree to which seasonal circumstances such as the summer slowdown, winter holiday variations and weather conditions may make our revenue unpredictable or lead to fluctuations in our quarterly operating results in the future.

In addition, due to the way that customers in our target markets acquire our products, a large percentage of our orders of cameras is booked during the last month of each quarterly accounting period. As such, a delivery delay of only a few days may significantly impact our quarter-to-quarter comparisons. Moreover, the sales cycle for our cameras is typically lengthy, which may cause us to experience significant revenue fluctuations.

For these reasons, quarterly and annual sales and operating results may vary significantly in the future and period-to-period comparisons of our results of operations are not necessarily meaningful and should not be relied upon as indicators of future performance. We cannot assure you that our sales will increase or be sustained in future periods. We have experienced, and may continue to experience, significant, unanticipated quarterly and annual losses. Because of these and other factors, our operating results in one or more future reporting periods may fail to meet the expectations of securities analysts or investors, which could cause our stock price to decline significantly.

We have incurred significant and recurring operating losses since our inception in 1985 and we may incur such losses and increased operating expenses in the near term.

We have incurred significant cumulative net losses since our inception in November 1985 and may incur such losses and increased operating expenses in the near term as we, among other things, expand our DIS business, increase marketing, sales and distribution of our current products, and conduct research and development to develop next-generation products and to enhance our existing products.

As a result of these activities, we may not be able to become profitable or, if we do, maintain profitability. If our revenue grows more slowly than anticipated, or if our operating expenses exceed our expectations, our ability to achieve our development and expansion goals would be adversely affected.

Risks Related to Our Intellectual Property and Potential Litigation

Our ability to protect our intellectual property and proprietary technology through patents and other means is uncertain.

Our success depends significantly on our ability to protect our proprietary rights to the technologies used in our products. We rely on patent protection, as well as a combination of copyright, trade secret and trademark laws, and nondisclosure, confidentiality and other contractual restrictions to protect our proprietary technology. However, these legal means afford only limited protection and may not adequately protect our rights, be enforceable or permit us to gain or keep any competitive advantage. Our pending U.S. and foreign patent applications, which include claims to material aspects of our products and procedures that are not currently protected by issued patents, may not issue as patents in a form that will be advantageous to us. Any patents we have obtained or do obtain may be challenged by re-examination or otherwise invalidated or eventually found unenforceable. Both the patent application process and the process of managing patent disputes can be time consuming and expensive. Competitors may attempt to challenge or invalidate our patents, or may be able to design alternative techniques or devices that avoid infringement of our patents, or develop products with functionalities that are comparable to ours.

In the event a competitor infringes upon our patent or other intellectual property rights, litigation to enforce our intellectual property rights or to defend our patents against challenge, even if successful, could be expensive and time consuming and could require significant time and attention from our management. We may not have sufficient resources to enforce our intellectual property rights or to defend our patents against challenges from others.

The medical device industry is characterized by patent litigation and we could become subject to litigation that could be costly, result in the diversion of our management's time and efforts, and require us to pay damages.

Any litigation or claims against us, or claims we bring against others, may cause us to incur substantial costs, could place a significant strain on our financial resources, divert the attention of our management from our core business and harm our reputation. If the relevant patents were upheld as valid and enforceable and we were found to be inadvertently infringing, we could be required to pay substantial damages and/or royalties and could be prevented from selling our products unless we could obtain a license or were able to redesign our system to avoid infringement. Any such license may not be available on reasonable terms, if at all. If we fail to obtain any required licenses or make any necessary changes to our products or technologies, we may be unable to commercialize one or more of our products, which could severely harm our business.

We may be subject to damages resulting from claims that we, or our employees, have wrongfully used or disclosed alleged trade secrets of their former employers.

Many of our employees were previously employed at other medical device companies, including our competitors or potential competitors. Although no claims against us are currently pending, we may be subject to claims that we or our employees have inadvertently or otherwise used or disclosed trade secrets or other proprietary information of their former employers. Litigation may be necessary to defend against these claims. Even if we are successful in defending against these claims, litigation could result in substantial costs and be a distraction to management. If we fail in defending such claims, in addition to paying money damages, we may lose valuable intellectual property rights or personnel. A loss of key research personnel or their work product could hinder or preclude our ability to commercialize our products, which could severely harm our business.

Risks Related to the Securities Markets and Ownership of Our Common Stock

Our stock price may be volatile.

The market price for our common stock has been and is likely to continue to be volatile. In addition, the market price of our common stock may fluctuate significantly in response to a number of factors, most of which we cannot control, including:

- volume and timing of orders for our products and services;
- · declining sales of nuclear imaging products and other adverse conditions affecting our target markets;
- the results of delays in introduction of new products, product enhancements, services or technologies by us or our competitors;
- period-to-period variations in our or our competitors' results of operations;
- conditions or trends in the medical device industry and the imaging service industry;
- · disputes or other developments with respect to intellectual property rights, product liability claims or other litigation;
- our ability to develop, obtain regulatory clearance for, and market, new and enhanced products on a timely basis, or changes in governmental regulations or in the status of our regulatory approvals or applications;
- · additions or departures of key personnel;
- sales of large blocks of our common stock, including sales by our executive officers and directors;
- changes in the availability of third-party reimbursement in the United States or other countries;
- · changes in earnings estimates or recommendations by securities analysts; and
- general market conditions and other factors, including factors unrelated to our operating performance or the operating performance of our competitors.

Future sales of our common stock may cause our stock price to decline.

A small number of our current stockholders hold a substantial number of shares of our common stock that they may sell in the public market. Sales by our current stockholders of a substantial number of shares, or the expectation that such sale may occur, could significantly reduce the market price of our common stock. Moreover, the holders of a substantial number of our shares of common stock, including shares issued upon the exercise of certain of our warrants, have rights, subject to some conditions, to require us to file registration statements to permit the resale of their shares in the public market or to include their shares in registration statements that we may file for ourselves or other stockholders. We have also registered all common stock that we may issue under our employee benefit plans. Accordingly, these shares can be freely sold in the public market upon issuance, subject to restrictions under the securities laws and the lock-up agreements described above. If any of these stockholders cause a large number of securities to be sold in the public market, the sales could reduce the trading price of our common stock. These sales also could impede our ability to raise future capital.

Our common stock is thinly traded and an active trading market may not be sustained.

Although we are currently listed for trading on the Nasdaq National Market, an active trading market for our common stock may not be sustained. An inactive market may impair your ability to sell your shares at the time you wish to sell them or at a price that you consider reasonable. Furthermore, an inactive market may impair our ability to raise capital by selling shares and may impair our ability to acquire other businesses, products and technologies by using our shares as consideration.

Anti-takeover provisions in our organizational documents, our Stockholders Rights Plan and Delaware law may prevent or delay removal of current management or a change in control.

Our restated certificate of incorporation and restated bylaws contain provisions that may delay or prevent a change in control, discourage bids at a premium over the market price of our common stock and deversely affect the market price of our common stock and the voting and other rights of the holders of our common stock. These provisions include:

- prohibiting our stockholders from calling a special meeting of stockholders unless they hold not less than 20% of the total number of votes to be cast at such a meeting;
- permitting the issuance of additional shares of up to 10,000,000 shares of preferred stock without stockholder approval upon terms and conditions, and with the rights, preferences and privileges as a board of directors may determine;
- prohibiting our stockholders from making certain changes to our restated certificate of incorporation or restated bylaws except with 66 ²/3% stockholder approval; and
- · requiring advance notice for raising matters of business or making nominations at stockholders' meetings.

The rights issued pursuant to our Stockholder Rights Plan will become exercisable, subject to certain exceptions, the tenth day after a person or group announces acquisition of 15% or more of our common stock or announces commencement of a tender or exchange offer the consummation of which would result in ownership by the person or group of 15% or more of our common stock.

In addition, as a Delaware corporation, we are subject to Delaware law, including Section 203 of the Delaware General Corporation Law. In general, Section 203 prohibits a Delaware corporation from engaging in any business combination with any interested stockholder for a period of three years following the date that the stockholder became an interested stockholder unless certain specific requirements are met as set forth in Section 203.

These provisions alone or together, could have the effect of deterring or delaying changes in incumbent management, proxy contests or changes in control.

If our officers, directors and principal stockholders choose to act together, they may be able to control our management and operations, acting in their best interests and not in the best interests of other stockholders.

Our officers, directors and holders of 5% or more of our outstanding common stock beneficially own a significant amount of our outstanding common stock. As a result, these stockholders, acting together, will be able to significantly influence all matters requiring approval by our stockholders, including the election of directors and the approval of mergers or other business combination transactions. The interests of this group of stockholders may not always coincide with our interests or the interests of other stockholders, and they may act in a manner that advances their best interests and not necessarily those of other stockholders. As a result of their actions or inactions our stock price may decline.

ITEM 2. UNREGISTERED SALES OF EQUITY SECURITIES AND USE OF PROCEEDS

We effected the initial public offering of our common stock pursuant to a Registration Statement on Form S-1 (File No. 333-113760) that was declared effective by the Securities and Exchange Commission on June 9, 2004.

ITEM 3. DEFAULTS UPON SENIOR SECURITIES

Restated Certificate of Incorporation

None.

ITEM 4. SUBMISSION OF MATTERS TO A VOTE OF SECURITY HOLDERS

None.

ITEM 5. OTHER INFORMATION

None.

ITEM 6. EXHIBITS

Description

Exhibit Number

3.1(1)

3.2(2)	Restated Bylaws
4.1(3)	Form of Specimen Stock Certificate
4.2(4)	Amended and Restated Investors' Rights Agreement by and among Digirad Corporation and the investors listed on the schedule attached thereto,
	dated April 23, 2002, as amended
10.1(1)	Digirad Corporation 2004 Stock Incentive Plan as Amended and Restated April 27, 2006
10.2	Asset Purchase Agreement by and between Digirad Corporation, Digirad Imaging Solutions, Inc., Digirad Ultrascan Solutions, Inc. and Ultrascan, Inc.
	dated May 1, 2007†
31.1	Certification of Chief Executive Officer pursuant to Rules 13a-14(a) and 15d-14(a) promulgated pursuant to the Securities Exchange Act of 1934, as
	amended
31.2	Certification of Chief Financial Officer pursuant to Rules 13a-14(a) and 15d-14(a) promulgated pursuant to the Securities Exchange Act of 1934, as
	amended
32.1	Certification of Chief Executive Officer pursuant to Section 906 of the Sarbanes-Oxley Act of 2002
32.2	Certification of Chief Financial Officer pursuant to Section 906 of the Sarbanes-Oxley Act of 2002

⁽¹⁾ Incorporated by reference to the Company's current report on Form 8-K filed with the Securities and Exchange Commission on May 3, 2006.

⁽²⁾ Incorporated by reference to the Company's quarterly report on Form 10-Q filed with the Securities and Exchange Commission on August 11, 2004.

- (3) Incorporated by reference to the Registration Statement on Form S-1 (File No. 333-113760) originally filed with the Securities and Exchange Commission on March 19, 2004, as amended thereafter.
- (4) Incorporated by reference to the Company's quarterly report on Form 10-Q filed with the Securities and Exchange Commission on November 2, 2004.

SIGNATURES

Pursuant to the requirements of the Securities and Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

DIGIRAD CORPORATION

Date: May 7, 2007

Date: May 7, 2007

By: /s/ MARK L. CASNER

Mark L. Casner

President and Chief Executive Officer (Principal Executive Officer)

By: /s/ TODD P. CLYDE

Todd P. Clyde

Chief Financial Officer

(Principal Financial and Accounting Officer)

INDEX TO EXHIBITS

Exhibit

Description

3.1(1)	Restated Certificate of Incorporation
3.2(2)	Restated Bylaws
4.1(3)	Form of Specimen Stock Certificate
4.2(4)	Amended and Restated Investors' Rights Agreement by and among Digirad Corporation and the investors listed on the schedule attached thereto,
	dated April 23, 2002, as amended
10.1(1)	Digirad Corporation 2004 Stock Incentive Plan as Amended and Restated April 27, 2006
10.2	Asset Purchase Agreement by and between Digirad Corporation, Digirad Imaging Solutions, Inc., Digirad Ultrascan Solutions, Inc. and Ultrascan, Inc.
	dated May 1, 2007†
31.1	Certification of Chief Executive Officer pursuant to Rules 13a-14(a) and 15d-14(a) promulgated pursuant to the Securities Exchange Act of 1934, as
	amended
31.2	Certification of Chief Financial Officer pursuant to Rules 13a-14(a) and 15d-14(a) promulgated pursuant to the Securities Exchange Act of 1934, as
	amended
32.1	Certification of Chief Executive Officer pursuant to Section 906 of the Sarbanes-Oxley Act of 2002
32.2	Certification of Chief Financial Officer pursuant to Section 906 of the Sarbanes-Oxley Act of 2002

- (1) Incorporated by reference to the Company's current report on Form 8-K filed with the Securities and Exchange Commission on May 3, 2006.
- (2) Incorporated by reference to the Company's quarterly report on Form 10-Q filed with the Securities and Exchange Commission on August 11, 2004.
- (3) Incorporated by reference to the Registration Statement on Form S-1 (File No. 333-113760) originally filed with the Securities and Exchange Commission on March 19, 2004, as amended thereafter.
- (4) Incorporated by reference to the Company's quarterly report on Form 10-Q filed with the Securities and Exchange Commission on November 2, 2004.
- † Portions of this exhibit (indicated by asterisks) have been omitted pursuant to a request for confidential treatment and have been separately filed with the Commission.

*** Portions of this Exhibit (indicated by asterisks) have been
Omitted Pursuant to a Request for Confidential Treatment and have been separately filed with the Commission.

ASSET PURCHASE AGREEMENT

by and between

DIGIRAD CORPORATION
DIGIRAD IMAGING SOLUTIONS, INC.
DIGIRAD ULTRASCAN SOLUTIONS, INC.

and

ULTRASCAN, INC.

WILLIAM "MICKEY" KING, JR.

May 1, 2007

TABLE OF CONTENTS

ARTICLE	I. DEFINITIONS	<u>Page</u> 2
1.1	Defined Terms	2
ARTICLE	II. PURCHASE AND SALE OF ASSETS	15
2.1	Transfer of Assets	15
2.2	Excluded Assets	17
2.3	No Assumption of Liabilities	17
2.4	Purchase Price	17
2.5	Closing Costs; Transfer Taxes; Fees and Expenses	18
2.6	Purchase Price Adjustment	19
2.7	Closing	21
ARTICLE	III. REPRESENTATIONS AND WARRANTIES OF THE SELLER AND PRINCIPAL STOCKHOLDER	22
3.1	Organization of the Seller	22
3.2	Authorization	22
3.3	Title to Purchased Assets; Condition of Purchased Assets	22
3.4	Absence of Certain Changes or Events	23
3.5	Material Contracts	24
3.6	Facilities	26
3.7	Permits	27
3.8	No Conflict or Violation	27
3.9	Consents and Approvals	28
3.10	Financial Information	28
3.11	Books and Records	29
3.12	Litigation	29
3.13	Labor Matters	29
3.14	Liabilities	30
3.15	Compliance with Law	30
3.16	No Brokers	31
3.17	Intellectual Property	31
3.18	Employee Plans	33
3.19	Transactions with Certain Persons	35
3.20	Tax Matters	35
3.21	Liabilities from Business Combination	36
3.22	Insurance	36
3.23	Compliance With Environmental Laws	37
3.24		37
3.25	Customers and Distributors	38

TABLE OF CONTENTS (Continued)

3.26	Sufficiency of Purchased Assets	38
3.27	WARN Act	38
3.28	Foreign Corrupt Practices Act	38
3.29	Other Agreements	38
3.30	Bulk Transfer Laws	39
3.31	Redemption of Seller Shares by Seller	39
3.32	Material Misstatements or Omissions	39
ARTICLE I	V. REPRESENTATIONS AND WARRANTIES OF THE PURCHASER	39
4.1	Organization	39
4.2	Authorization	39
4.3	No Conflict or Violation	40
4.4	Consents and Approvals	40
4.5	SEC Filings	40
4.6	No Brokers	40
ARTICLE V	7. COVENANTS AND AGREEMENTS	40
5.1	Omitted	40
5.2	Access and Investigation	41
5.3	Stock Registration	41
5.4	Filings and Consents	42
5.5	Reasonable Efforts	42
5.6	Certain Tax Matters	42
5.7	Employee Matters	43
5.8	Cessation of Use of Names and Web Site	44
5.9	Non-Competition/Non-Solicitation	45
5.10	Estoppel Certificates	45
5.11	Delivery of Title Commitment and Survey	45
5.12	Delivery of Title to Leased Equipment	46
5.13	*****	46
5.14	*****	46
5.15	Tax Disclosure Agreement	46
5.16	Further Assurances	46
ARTICLE V	I. CONDITIONS PRECEDENT TO OBLIGATIONS OF THE PURCHASER	46
6.1	Conditions	46
ARTICI E V	II. CONDITIONS PRECEDENT TO OBLIGATIONS OF THE SELLER	47
ANTICLE V	II. CONDITIONS INECEDENT TO ODEIGATIONS OF THE SELLEN	4,
7.1	Conditions	48
ARTICLE V	VIII. CLOSING	48

48

8.1

Deliveries at Closing

TABLE OF CONTENTS (Continued)

ARTICLE IX	X. INDEMNIFICATION	49
9.1	Survival of Representations and Warranties and Indemnification Obligations	49
9.2	Indemnification	50
9.3	Notice of Claims	51
9.4	Resolution of Notice of Claim	52
9.5	Release of Closing Escrow Fund	53
9.6	Third-Party Claims	54
9.7	Limitation on Indemnity; Remedies; Maximum Liability	54
ARTICLE X.	TERMINATION	55
10.1	Termination	55
ARTICLE X	I. MISCELLANEOUS	56
11.1	Assignment	56
11.2	Notices	56
11.3	Choice of Law	57
11.4	Entire Agreement; Amendments and Waivers; Interpretation	57
11.5	Counterparts	58
11.6	Severability	58
11.7	Headings	58
11.8	Expenses	58
11.9	Schedules	58
11.10	Publicity	58
11.11	No Third-Party Beneficiaries	58
11.12	Remedies Cumulative; Specific Performance	58
11.13	Further Assurances	59

TABLE OF CONTENTS (Continued)

EXHIBIT A-1	FORM OF PROTECTIVE COVENANT AGREEMENT
Ехнівіт В	FORM OF EMPLOYMENT AGREEMENT
Ехнівіт С	ASSIGNMENT AND ASSUMPTION AGREEMENT
EXHIBIT D	FORM OF NON-COMPETITION AND NON-SOLICITATION AGREEMENT
EXHIBIT E	BILL OF SALE
EXHIBIT F	EXCLUDED ASSETS
Eхнівіт G	ASSUMED LIABILITIES
EXHIBIT G-1	ASSUMED OPERATIONAL LEASE OBLIGATIONS
Ехнівіт Н	ESCROW AGREEMENT
EXHIBIT I	WORKING CAPITAL
EXHIBIT J	EBITDA ALLOCATIONS
Ехнівіт К	MOLCHAN SETTLEMENT AND RELEASE AGREEMENT
EXHIBIT L	CONSENT OF HILIE KING

FORM OF PRINCIPAL STOCKHOLDER EMPLOYMENT AGREEMENT

Ехнівіт А

ASSET PURCHASE AGREEMENT

This Asset Purchase Agreement (this "Agreement"), effective as of May 1, 2007, is entered into by and among Ultrascan, Inc., a Georgia corporation (the "Seller") and William "Mickey" King, Jr. (the "Principal Stockholder"), on the one hand, and Digirad Corporation, a Delaware corporation (the "Parent"), Digirad Imaging Solutions, Inc., a Delaware corporation ("DIS") and Digirad Ultrascan Solutions, Inc., a newly formed Delaware corporation (the "Purchaser"), on the other hand.

RECITALS

WHEREAS, the Seller owns and operates that certain medical imaging services business operating under the trade name, "Ultrascan" located in Suwanee, Georgia, that provides ultrasound and nuclear imaging services and other services to physicians (the "Business");

WHEREAS, the Purchaser desires to purchase or license from the Seller, and the Seller desires to sell or license to the Purchaser, all or substantially all of the assets, properties, rights and claims of, or related to, the Business on the terms and conditions set forth herein;

WHEREAS, concurrently with the execution and delivery of this Agreement, and as a material inducement to Parent, DIS and Purchaser to enter into this Agreement, (i) the Principal Stockholder is entering into an Employment Agreement in substantially the form attached hereto as Exhibit A (the "Principal Stockholder Employment Agreement") and a Protective Covenant Agreement in substantially the form attached hereto as Exhibit A-1, and (ii) Matthew G. Molchan is entering into an Employment Agreement in substantially the form attached hereto as Exhibit B (the "Employment Agreement"); and

NOW THEREFORE, in consideration of the foregoing recitals and the mutual representations, warranties, covenants and agreements contained herein and for other good and valuable consideration, the receipt and adequacy of which are hereby acknowledged, the parties hereto agree as follows:

AGREEMENT

ARTICLE I. DEFINITIONS

- 1.1 <u>Defined Terms</u>. As used herein, the terms below shall have the following meanings. Any of such terms, unless the context otherwise requires, may be used in the singular or plural, depending upon the reference.
 - "Accounts Receivable" shall have the meaning set forth in Section 3.10(b) of this Agreement.
- "Action" shall mean any action, claim, suit, litigation, proceeding, labor dispute, arbitral action, governmental audit, inquiry, criminal prosecution, investigation or unfair labor practice charge or complaint.
 - "Adjustment Amount" shall have the meaning set forth in Section 2.6(f) of this Agreement.
- "Affiliate" shall have the meaning set forth in the Securities Exchange Act of 1934, as amended, and the rules and regulations promulgated thereunder.
 - "Agreement" shall have the meaning set forth in the preamble to this Agreement.
 - "Allocation" shall have the meaning set forth in Section 2.4(d) of this Agreement.
- "Ancillary Agreements" shall mean the Assignment and Assumption Agreement, the Bill of Sale, the Escrow Agreement, the Employment Agreements, the Protective Covenant Agreement and Non-Competition and Non-Solicitation Agreements.
 - "Arbitration Firm" shall mean JAMS located in San Diego, California or an American Arbitration Association organization in San Diego, California.
- "Assignment and Assumption Agreement" shall mean the Assignment and Assumption Agreement dated the Closing Date between the Seller and the Purchaser, substantially in the form of Exhibit C hereto.
 - "Assumed Liabilities" shall have the meaning set forth in Section 2.3 of this Agreement.
- "<u>Assumed Operational Lease Obligations</u>" shall mean those operational lease obligations of the Seller as set forth on <u>Exhibit G-1</u> to be assumed by the Purchaser at Closing.
- "Average Price" shall mean as of any measurement date the average of the closing bid price of one share of Common Stock for the five business days immediately preceding the

measurement date, as quoted on the NASDAQ stock exchange, or, in the event the Common Stock is not listed on the NASDAQ stock exchange at the measurement date, such other exchange or quotation system where the Common Stock is then listed or quoted, as applicable. In the event that the Common Stock is no longer publicly listed at the measurement date, then the Average Price shall mean the fair market value of one share of Common Stock as determined in good faith by Parent's Board of Directors.

- "Backlog" shall have the meaning set forth in Section 3.10(d) of this Agreement.
- "Benchmark" shall have the meaning set forth in Section 2.6(c) of this Agreement.
- "Bill of Sale" shall have the meaning set forth in Section 2.1 of this Agreement.
- "Bulk Transfer Law" shall mean the bulk transfer provisions of the Uniform Commercial Code (or any similar Regulation).
- "Business Day" shall mean any day other than a Saturday, a Sunday or a day on which banking institutions in San Diego, California are authorized or obligated by any applicable Regulation to be closed.
 - "Business Employee" shall mean those employees and officers of the Seller engaged in performing services related to the Business.
 - "Business" shall have the meaning set forth in the recitals to this Agreement.
 - "Cash Consideration" shall mean the sum of (x) the Closing Payment, plus (y) the Closing Escrow Amount, plus (z) the ****** Escrow Amount.
 - "Closing" shall have the meaning set forth in Section 2.7 of this Agreement.
 - "Closing Date" shall have the meaning set forth in Section 2.7 of this Agreement.
 - "Closing Escrow Amount" shall have the meaning set forth in Section 2.4(b) of this Agreement.
- "Closing Escrow Fund" shall mean the fund established pursuant to Section 2.4(b) into which the Closing Escrow Amount and the ****** Escrow Amount are to be deposited.
 - "Closing Payment" shall have the meaning set forth in Section 2.4(a) of this Agreement.
 - "Closing" shall have the meaning set forth in Section 2.7 of this Agreement.
 - "COBRA" shall have the meaning set forth in Section 3.18(d) of this Agreement.
 - "Code" shall mean the Internal Revenue Code of 1986, as amended.

- "Commercial Debt" shall have the meaning set forth in Section 2.3 of this Agreement.
- "Common Stock" shall mean restricted common stock, par value \$.001 per share, of Parent.
- "Confidentiality Agreement" shall mean the Mutual Confidential Disclosure Agreement between Parent and Seller dated as of December 8, 2006.
- "Consents" shall mean any and all consents, assignments, approvals, authorizations, registrations, filings with or waivers of any public, governmental or regulatory body or authority or from parties to Contracts that are (a) required for the consummation of the transactions contemplated by this Agreement and the Ancillary Agreements or (b) necessary or desirable in order that the Purchaser can conduct the Business after the Closing Date substantially in the same manner as the Business was conducted by the Seller before the Closing Date.
 - "Contested Claim" shall have the meaning set forth in Section 9.4(b) of this Agreement.
 - "Contract Rights" shall have the meaning set forth in Section 2.1(b) of this Agreement.
- "Contract" shall mean any agreement, contract, note, loan, evidence of indebtedness, Lease, purchase order, letter of credit, indenture, security or pledge agreement, undertaking, practice, covenant not to compete, employment agreement, severance agreement, license, instrument, obligation or commitment related to the Business to which the Seller is a party or is bound, whether oral or written.
- "Court Order" shall mean any judgment, decision, consent decree, injunction, ruling or order of any federal, state or local court or Governmental Body that is binding on any Person or its property under applicable Regulations.
 - "Damages" shall have the meaning set forth in Section 9.2(a) of this Agreement.
- "<u>Default</u>" shall mean (a) any actual breach or default, (b) the occurrence of an event that with the passage of time or the giving of notice or both would constitute a breach or default or (c) the occurrence of an event that with or without the passage of time or the giving of notice or both would give rise to a right of termination, renegotiation or acceleration.
 - "Determination Date" shall have the meaning set forth in Section 2.6(e) of this Agreement.
 - "DIS" shall have the meaning set forth in the preamble of this Agreement.

- "<u>Disclosure Schedule</u>" shall mean a schedule executed and delivered by the Seller to the Purchaser as of the date hereof which sets forth the exceptions to the representations and warranties contained in Article III of this Agreement and certain other information called for by this Agreement.
- "Earnout Payment," means collectively the First Earnout Payment, Second Earnout Payment, Third Earnout Payment, Fourth Earnout Payment and Fifth Earnout Payment.
 - "Earnout Shares" shall have the meaning set forth in Section 5.3.
- "EBITDA" shall mean the aggregate net revenues including, but not limited to adjustments and offsets from all sources for the applicable period minus expenses (excluding taxes, interest, depreciation and amortization) of Purchaser for the applicable period, calculated in accordance with GAAP as then currently applied by Parent using complete calendar months in defining any 12-month period or three-month period. In calculating Purchaser's EBITDA for any period, **********. In addition, in calculating Purchaser's EBITDA for any period, the ***************************** by the Chief Operating Officer of the Seller and the Chief Financial Officer of Parent.
- "*****" shall mean (i) the First Amendment to Leasing Agreement between *****. and Ultrascan, Inc., amending the Leasing Agreement between these parties dated July 1, 2002, (ii) First Amendment to Leasing Agreement between ***** and Ultrascan, Inc., amending the Leasing Agreement between these parties dated March 5, 1999, and (iii) The Technology and Clerical Billing Services Agreement between ***** and Ultrascan, Inc.
- "****** Escrow Amount" shall have the meaning set forth in Section 2.4(b). This one above goes after "****" shall mean (i) the First Amendment to Leasing Agreement between *****. and Ultrascan, Inc., amending the Leasing Agreement between these parties dated July 1, 2002, (ii) First Amendment to Leasing Agreement between ***** and Ultrascan, Inc., amending the Leasing Agreement between these parties dated March 5, 1999, and (iii) The Technology and Clerical Billing Services Agreement between ***** and Ultrascan, Inc.
- "*******" shall mean a written consent executed by ***** to permit the Seller to assign the Leasing Agreement between the parties dated July 1, 2002 to the Purchaser.
 - "Employee Loans" shall mean any loans to employees or non-employees with an outstanding balance as of the Closing Date.
 - "Employee Offers" shall have the meaning set forth in Section 5.7 of this Agreement.
- "Employee Plan" shall mean any "employee pension benefit plan" as defined in Section 3(2) ERISA, any "employee welfare benefit plan" as defined in Section 3(1) of ERISA, and any other benefit arrangement that is neither an employee pension benefit plan nor an employee welfare benefit plan, within the meaning of ERISA, including any employment, consulting, severance or other similar contract, arrangement or policy and each plan, arrangement (written or oral), program, agreement or commitment providing for insurance coverage (including without limitation any self-insured arrangements), workers' compensation, disability benefits, supplemental unemployment benefits, vacation benefits, fringe benefits, retirement benefits, life, health, disability or accident benefits or for deferred compensation, profit-sharing bonuses, stock options, stock appreciation rights, stock purchases or other equity compensation benefits or other forms of incentive compensation, termination allowance or post-retirement insurance sponsored, maintained or contributed to by the Seller or any ERISA Affiliate with respect to the Business.

"Encumbrance" shall mean any claim, lien, pledge, option, charge, easement, security interest, deed of trust, mortgage, right-of-way, encroachment, building or use restriction, conditional sales agreement, encumbrance or other right of third parties, whether voluntarily incurred or arising by operation of law, and includes, without limitation, any agreement to give any of the foregoing in the future, and any contingent sale or other title retention agreement or lease in the nature thereof.

"Environmental Claim" means any claim, action, litigation, notice of violation, cause of action, consent order, consent decree, investigation, written notice, penalties, expenses, losses or damages (including property damages and natural resource damages) related to any potential Liability arising out of, based on or resulting from (a) the presence or Release (of any Hazardous Substances in, on, from or under any of the Operating Sites; (b) requirements imposed by any Environmental Law, including costs of investigation or remediation or costs incurred in obtaining applicable Permits or complying with any Environmental Law; or (c) any violation or alleged violation of any Environmental Law.

"Environmental Exposure Claim" means any third party lawsuits, claims, actions, litigation, class action lawsuits, proceedings or demands asserted or instituted by any Person alleging bodily injury, adverse health effects or death, or seeking medical monitoring, arising out of or related to actual or alleged exposure of any Person to any Hazardous Substance, radioactive materials, asbestos or asbestos containing materials actually or allegedly (a) present at any Operating Site or (b) contained in any product or part manufactured, sold or distributed by Seller or any of its predecessors (or Affiliate of the Seller or any of its predecessors) on or prior to the Closing Date.

"Environmental Law" shall mean any and all Regulations relating to occupational safety and health; the protection of the environment or human health; the procurement, handling, use and disposal of any radioactive materials; emissions, discharges or Releases of Hazardous Substances, including radioactive materials, into the environment, including, without limitation, ambient air, surface water, groundwater or land; or otherwise relating to the handling of Hazardous Substances or the investigation, clean-up or other remediation or analysis thereof.

"ERISA Affiliate" shall mean each other Person that, together with the Seller as of the relevant measuring date under ERISA, is required to be treated as a single employer under Section 414 of the Code.

"ERISA" shall mean the Employee Retirement Income Security Act of 1974, as amended, and the rules, regulations and interpretations promulgated thereunder.

"Escrow Agent" shall mean LaSalle Bank National Association or another institution reasonably acceptable to Parent and Seller.

"Escrow Agreement" shall have the meaning set forth in Section 2.4(b).

- "Estimated Working Capital" shall have the meaning set forth in Section 2.6(a) of this Agreement.
- "Excluded Assets" shall have the meaning set forth in Section 2.2 of this Agreement.
- "Expiration Date" shall have the meaning set forth in Section 9.1(a) of this Agreement.
- "<u>Facilities</u>" shall mean all offices, manufacturing facilities, stores, warehouses, retail establishments, improvements, administration buildings, and all other facilities owned, leased or used by the Seller in connection with the Business.
- "<u>Fifth Earnout Milestone</u>" refers to Purchaser's achieving EBITDA of at least \$***** during the Fifth Earnout Period, after deducting the aggregate amount (if any) paid to Seller as of December 31, 2011 in respect of the Earnout Payments.
- "<u>Fifth Earnout Payment</u>" shall mean additional consideration in an amount equal to the lesser of (i) \$*******, minus ********, and (ii) the amount by which Purchaser's EBITDA during the Fifth Earnout Period exceeds \$******* after deduction of the aggregate amount (if any) paid to Seller during such period in respect of *******. The Fifth Earnout Payment shall be payable in cash and Common Stock, as follows: (i) cash in an amount equal to ********; and (ii) a number of shares of Common Stock equal to (x) **********, divided by (y) the then applicable Average Price.
 - "Fifth Earnout Period" shall mean the period commencing on the Closing Date and ending on April 30, 2011.
 - "Final Working Capital" shall have the meaning set forth in Section 2.6(d) of this Agreement.
 - "Financial Statements" shall have the meaning set forth in Section 3.10(a) of this Agreement.
- "<u>First Earnout Milestones</u>" refers collectively to the following: (i) Purchaser's Total Revenue equals or exceeds \$****** during the First Earnout Period; (ii) Purchaser's EBITDA equals or exceeds \$****** during the First Earnout Period; (iii) Purchaser enters into ******* during the First Earnout Period; and (iv) Purchaser *******
- "First Earnout Payment" shall mean (i) additional cash consideration of \$******, plus (ii) a number of shares of Common Stock equal to (x) \$*******, divided by (y) the then applicable Average Price.
 - "First Earnout Period" shall mean the 12 month period commencing on the Closing Date.
- "Fourth Earnout Milestone" refers to Purchaser achieving EBITDA of at least \$****** during the Fourth Earnout Period, after deducting the aggregate amount (if any) paid to Seller as of December 31, 2010 in respect of the Earnout Payments.

"Fourth Earnout Payment" shall mean additional consideration in an amount equal to the lesser of (i) \$******, minus the aggregate amount (if any) paid to Seller in respect of the Earnout Payments, and (ii) the amount by which Purchaser's ****** exceeds \$****** after deduction of *****. The Fourth Earnout Payment shall be payable in cash and Common Stock, as follows: (i) cash in an amount equal to *******; and (ii) a number of shares of Common Stock equal to (x) *******, divided by (y) the then applicable Average Price.

"Fourth Earnout Period" shall mean the period commencing on the Closing Date and ending on December 31, 2010.

"GAAP" shall mean generally accepted accounting principles as applied in the United States.

"Governmental Body" shall mean any: (a) nation, principality, state, commonwealth, province, territory, county, municipality, district or other jurisdiction of any nature; (b) federal, state, local, municipal, foreign or other government; (c) governmental or quasi-governmental authority of any nature (including any governmental division, subdivision, department, agency, bureau, branch, office, commission, council, board, instrumentality, officer, official, representative, organization, unit, body or entity and any court or other tribunal); (d) multi-national organization or body; or (e) individual, entity or body exercising, or entitled to exercise, any executive, legislative, judicial, administrative, regulatory, police, military or taxing authority or power of any nature.

"Governmental Order" shall have the meaning ascribed to it in Section 3.15(b) of this Agreement.

"Hazardous Substances" means any materials, substances or wastes defined by, or regulated under, any Environmental Law as a hazardous waste, hazardous material, hazardous substance, extremely hazardous waste, restricted hazardous waste, contaminant, pollutant, toxic waste, or toxic substance, including, without limitation, petroleum, petroleum products, asbestos, urea formaldehyde, radioactive materials, polychlorinated biphenyls ("PCBs") and toxic mold.

"Healthcare Laws" shall mean, without limitation, the administrative simplification provisions of the Health Insurance Portability and Accountability Act of 1996 (42 U.S.C. §§ 1320d-1320d-8), the Medicare Prescription Drug, Improvement and Modernization Act of 2003, Medicare (Title XVIII of the Social Security Act), Medicare rules governing the operation of independent diagnostic testing facilities at 42 C.F.R. § 410.33, Medicare rules governing the supervision of diagnostic testing, all Medicare standards for reimbursement of diagnostic tests, the prohibition on the mark-up of purchased diagnostic test, and all other Medicare, Medicaid, and other governmental program guidance, the regulations promulgated pursuant to such laws, and any other state or federal law, regulation, guidance document, manual provision, program memorandum, opinion letter, or other issuance which regulates patient or program charges, recordkeeping, referrals, the hiring of employees or acquisition of services or supplies from those who have been excluded from government health care programs, quality, safety, privacy, security, licensure, accreditation, or provider credentialing, and any governmental orders or authorizations related thereto.

- "Hired Employees" shall have the meaning set forth in Section 5.7(a) of this Agreement.
- "Indemnified Party" shall have the meaning set forth in Section 9.3(a) of this Agreement.
- "Indemnifying Party" shall have the meaning set forth in Section 9.3(a) of this Agreement.

"Intellectual Property Rights" shall mean all (a) United States and foreign patents and patent applications and disclosures relating thereto (and any patents that issue as a result of those patent applications), and any renewals, reissues, reexaminations, extensions, continuations, continuations-in-part, divisions and substitutions relating to any of the patents and patent applications, as well as all related foreign patent and patent applications that are counterparts to such patents and patent applications; (b) United States and foreign trademarks, service marks, trade dress, logos, trade names and corporate names, whether registered or unregistered, and the goodwill associated therewith, together withy any registrations and applications for registration therefore; (c) United States and foreign copyrights under copyrights, whether registered or unregistered, including moral rights, and any registrations and applications for registration therefore; (d) United States and foreign mask work rights and registrations and applications for registration thereof; (e) rights in databases and data collections (including knowledge databases, customer lists and customer databases) under the laws of the United States or any other jurisdiction, whether registered or unregistered, and any applications for registration therefor; (f) trade secrets and other rights in know-how and confidential or proprietary information (including any business plans, designs, technical data, customer data, financial information, pricing and cost information, bills of material, or other similar information); (g) URL and domain name registrations; (h) inventions (whether or not patentable) and improvements thereto; (i) all claims and causes of action arising out of or related to infringement or misappropriation of any of the foregoing and (j) other proprietary or intellectual property rights now known or hereafter recognized in any jurisdiction.

"Inventory." shall mean all of the Seller's inventory held for sale or otherwise in connection with the operation of the Business, and all of the Seller's raw materials, work in process, finished products, supply and packaging items and similar items, with respect to the Business, in each case wherever the same may be located.

"IRS" shall have the meaning set forth in Section 3.18(f) of this Agreement.

"Kickback Laws" shall mean, without limitation, the federal Anti-Kickback Statute (42 U.S.C. § 1320a-7b(b)), the Stark Law (42 U.S.C. § 1395nn), the Anti-Inducement Law (42 U.S.C. § 1320a-7a(a)(5)), the civil False Claims Act (31 U.S.C. § 3729 et seq.), the administrative False Claims Law (42 U.S.C. § 1320a-7b(a)), the exclusion laws (42 U.S.C. § 1320a-7), and the civil monetary penalty laws (42 U.S.C. § 1320a-7a), any state law that regulates kickbacks or physician self-referral and the regulations promulgated pursuant to such laws.

- "Knowledge" shall mean, with respect to any party, the actual knowledge of such party's officer, director or key employee with respect to any fact, circumstance, event or other matter in question, after reasonable inquiry by such party.
 - "Leased Equipment" means those equipment leases of the Seller currently in effect as of the Closing Date.
 - "Leased Real Property" shall mean all leased property described in the Leases related to Facilities.
 - "Leasehold Estates" shall mean all of the Seller's rights and obligations under the Leases.
 - "Leasehold Improvements" shall mean all leasehold improvements situated in or on the Leased Real Property and owned by the Seller.
- "<u>Leases</u>" shall mean all of the existing leases, subleases, occupancy agreements, options, rights, concessions or other agreements or arrangements with respect to the personal or real property of the Seller related to the Business.
 - "Liens" shall have the meaning as set forth in Section 3.20 of this Agreement.
- "<u>Liabilities</u>" shall mean any direct or indirect liability, indebtedness, obligation, commitment, expense, claim, deficiency, guaranty or endorsement of or by any Person of any type, known or unknown, and whether accrued, absolute, contingent, matured, unmatured or other.
 - "March Balance Sheet" shall have the meaning set forth in Section 3.10(b) of this Agreement.
 - "March Inventory" shall have the meaning set forth in Section 3.10(c) of this Agreement.
- "Material Adverse Effect" or "Material Adverse Change" shall mean any event, change or effect that, when taken individually or together with all other adverse events, changes or effects has had, or reasonably would be expected to have, a material adverse effect or change on the Seller, the Business, the Purchased Assets or on the ability of the Seller to consummate the transactions contemplated hereby; provided, however, that the terms "Material Adverse Effect" and "Material Adverse Change" shall not include any such effect resulting from general economic or industry conditions that did not disproportionately affect the Seller, the Business or the Purchased Assets or any occurrence or condition arising out of the transactions contemplated by this Agreement or the public announcement thereof.
 - "Material Contracts" shall have the meaning set forth in Section 3.5(b) of this Agreement.

- "Multiemployer Plan" shall have the meaning set forth in Section 3.18(c) of this Agreement.
- "Notice of Claim" shall have the meaning set forth in Section 9.1(b) of this Agreement.
- "Notice of Disagreement" shall have the meaning set forth in Section 2.6(e) of this Agreement.
- "Operating Site" shall have the meaning set forth in the definition of Pre-Closing Environmental Matters.
- "Ordinary Course of Business" or "Ordinary Course" or any similar phrase shall mean the ordinary course of the Business, consistent with the Seller's past practice.
 - "Parent" shall have the meaning set forth in the preamble of this Agreement.
 - "PCBs" shall have the meaning set forth in the definition of Hazardous Substances.
- "Permits" shall mean all licenses, permits, franchises, approvals, authorizations, consents or orders of, or filings with, any Governmental Body, whether foreign, federal, state or local, or any other Person, necessary or desirable for the past, present or anticipated conduct of, or relating to the operation of, the Business.
- "Person" shall mean any person or entity, whether an individual, trustee, corporation, partnership, limited partnership, limited liability company, trust, unincorporated organization, business association, firm, joint venture, or Governmental Body.
 - "Post-Closing Statement" shall have the meaning set forth in Section 2.6(d) of this Agreement.
- "Pre-Closing Environmental Matters" shall mean (a) the presence as of or prior to the Closing Date of Hazardous Substances in, on or under any Real Property or Facilities conveyed by Seller to Purchaser pursuant to this Agreement (an "Operating Site"); (b) the production, use, generation, storage, treatment, transportation, recycling, Release or other handling or disposition of any kind at any time on or prior to the Closing Date of any Hazardous Substances, either in, on, from or under any Operating Site; (c) any Environmental Exposure Claim or other Environmental Claim against the Seller or Purchaser relating to or arising out of any act, omission or occurrence taken by or on behalf of the Seller or any of its predecessors (or Affiliate of the Seller or any of its predecessors), or otherwise occurring, on or prior to the Closing Date; (d) any failure on or prior to the Closing Date of the Seller or any Operating Site to be in compliance with any Environmental Law; (e) any Environmental Claim or Liability pursuant to Environmental Law at or otherwise related to any third party location to which the Seller sent, or caused to be sent, Hazardous Materials or to any real property currently or formerly owned or leased by or used, by the Seller but that is not an Operating Site; and (f) any other act, omission or occurrence of the Seller or any of its

predecessors (or Affiliate of the Seller or any of its predecessors) prior to the Closing Date that gives rise to Liability or potential Liability to the Purchaser under any Environmental Law, Environmental Claim or Environmental Exposure Claim.

"Pre-Closing Healthcare Regulatory Matters" shall mean, without limitation: (i) any violation of, or non-compliance with, Healthcare Laws, Kickback Laws and any regulations related thereto; and (ii) any other act, omission or occurrence of the Seller or any of its predecessors (or Affiliate of the Seller or any of its predecessors), prior to or on the Closing Date that give rise to Liability or potential Liability to the Purchaser under any Healthcare Laws or Kickback Laws.

- "Pre-Closing Tax Periods" shall mean those periods of taxation prior to the Closing Date.
- "Preliminary Statement" shall have the meaning set forth in Section 2.6(a) of this Agreement.
- "Principal Stockholder" shall have the meaning set forth in the preamble of this Agreement.
- "Protective Covenant Agreement" shall have the meaning ascribed to it as set forth in Section 5.8 of this Agreement.
- "<u>Purchase Price</u>" shall mean the sum of (x) the Cash Consideration, as adjusted pursuant to the Escrow Agreement and Sections 2.4 and 2.6 of this Agreement, plus (y) the aggregate Earnout Payments payable to Seller pursuant to this Agreement.
 - "Purchased Assets" shall have the meaning set forth in Section 2.1 of this Agreement.
 - "Purchaser Indemnified Parties" shall have the meaning set forth in Section 9.2(a) of this Agreement.
 - "Purchaser" shall have the meaning set forth in the preamble to this Agreement.
- "Real Property" shall mean all real property owned or leased by or used, by the Seller in connection with the Business, together with all buildings, improvements, fixtures, easements, licenses, options, insurance proceeds and condemnation awards and all other rights of the Seller in or appurtenant thereto.
- "Regulations" shall mean any laws, statutes, ordinances, regulations, rules, notice requirements, court decisions, agency guidelines, principles of law and orders of any foreign, federal, state or local government and any other Governmental Body, and including without limitation Environmental Laws, Healthcare Laws, Kickback Laws, laws governing radioactive materials, energy, motor vehicle safety, public utility, zoning, building and health codes, occupational safety and health regulations, medical licensure regulations, ant-kickback regulations, regulations of federal and state healthcare industry regulatory organizations and laws respecting employment practices, employee documentation, terms and conditions of employment and wages and hours.

- "Release" means any actual or threatened spilling, leaking, pumping, pouring, emitting, emptying, discharging, injecting, escaping, leaching, dumping, disposing, abandoning or migrating into the environment.
- "Representative" with respect to any Person shall mean any officer, director, principal, attorney, agent, employee or other representative of such Person.
 - "Requisite Stockholder Approval" shall mean the approval of a majority of the holders of Seller's capital stock.
- "Second Earnout Milestone" refers to Purchaser's achieving EBITDA of at least \$***** during the Second Earnout Period, after deducting the aggregate amount (if any) paid to Seller as of December 31, 2008 in respect of the Earnout Payments.
- "Second Earnout Payment" shall mean additional consideration in an amount equal to the lesser of (i) \$******, minus the aggregate amount (if any) paid to Seller in respect of the ******, and (ii) the amount by which Purchaser's ******after deduction of the aggregate amount (if any) paid to Seller during such period in respect of ******. The Second Earnout Payment shall be payable in cash and Common Stock, as follows: (i) cash in an amount equal to ******; and (ii) a number of shares of Common Stock equal to (x) ***** of the total dollar value of the *****, divided by (y) the then applicable Average Price.
 - "Second Earnout Period" shall mean the period commencing on the Closing Date and ending on December 31, 2008.
 - "Seller Indemnified Parties" shall have the meaning set forth in Section 9.2(b) of this Agreement.
 - "Seller's Counsel" shall mean Troutman Sanders LLP.
 - "Seller" shall have the meaning set forth in the preamble of this Agreement.
- "Software" shall mean computer software, programs and databases in any form, including Internet web sites, web content and links, source code, executable code, tools, developers kits, utilities, graphical user interfaces, menus, images, icons and forms, and all versions, updates, corrections, enhancements and modifications thereof, and all related documentation, developer notes, comments and annotations related thereto.
 - "Straddle Period Tax" shall have the meaning set forth in Section 5.6(c) of this Agreement.

"<u>Subsidiary</u>" of a Person shall mean any other Person more than 50% of the voting stock (or of any other form of other voting or controlling equity interest in the case of a Person that is not a corporation) of which is beneficially owned by the Person directly or indirectly through one or more other Persons.

"<u>Tax Returns</u>" shall mean any return, declaration, report, claim for refund, or information return or statement relating to Taxes, including any schedule or attachment thereto, and including any amendment thereof.

"Tax" or "Taxes" shall mean (i) any and all U.S. federal, state, local and non-U.S. taxes, assessments and other governmental charges, duties, impositions and liabilities, including taxes based upon or measured by gross receipts, income, profits, sales, use and occupation, and value added, ad valorem, transfer, franchise, withholding, payroll, recapture, employment, excise and property taxes (including personal and real property taxes), together with all interest, penalties and additions imposed with respect to such amounts, (ii) any liability for the payment of any amounts of the type described in clause (i) of this definition as a result of being a member of an affiliated, consolidated, combined or unitary group for any period, and (iii) any liability for the payment of any amounts of the type described in clauses (i) or (ii) of this definition as a result of any express or implied obligation to indemnify any other person or as a result of any obligations under any agreements or arrangements with any other person with respect to such amounts and including any liability for taxes of a predecessor or transferor entity.

"<u>Technology</u>" shall mean tangible embodiments of Intellectual Property Rights, whether in electronic, written or other media, including Software, technical documentation, specifications, "recipe books," designs, bills of material, build instructions, test reports, schematics, algorithms, user interfaces, routines, formulae, test vectors, IP cores, net lists, photomasks, databases, lab notebooks, processes, prototypes, samples, studies or other know-how and other works of authorship.

"Third Earnout Milestone" refers to Purchaser's achieving EBITDA of at least \$****** during the Third Earnout Period, after deducting the aggregate amount (if any) paid to Seller as of December 31, 2009 in respect of the Earnout Payments.

"Third Earnout Payment" shall mean additional consideration in an amount equal to the lesser of (i) \$******, minus the aggregate amount (if any) paid to Seller in respect of the *******, and (ii) the amount by which Purchaser's ****** after deduction of the aggregate amount (if any) paid to Seller during such period in respect of the ******. The Third Earnout Payment shall be payable in cash and Common Stock, as follows: (i) cash in an amount equal to ********; and (ii) a number of shares of Common Stock equal to (x) ******** of the total dollar value of the ********, divided by (y) the then applicable Average Price.

"Third Earnout Period" shall mean the period commencing on the Closing Date and ending on December 31, 2009.

- "Third Party Claim" shall have the meaning set forth in Section 9.6 of this Agreement.
- "Threshold Amount" shall have the meaning set forth in Section 9.7(a) of this Agreement.
- "<u>Total Revenue</u>" shall mean the aggregate net revenues including, but not limited to adjustments and offsets from all sources for the applicable period, calculated in accordance with GAAP as then currently applied by Parent.
 - "Transfer Taxes" shall have the meaning set forth in Section 2.5 of this Agreement.
 - "Treasury Regulations" shall mean the Treasury Regulations promulgated under the Code.
- "<u>University Agreement</u>" means an agreement by and between Purchaser and/or DIS on the one hand and ****** on the other hand, pursuant to which ****** through appropriate affiliated medical personnel and/or facilities agrees to provide to Purchaser's and/or DIS's physician customers professional interpretations of nuclear ultrasound images and reasonable and customary related services.
 - "Unresolved Claims" shall have the meaning set forth in Section 9.5(a) of this Agreement.
 - "*****" means ******* and affiliated institutions and organizations.
- "<u>Vehicles</u>" means a 2003 Ford Excursion with Vehicle Identification Number of 1FMSU45P73ED20509 and a Chevrolet Express with Vehicle Identification Number of 1GCGG25V661221583.
 - "Working Capital" shall have the meaning set forth in Section 2.6(b) of this Agreement.

ARTICLE II. PURCHASE AND SALE OF ASSETS

- 2.1 <u>Transfer of Assets</u>. Upon the terms and subject to the conditions contained in this Agreement, at the Closing, the Seller, pursuant to a Bill of Sale substantially in the form of <u>Exhibit</u> E hereto (the "<u>Bill of Sale</u>"), shall sell, convey, transfer, assign and deliver to the Purchaser, and the Purchaser shall acquire from the Seller, free and clear of all Encumbrances, all of the Seller's right, title and interest in and to the properties and assets and rights of any kind, whether tangible or intangible, real or personal, used in or related to the Business (collectively, the "<u>Purchased Assets</u>"), including, but not limited to:
- (a) all accounts and notes receivable (whether current or noncurrent), refunds, deposits, prepayments or prepaid expenses (including without limitation any prepaid insurance premiums) of the Business;

- (b) all of the Seller's rights under the Contracts (the "Contract Rights");
- (c) all of the fixed assets related to the Business within the Seller's Facilities and any services or maintenance Contracts with Customers of the Business and any Software used to operate equipment used in the Business;
 - (d) the Seller's rights under the Leases;
 - (e) Leasehold Improvements;
- (f) the Seller's right to enforce the Settlement and Release Agreement by and between the Seller and Matthew G. Molchan dated as of April 30, 2007 and attached hereto as Exhibit K (the "Molchan Settlement Agreement");
 - (g) Inventory;
 - (h) all furniture, fixtures and equipment related to the Business within the Seller's Facilities;
 - (i) Backlog;
 - (j) books and records related to the Business, the Purchased Assets or the Assumed Liabilities;
- (k) the supplies, sales literature, catalogues, brochures, promotional literature, customer, supplier and distributor lists, art work, other marketing materials, telephone and fax numbers and purchasing records related to the Business;
- (l) all Seller's right, title and interest in the name "Ultrascan", any trade names, trade dress, designs and marketing materials currently in use by the Seller, and the "www.ultrascaninc.com" web site including the image and likeness of Hilie King;
 - (m) the Seller's insurance policies with respect to the Business, to the extent assignable;
- (n) all claims, causes of action, choses in action, rights of recovery and rights of set-off of any kind, against any Person, including, without limitation, any liens, security interests, pledges or other rights to payment or to enforce payment in connection with products or services delivered by the Seller with respect to the Business on or prior to the Closing Date; and
 - (o) any and all instruments and documents relating to any of the foregoing.

- 2.2 <u>Excluded Assets</u>. Notwithstanding the foregoing, there shall be excluded from the Purchased Assets and not transferred to the Purchaser, and to the extent in existence on the Closing Date, there shall be retained by the Seller, the assets related to the Business set forth on <u>Exhibit F</u> hereto or otherwise specified by the Purchaser prior to the Closing (the "<u>Excluded Assets</u>").
- 2.3 No Assumption of Liabilities. Notwithstanding any provision in this Agreement or any of the other Ancillary Agreements to the contrary, the Purchaser shall not assume any Liabilities of the Seller (or any predecessor owner of all or part of the Seller's business and assets), including, without limitation, any Taxes of Seller (including any liability for Taxes arising from or attributable to Seller's operation of the Business or use or ownership of the Purchased Assets for all Pre-Closing Tax Periods, and including any Transfer Taxes and Straddle Period Taxes attributable to Seller pursuant to this Agreement), liabilities arising from the merger or other combination of Ultrascan, Inc. and Ultrascan of Macon, Inc., liabilities arising from the failure to pay employee overtime or any other such employee matter, any such Liabilities related to any Pre-Closing Environmental Matters, Pre Closing Healthcare Regulatory Matters, operating lease obligations, Employee Loans or any Liabilities of any Employee Plan, except the payment of up to \$1,500,000 of any commercial debt of Seller to Gwinnett Community Bank and the lease pay out obligations for the Vehicles (the "Commercial Debt"), and other liabilities of the Seller that have been agreed to by Purchaser, all as set forth on Exhibit G hereto (the "Assumed Liabilities"). All such non-assumed Liabilities shall be retained by and remain Liabilities of the Seller. Purchaser shall not assume any operational lease obligations of the Seller except for the Assumed Operational Lease Obligations set forth on Exhibit G-1 hereto. All such non-assumed operational lease obligations shall be retained by and remain liabilities and obligations of the Seller.
 - 2.4 <u>Purchase Price</u>. As consideration for the sale, assignment, transfer, conveyance and delivery of the Purchased Assets to the Purchaser:
- (a) <u>Closing Payment</u>. At the Closing, Purchaser shall pay to Seller cash in the amount of \$******, by wire transfer of immediately available funds to account numbers provided to the Purchaser by the Seller prior to the Closing (the "Closing Payment").
- (b) <u>Closing Escrow Fund</u>. Effective as of Closing, Parent, DIS, Purchaser, Seller and the Escrow Agent are entering into the Escrow Agreement substantially in the form attached hereto as <u>Exhibit H</u> (the "<u>Escrow Agreement</u>"). Notwithstanding anything to the contrary set forth in this Agreement, at the Closing, Parent shall deposit with the Escrow Agent (i) cash in the amount of \$***** (the "<u>Closing Escrow Amount</u>") and (ii) cash in the amount of \$***** (the "***** Escrow Amount,"). The Escrow Agent shall hold the Closing Escrow Amount and the ****** Escrow Amount in an account (the "<u>Closing Escrow Fund</u>"). The Escrow Agent shall hold the Closing Escrow Amount as security for Seller's indemnification obligations under <u>Article IX</u>. All interest earned on the Closing Escrow Fund shall inure to the benefit of the Seller. If not paid to Parent in connection with the indemnification obligations owed to any Parent Indemnified Party, the failure to execute the ********, the failure to obtain the ********, or to cover Taxes, any income and gains of the Closing Escrow Fund shall ultimately be distributable to the Seller in accordance with this Agreement and the Escrow Agreement.

- (c) <u>Earnout Payments</u>. In addition to the Cash Consideration and subject to the terms and conditions of this Agreement, Purchaser shall pay to Seller the Earnout Payments, as follows:
- (i) If all of the First Earnout Milestones are achieved, Purchaser shall deliver the First Earnout Payment to Seller within sixty (60) days after the first anniversary of the Closing Date; *provided* that Seller shall be entitled to receive no portion of the First Earnout Payment if any one of the First Earnout Milestones is not achieved, but, in such event, Seller will remain eligible to receive the Second Earnout Payment, Third Earnout Payment, Fourth Earnout Payment and Fifth Earnout Payment, subject in each case to the terms and conditions contemplated herein.
- (ii) If the Second Earnout Milestone is achieved, Purchaser shall deliver the Second Earnout Payment to Seller within sixty (60) days after December 31, 2008.
- (iii) If the Third Earnout Milestone is achieved, Purchaser shall deliver the Third Earnout Payment to Seller within sixty (60) days after December 31, 2009.
- (iv) If the Fourth Earnout Milestone is achieved, Purchaser shall deliver the Fourth Earnout Payment to Seller within sixty (60) days after December 31, 2010.
- (v) If the Fifth Earnout Milestone is achieved, Purchaser shall deliver the Fifth Earnout Payment to Seller within sixty (60) days after April 30, 2011.
- (d) The Purchase Price (plus Assumed Liabilities to the extent properly taken into account under the Code and the applicable Treasury Regulations), shall be allocated among the Purchased Assets as reasonably agreed to by the Seller and the Purchaser (the "Allocation") within sixty (60) days of the Closing Date. The Seller and the Purchaser agree to (i) be bound by the Allocation; (ii) act in accordance with the Allocation in the preparation of financial statements and filing of all Tax Returns (including, without limitation, filing Form 8594 with their United States federal income Tax Return for the taxable year that includes the date of the Closing) and in the course of any Tax audit, Tax review or Tax litigation relating thereto and (iii) take no position and cause their Affiliates to take no position inconsistent with the Allocation for income Tax purposes, including United States federal and state income Tax and foreign income Tax. Not later than thirty (30) days prior to the filing of their respective Forms 8594 relating to this transaction, each party shall deliver to the other party a copy of its Form 8594.
- 2.5 <u>Closing Costs</u>; <u>Transfer Taxes</u>; <u>Fees and Expenses</u>. Notwithstanding any other provision of this Agreement, the Seller shall be responsible for any documentary and transfer taxes and any sales, use or other taxes imposed by reason of the transfers of Purchased Assets provided hereunder and any deficiency, interest or penalty asserted with respect thereto ("<u>Transfer Taxes</u>"). The parties hereto shall cooperate, to the extent reasonably requested and permitted by applicable

law, in minimizing any such Transfer Taxes, including but not limited to the transfer by remote electronic transmission of all Purchased Assets capable of being so transmitted. The party required by law to file a Tax Return with respect to such Transfer Taxes shall do so within the time period prescribed by law, and Seller shall promptly reimburse the Purchaser for any Transfer Taxes so paid by the Purchaser upon receipt of notice that such Transfer Taxes have been paid. The Seller shall pay the premium for a leasehold policy of title insurance with regard to each material parcel of Leased Real Property, in an amount and in a form acceptable to the Purchaser. Except as expressly provided otherwise in this Agreement, each party to this Agreement shall bear its own costs, fees and expenses incurred in connection with the negotiation and preparation of this Agreement and the consummation of the transactions contemplated by this Agreement and the Ancillary Agreements.

2.6 Purchase Price Adjustment

- (a) <u>Delivery of Preliminary Statement</u>. Not later than three (3) Business Days prior to the Closing Date, the Seller shall deliver to the Purchaser a certificate (the "<u>Preliminary Statement</u>") setting forth the Seller's estimate of the Working Capital as of the close of business on the Closing Date (the "<u>Estimated Working Capital</u>"). The Preliminary Statement shall be certified and signed by an authorized officer of the Seller and shall be prepared in accordance with the requirements of this Section 2.6.
- (b) <u>Working Capital</u>. "<u>Working Capital</u>" shall mean those line items comprising the current assets to be acquired by Purchaser, calculated on a GAAP basis, of the Business minus those line items comprising the current liabilities of the Business to be assumed by Purchaser (excluding the Commercial Debt), in each case, as set forth on <u>Exhibit J</u> hereto. Without limiting the generality of the foregoing, Working Capital is to be calculated in the same way, using the same methodologies and accounting practices and principles applied on a consistent basis (including with respect to determining estimates and allowances) as the line items comprising Working Capital as set forth on <u>Exhibit I</u>.
- (c) <u>Initial Purchase Price Adjustment</u>. At the Closing, the Purchase Price and the Cash Consideration shall be (i) increased, dollar for dollar, by the amount by which the Estimated Working Capital exceeds the Working Capital (the "<u>Benchmark</u>") or (ii) decreased, dollar for dollar, by the amount by which the Estimated Working Capital is less than the Benchmark.
- (d) <u>Post-Closing Statement</u>. Within ninety (90) calendar days after the Closing Date, the Purchaser shall prepare and deliver to the Seller a statement (the "<u>Post-Closing Statement</u>") setting forth its calculation of the Working Capital as of the close of business on the Closing Date (the "<u>Final Working Capital</u>"). The Post-Closing Statement shall be prepared in accordance with the requirements of this Section 2.6.
- (e) <u>Disputes</u>. During the thirty- (30)-calendar-day period following receipt of the Post-Closing Statement, the Seller and its Representatives shall be permitted to review the books, records, working papers and all other information and materials relating to the Post-Closing Statement, and the Purchaser shall cooperate with the Seller to provide it and its Representatives with access to facilities and personnel (including those responsible for the

preparation of the Post-Closing Statement) and any other information used in preparing the Post-Closing Statement reasonably requested by them. The Post-Closing Statement shall become final and binding on the thirtieth (30th) calendar day following the Seller's receipt thereof, provided the Purchaser has reasonably cooperated as provided above, unless the Seller gives written notice of its disagreement with the Post-Closing Statement (a "Notice of Disagreement") to the Purchaser on or prior to such date. Any Notice of Disagreement shall specify the nature of any disagreement so asserted. If a Notice of Disagreement is received by the Purchaser in a timely manner, then the Post-Closing Statement (as revised in accordance with this sentence) shall become final and binding on the earlier of (i) the date on which the Purchaser and the Seller resolve in writing any differences they have with respect to the matters specified in the Notice of Disagreement or (ii) the date on which any disputed matters are finally resolved in writing by the Accounting Firm (as defined below). During the fifteen-(15)calendar-day period following the delivery of a Notice of Disagreement, the Purchaser and the Seller shall seek in good faith to resolve in writing any differences that they may have with respect to the matters specified in the Notice of Disagreement. If such dispute has not been resolved by the end of such fifteen-(15)calendar-day period, the Purchaser and the Seller shall escalate the dispute to their respective Chief Financial Officers or similar level executives who shall consider the dispute in a mutually agreeable location and shall attempt in good faith to resolve the dispute. If such officers are unable to resolve such dispute within ten (10) calendar days, the Purchaser and the Seller shall submit for arbitration any and all matters that remain in dispute and were properly included in the Notice of Disagreement to Ernst & Young, LLP in Atlanta, Georgia (or such other nationally recognized independent firm that is mutually agreed upon by the Purchaser and the Seller) (the "Accounting Firm") for arbitration by a mutually acceptable partner of the Accounting Firm (provided that if either party shall reasonably discover a conflict of interest associated with Ernst & Young, LLP, the parties shall retain Deloitte & Touche USA LLP to act as the Accounting Firm). The Purchaser and the Seller shall retain the Accounting Firm no later than five (5) Business Days following the expiration of such ten-(10)-calendar-day period. In the event of a failure to retain the Accounting Firm during such time period, either the Purchaser or the Seller, acting individually, shall have the right to retain the Accounting Firm on behalf of both the Purchaser and the Seller. The Purchaser and the Seller shall use their commercially reasonable efforts to cause the Accounting Firm to render a decision resolving any matters submitted to the Accounting Firm within thirty (30) calendar days following submission thereof. Judgment may be entered upon the determination of the Accounting Firm in any court having jurisdiction over the party against which such determination is to be enforced. Without limiting the generality of the foregoing, the Purchaser and the Seller shall each promptly provide, or cause to be provided, to the Accounting Firm all information, and to make available to the Accounting Firm all personnel, as are reasonably necessary to permit the Accounting Firm to resolve any disputes pursuant to this Section 2.6. The fees and expenses of the Accounting Firm and reasonable attorneys' fees and expenses of the parties pursuant to this Section 2.6(e) shall be borne by the Purchaser and the Seller proportionately based on which party's position was closest to the determination of the Accounting Firm. The scope of the disputes to be resolved by the Accounting Firm, and the scope of the Accounting Firm's review, shall be limited to disputes concerning whether such calculation was performed in accordance with the guidelines set forth in this Section 2.6, whether or not there were errors in the Post-Closing Statement and the other matters specifically set forth in this Section 2.6(e), and the Accounting Firm shall not make any other determination. The date on which the Final Working Capital is finally determined in accordance with this Section 2.6(e) is hereinafter referred to as the "Determination Date."

- (f) <u>Final Purchase Price Adjustment</u>. The "<u>Adjustment Amount</u>" shall equal the Final Working Capital minus the Estimated Working Capital, which may be positive or negative. If the Adjustment Amount is a positive number, then the Purchase Price shall be increased by the absolute value of the Adjustment Amount, and if the Adjustment Amount is a negative number, then the Purchase Price shall be decreased by the absolute value of the Adjustment Amount.
- (i) If the Adjustment Amount is a positive number then, promptly following the Determination Date, and in any event within five (5) Business Days thereafter, the Purchaser shall pay to the Seller the Adjustment Amount, as finally determined, plus interest on the Adjustment Amount from the Closing Date to the date of payment at the "Prime Rate" of interest published in the "Money Rates" column of *The Wall Street Journal* (or the average of such rates if more than one rate is indicated) on the Closing Date.
- (ii) If the Adjustment Amount is a negative number then, promptly following the Determination Date, and in any event within five (5) Business Days thereafter, the Seller shall pay to the Purchaser an aggregate amount equal to the absolute value of the Adjustment Amount, as finally determined, plus interest on the absolute value of the Adjustment Amount from the Closing Date to the date of payment at the "Prime Rate" of interest published in the "Money Rates" column of *The Wall Street Journal* (or the average of such rates if more than one rate is indicated) on the Closing Date.
- (iii) Notwithstanding the foregoing, upon the final determination of Final Working Capital and after the Closing Date, to the extent that any portion of the Assumed Liabilities constituting payroll liabilities or non-payroll liabilities reflected on the Post-Closing Statement is not transferable to, or able to be assumed by, the Purchaser under applicable Regulations or otherwise (such Assumed Liabilities, the "Nontransferable Liabilities"), to the extent that such Liabilities remain Liabilities of the Seller after the Closing Date, then the Purchaser shall remit to the Seller an amount of cash equal to the amount of such Nontransferable Liabilities at such time as the Adjustment Amount is paid by the Seller or the Purchaser, as the case may be. To the extent that an Adjustment Amount is not payable pursuant to this Section 2.6(f), then the Purchaser shall reimburse the Seller for any Nontransferable Liabilities within five (5) Business Days following the Determination Date.
- 2.7 <u>Closing</u>. Subject to the satisfaction or waiver of the conditions set forth in Articles VI and VII of this Agreement, the closing of the sale of the Purchased Assets to the Purchaser and the other transactions contemplated by this Agreement and the Ancillary Agreements (the "<u>Closing</u>") shall take place by the electronic exchange of signature pages, at a time to be agreed upon by the Purchaser and the Seller, on a date (no later than the fifth (5th) Business Day following the satisfaction or waiver of the last to be satisfied or waived of the conditions set forth in Articles VI and VII of this Agreement) to be agreed upon by the Purchaser and the Seller (the "<u>Closing Date</u>").

ARTICLE III. REPRESENTATIONS AND WARRANTIES OF THE SELLER AND PRINCIPAL STOCKHOLDER

As an inducement to Parent, DIS and Purchaser to enter into this Agreement, except as set forth in the appropriate section of the Disclosure Schedule, the Seller, the Seller's Subsidiaries and Principal Stockholder hereby represent and warrant to the Purchaser, as of the date hereof and as of the Closing, as follows:

- 3.1 <u>Organization of the Seller</u>. The Seller is a corporation duly organized and validly existing under the laws of the State of Georgia with full corporate power and corporate authority to conduct its business as it is presently being conducted, to own or lease, as applicable, its Real Property and the Purchased Assets, and to perform all its obligations under the Contracts to which it is a party. The Seller is duly qualified to do business as a foreign corporation and is in good standing in each jurisdiction where the character of its properties owned or leased or the nature of its activities make such qualification necessary, except where the failure to be so qualified or in good standing would not have a Material Adverse Effect. Copies of the Articles of Incorporation and Bylaws of the Seller heretofore delivered to the Purchaser are accurate and complete as of the date hereof.
- 3.2 <u>Authorization</u>. The Seller has all requisite power and authority to execute, deliver and perform this Agreement, the Ancillary Agreements and the Molchan Settlement Agreement to which it is a party, to consummate the transactions contemplated hereby and thereby and to perform its obligations hereunder and thereunder. The execution and delivery of this Agreement, the Ancillary Agreements and the Molchan Settlement Agreement by the Seller, and the consummation by the Seller of the transactions contemplated hereby and thereby have been duly approved by the board of directors of the Seller. This Agreement has been duly executed and delivered by the Seller and is, and upon execution and delivery of the Ancillary Agreements, this Agreement, the Ancillary Agreements and the Molchan Settlement Agreement to which the Seller is a party will be, the legal, valid and binding obligations of the Seller, enforceable against the Seller in accordance with their terms, except as enforcement may be limited by applicable bankruptcy, insolvency, reorganization, moratorium and other laws affecting enforcement of creditor's rights generally and except insofar as the availability of equitable remedies may be limited by applicable Regulations.
- 3.3 <u>Title to Purchased Assets</u>; <u>Condition of Purchased Assets</u>. The Seller has, and will transfer to the Purchaser, sole ownership of (and, in the case of Real Property, good and marketable title to or valid leasehold interests in) each of the Purchased Assets and upon the consummation of the transactions contemplated hereby, the Purchaser will acquire sole ownership of (and, in the case of Real Property, good and marketable title to or valid leasehold interests in) all of the Purchased Assets, free and clear of any Encumbrances. All tangible assets and properties which are part of the Purchased Assets are in good operating condition and repair, normal wear and tear excepted, and are usable in the Ordinary Course of Business as currently conducted. Section 3.3 of the Disclosure Schedule sets forth a list of all tangible Purchased Assets as of the date of this Agreement.

- 3.4 <u>Absence of Certain Changes or Events</u>. Except as set forth in Section 3.4 of the Disclosure Schedule, since December 31, 2006, there has not been any:
 - (a) Material Adverse Change;
- (b) sale, assignment or transfer of any of the assets of the Seller related to the Business, singly or in the aggregate, other than sales, assignments or transfers from Inventory in the Ordinary Course of Business;
 - (c) waiver of any rights of substantial value to the Seller with respect to the Business;
 - (d) amendment, cancellation or termination of any Contract or Permit material to the operation of the Business;
 - (e) capital expenditure or execution of any lease relating to the Business or any incurring of \$***** in the aggregate relating to the Business;
 - (f) failure to pay any material obligation related to the Business except those contested in good faith and for which proper reserves have been

made;

- (g) failure to operate the Business in the Ordinary Course so as to use reasonable efforts to preserve the Purchased Assets and the Business intact and to preserve the goodwill of suppliers, customers, employees and others having business relations with the Seller or its Representatives with respect to the Business, including, without limitation, continuing to invest in maintenance and ordinary course capital expenditures for the manufacturing Facility and related equipment;
 - (h) change in accounting methods or practices by the Seller with respect to the Business;
- (i) revaluation of any of the Purchased Assets, including without limitation, writing off notes or accounts receivable other than (i) for which reserves have been established or (ii) in the Ordinary Course of Business;
- (j) damage, destruction or loss (whether or not covered by insurance) adversely affecting the Purchased Assets or the operation of the Business:
 - (k) creation of a new Encumbrance against any of the Purchased Assets, singly or in the aggregate;
- (l) indebtedness incurred by the Seller for borrowed or purchase money or any commitment to borrow money entered into by the Seller, or any loans made or agreed to be made by the Seller, in each case, with respect to the Business;

(m) any Liability incurred by the Seller with respect to the Business except in the Ordinary Course of Business;

Business;

- (n) any change in any assumptions underlying or methods of calculating any bad debt, contingency or other reserves with respect to the
- (o) acquisition of any interest in any other business entity by the Seller that is engaged in the same business as the Business;
- (p) except in the Ordinary Course of Business consistent with past practice, increase in, or promise to increase, compensation or benefits payable to or to become payable to any employees, officers or directors of the Seller relating to the Business, any amendment to any Employee Plans that would materially increase the cost thereof, or establishment of any new Employee Plans;
- (q) other event or condition of any character which in any one case or in the aggregate has had or might be reasonably expected to have a Material Adverse Effect; or
 - (r) agreement by the Seller or any of its Affiliates to do any of the foregoing.

3.5 Material Contracts.

- (a) <u>Section 3.5</u> of the Disclosure Schedule lists all of the Material Contracts to which the Seller is a party or is bound related to the Business as of the date of this Agreement.
 - (b) "Material Contracts" shall mean:
 - (i) Contracts related to the Business not made in the Ordinary Course of Business;
 - (ii) license agreements or royalty Contracts related to the Business, whether the Seller is the licensor or licensee thereunder;
 - (iii) confidentiality and non-disclosure agreements related to the Business, whether the Seller is the beneficiary or the obligated party

thereunder;

- (iv) Contracts related to the Business involving future expenditures or Liabilities, actual or potential, or future revenues in excess of \$***** after the date hereof or otherwise material to the Business or the Purchased Assets;
- (v) employment contracts, consulting contracts and severance agreements with any director, officer or employee of the Seller engaged in the Business, including Contracts (A) to employ or terminate executive officers or other personnel and other contracts with present or former officers or directors of the Seller or (B) that will result in the payment by, or the

creation of any Liability to pay on behalf of the Seller or any of Parent, DIS or Purchaser any severance, termination, "golden parachute," or other similar payments to any present or former personnel of the Seller following termination of employment or otherwise as a result of the consummation of the transactions contemplated by this Agreement and the Ancillary Agreements;

- (vi) indemnification agreements providing for the indemnification by the Seller of any director, officer or employee of the Seller engaged in the Business;
 - (vii) Contracts to procure key raw materials used in the Business, such as wafers and chemicals;
- (viii) promissory notes, loans, agreements, indentures, evidences of indebtedness, guarantees, currency or interest rate agreements or other instruments relating to an obligation to pay money with respect to the Business, whether the Seller shall be the borrower, lender or guarantor thereunder in excess of \$***** in the aggregate;
 - (ix) any licenses or Contracts with equipment or Software vendors;
- (x) Contracts containing covenants limiting the freedom of the Seller or any officer, director, employee, or other Affiliates of the Seller, to engage in any line of business or compete with any Person that relates to the Business;
 - (xi) Contracts with any Governmental Body that relate to the Business;
 - (xii) Leases;
 - (xiii) distribution agreements related to the Business;
 - (xiv) letters of credit related to the Business;
 - (xv) Contracts with any customers of the Business; and
 - (xvi) any other Contract material to the operation of the Business.

True, correct and complete copies of all of the Material Contracts which are written, or written summaries of oral Material Contracts, including all amendments and supplements thereto, have been delivered to the Purchaser. Each such Material Contract is in full force and effect, paid currently, and has not been materially impaired by any acts or omissions of the Seller or any of its Representatives. Except as set forth in Section 3.5 of the Disclosure Schedule, no Material Contract requires the consent of, or notification to, any other contracting party to the transactions contemplated by this Agreement and the Ancillary Agreements. Each of the Material Contracts is valid, binding and enforceable in accordance with its terms except as enforcement may be limited by applicable bankruptcy, insolvency, reorganization, moratorium and other laws affecting enforcement of creditor's rights generally and except insofar as the availability of equitable remedies may be

limited by applicable Regulations. The Seller has fulfilled, or taken all action necessary to enable it to fulfill when due, all of its obligations under each of such Material Contracts. To the Knowledge of the Seller, all other parties to such Material Contracts have complied with the provisions thereof, no party is in Default thereunder and no notice, whether written or oral, of any claim of Default has been given to the Seller. The products and services called for by any unfinished Material Contract can be supplied in accordance with the terms of such Material Contract, including time specifications, and, to the Knowledge of the Seller, as of the date of this Agreement and as of the Closing Date, no unfinished Material Contract will, upon performance by the Seller, result in a loss to the Seller. To the Knowledge of the Seller, no party to any Material Contract intends to terminate or amend the terms thereof or to refuse to renew any such Material Contract upon expiration of its term.

3.6 Facilities.

- (a) <u>Actions</u>. There are no pending or, to the Knowledge of the Seller, threatened condemnation proceedings or other Actions relating to any Facility.
- (b) <u>Leases or Other Agreements</u>. There are no Leases or other agreements granting to any Person the right to purchase, use or occupy any Facility, or any Real Property in connection with the Business or any portion thereof or interest in any such Facility or Real Property.
- (c) <u>Facility Leases and Leased Real Property</u>. With respect to each Lease listed on Section 3.5 of the Disclosure Schedule, the Seller has an unencumbered interest in the Leasehold Estate. The Seller enjoys peaceful and undisturbed possession of all the Leased Real Property. The Seller has delivered to the Purchaser a true and correct copy of each of the Leases. Each Lease is valid and in full force and effect, is unmodified and represents the entire agreement between the Seller and the applicable landlord. There exist no obligations on the part of the Seller to be paid or performed with respect to the Leases other than those set forth in the Leases. The Seller and, to the Knowledge of the Seller, no other party to such Leases, is in default of its respective obligations under any Lease.
- (d) <u>Certificate of Occupancy; Compliance with Law.</u> To the Knowledge of the Seller, all Facilities have received all required approvals of Governmental Bodies (including, without limitation, Permits and a certificate of occupancy or other similar certificate permitting lawful occupancy of the Facilities) required in connection with the operation thereof and have been operated and maintained in accordance with, and the Seller is not in violation of, any Regulation applicable to any Facility or any parcel of Real Property, including without limitation the Americans with Disabilities Act and the Map Act.
- (e) <u>Utilities</u>. All Facilities are supplied with utilities (including, without limitation, water, sewage, disposal, electricity, gas and telephone) and other services necessary for the operation of such Facilities as currently operated, and there is no condition which would reasonably he expected to result in the termination of the present access from any Facility to such utility services. All utilities that service the Facilities are provided via public roads or via permanent, irrevocable, appurtenant easements benefiting the applicable Facility.

- (f) Improvements, Fixtures and Equipment. The improvements constructed on the Facilities, including, without limitation, all Leasehold Improvements, and all fixtures and equipment and other tangible assets owned, leased or used by the Seller at the Facilities are (i) insured to the extent and in a manner customary in the industry; (ii) structurally sound with no known material defects; (iii) in good operating condition and repair, subject to ordinary wear and tear; (iv) not in need of maintenance, repair or correction except for ordinary routine maintenance and repair, the cost of which would not be material; (v) sufficient for the operation of the Business as presently conducted and (vi) in conformity with all applicable Regulations.
- (g) No Special Assessment. The Seller has not received notice of any special assessment relating to any Facility or any portion thereof and there is no pending or threatened special assessment.
- (h) <u>Access; Shared Facilities</u>. Each Facility abuts on and has direct vehicular access to a public road, or has access to a public road via a permanent, irrevocable, appurtenant easement benefiting the parcel, and access to each Facility is provided by paved public right-of-way with adequate curb cuts available. There are no shared facilities or services at the Facilities which are used in connection with any business or other operations of the Seller or any of its affiliates, other than the Business.
- 3.7 Permits. Section 3.7 of the Disclosure Schedule sets forth a complete list of all Permits, all of which are as of the date hereof, and all of which will be as of the Closing Date, in full force and effect. The Seller has, and at all times has had, all material Permits required under any Regulation in the operation of the Business or in the ownership of the Purchased Assets, including all requisite approvals by the United States Department of Commerce with respect to all of the Seller's products exported or sold outside of the United States, and owns or possesses such Permits free and clear of all Encumbrances. The Seller is not in Default, nor has it received any notice, whether written or oral, of any claim of Default, with respect to any such Permit. Except as otherwise governed by law, all such Permits are renewable by their terms or in the Ordinary Course of Business without the need to comply with any special qualification procedures or to pay any amounts other than routine filing fees and will not be adversely affected by the completion of the transactions contemplated by this Agreement and the Ancillary Agreements. No present or former shareholder, director, officer or employee of the Seller or any Affiliate thereof, or any other Person, owns or has any proprietary, financial or other interest (direct or indirect) in any Permit which the Seller owns, possesses or uses.
- 3.8 No Conflict or Violation. Except as set forth in Section 3.8 of the Disclosure Schedule, the execution, delivery or performance of this Agreement and the Ancillary Agreements and the consummation of the transactions contemplated hereby or thereby, and compliance by the Seller with any of the provisions hereof or thereof, will not (a) violate or conflict with any provision of the Articles of Incorporation or Bylaws of the Seller; (b) violate, conflict with, or result in or constitute a Default under, or result in the termination of, or accelerate the performance required by, or result in a right of termination or acceleration under, any of the terms, conditions or provisions of any Material Contract or Permit; (c) violate, conflict with, contravene or give any Person the right to exercise any remedy or obtain any relief under any Regulation or Court Order or (d) impose any Encumbrance on the Purchased Assets or the Business.

3.9 <u>Consents and Approvals</u>. Except as set forth in Section 3.9 of the Disclosure Schedule and the Requisite Stockholder Approval, no Consent is required to be made or obtained by the Seller or any of its Affiliates in connection with the execution, delivery and performance by the Seller of this Agreement and the Ancillary Agreements and the consummation of the transactions contemplated hereby and thereby.

3.10 Financial Information.

- (a) The Seller has delivered to the Purchaser (i) the unaudited schedule of assets and liabilities of the Business as of December 31, 2006, and the related unaudited statements of profit and loss and cash flows for the twelve (12) months then ended and (ii) the unaudited schedule of assets and liabilities of the Business as of March 31, 2007 (the "March Balance Sheet"), and the related unaudited statements of profit and loss and cash flows for the period then ended (collectively, the "Financial Statements"). The Financial Statements (i) are true, accurate and complete, (ii) are consistent with the Seller's books and records and (iii) present fairly and accurately the financial position of the Business as of the respective dates thereof and the results of operations of the Business as of and through such dates.
- (b) Section 3.10(b) of the Disclosure Schedule sets forth a true, correct and complete itemization of the accounts receivable (including aging) of the Business as of March 31, 2007 (the "Accounts Receivable"). The Accounts Receivable represent bona fide claims against debtors for sales, services performed or other charges arising on or before the respective dates of recording thereof, and all of the goods delivered and services performed which gave rise to the Accounts Receivable were delivered or performed in accordance with applicable orders, Contracts or customer requirements. To the Knowledge of the Seller, all Accounts Receivable are fully collectible in the Ordinary Course of Business within three (3) months except to the extent of an amount not in excess of the reserve for doubtful accounts reflected on the March Balance Sheet.
- (c) <u>Section 3.10(c)</u> of the Disclosure Schedule sets forth a true, correct and complete itemization of the Inventory of the Seller as of March 31, 2007 (the "<u>March Inventory</u>"). Within 30 days following Closing, the Seller will deliver to the Purchaser a true, correct and complete itemization of the Inventory of the Seller as of April 30, 2007. The March Inventory has been determined in accordance with the normal valuation policy of the Seller, consistently applied and in accordance with GAAP. The March Inventory consists only of items of quality and quantity commercially usable and salable in the Ordinary Course of Business, except for any items of obsolete material or material below standard quality, all of which have been written down to realizable market value, or for which adequate reserves have been provided and reflected on the March Balance Sheet, and the present quantity of all March Inventory is reasonable in the present circumstances of the Business.
- (d) Section 3.10(d) of the Disclosure Schedule sets forth a true, correct and complete itemization of all accepted and unfulfilled orders for services entered into by the Seller

with respect to the Business as of March 31, 2007 (the "Backlog"). No outstanding purchase commitment of the Seller with respect to the Business presently is in excess of the normal, ordinary and usual requirements of the Business or was made at any price in excess of the now current market price or contains terms or conditions more onerous than those usual and customary in the Business. There is no outstanding bid, proposal, contract or unfulfilled order of the Seller which will or would, if accepted, result in a net loss to the Business.

- 3.11 <u>Books and Records</u>. The Seller has made and kept (and given the Purchaser access to) its true, correct and complete books and records and accounts, which, in reasonable detail, accurately and fairly reflect the activities of the Business. The Seller has not engaged in any transaction, maintained any bank account or used any corporate funds with respect to the Business, except for transactions, bank accounts and funds which have been and are reflected in the books and records with respect to the Business to which the Seller has given the Purchaser access.
- 3.12 <u>Litigation</u>. There is no Action pending or, to the Knowledge of the Seller, threatened, (i) against, related to or affecting (A) the Business or the Purchased Assets (including with respect to Environmental Laws, Healthcare Laws, Kickback Laws, Environmental Claim or Environmental Exposure Claim) or (B) any officer, director or employee of the Seller, in such capacity, with respect to the Business; (ii) seeking to delay, limit or enjoin the transactions contemplated by this Agreement or the Ancillary Agreements; (iii) that involves the risk of criminal liability by the Seller with respect to the Business or (iv) in which the Seller is a plaintiff, including any derivative suits brought by or on behalf of the Seller, with respect to the Business. The Seller is not in Default with respect to or subject to any Court Order applicable to the Business, and there are no unsatisfied judgments against the Business or the Purchased Assets. To the Knowledge of the Seller, there is not a reasonable likelihood of an adverse determination of any pending Action related to the Business. There are no Court Orders or agreements with, or liens by, any Person, Governmental Body or quasi-Governmental Body relating to any Environmental Law which regulate, obligate, bind or in any way affect the Purchased Assets (including any Operating Site) or the Business.
- 3.13 <u>Labor Matters</u>. The Seller is not a party to any labor agreement with respect to its employees relating to the Business with any labor organization, union, group or association and there are no employee unions (nor any other similar labor or employee organizations) under local statutes, custom or practice, which pertain to any current or former Business Employees. The Seller has not experienced any attempt by organized labor or its Representatives to make it conform to demands of organized labor relating to its Business Employees or to enter into a binding agreement with organized labor that would cover the Business Employees. There is no labor strike or labor disturbance pending or, to the Knowledge of the Seller, threatened against the Seller, nor is any grievance currently being asserted, and the Seller has not experienced a work stoppage or other labor difficulty, and is not and has not engaged in any unfair labor practice relating to the Business. The Seller is in material compliance with all applicable Regulations relating to the employment of labor relating to the Business, including but not limited to wages, hours, benefits, nondiscrimination, immigration, the payment of Social Security and similar Taxes and collective bargaining with respect to the Business. Without limiting the foregoing, with respect to the Business, the Seller is in

compliance with the Immigration Reform and Control Act of 1986 and maintains a current Form I-9, as required by such act, in the personnel file of each Business Employee hired after November 9, 1986. The Seller has not received any notice of a sex, age, race, disability, gender or religious discrimination claim brought or threatened to be brought against the Seller relating to the Business. Section 3.13 of the Disclosure Schedule sets forth the names and lists the current annual salary rates or current hourly wages of all Hired Employees, and also sets forth the earnings for each such employee as reflected on Form W2 for the 2006 calendar year, the employment position of each, the location of each and the date of hire and, if applicable, date of rehire of each, with each such individual identified as (i) salaried or hourly; (ii) exempt or nonexempt; (iii) full-time or part-time; (iv) temporary, permanent or leased and (v) active or inactive (with the reason for such inactive status specified, e.g., leave of absence, Family Medical Leave Act, disability, layoff, etc.). To the Knowledge of the Seller, no Business Employee is a party to, or is otherwise bound by, any agreement or arrangement, including any confidentiality, non-competition or proprietary rights agreement, between such employee and any other Person that in any way adversely affects or will affect the performance of his or her duties as an employee of the Seller or DIS or Purchaser. The Seller is not and will not be required to make any filing with or give any notice to, or to obtain any approval, consent, ratification, waiver or other authorization from, any Governmental Body in relation to any of the Hired Employees in connection with the execution and delivery of this Agreement or the Ancillary Agreements or the consummation or performance of the transactions contemplated hereby or thereby.

3.14 <u>Liabilities</u>. The Seller does not have Liabilities due or to become due with respect to the Business, except (a) Liabilities arising in the Ordinary Course of Business under Contracts and Permits and (b) Liabilities incurred since December 31, 2006 in the Ordinary Course of Business and in accordance with this Agreement (none of which relates to any Default under any Contract or Permit, breach of warranty, tort, infringement or violation of any Regulation or Court Order or arose out of any Action) and none of which, individually or in the aggregate, has had or would be reasonably likely to have a Material Adverse Effect.

3.15 Compliance with Law.

- (a) <u>General</u>. Other than with respect to Healthcare Laws and Kickback Laws, to the Seller's Knowledge, the Seller's conduct of the Business has not materially violated and is in material compliance with all Regulations and Court Orders relating to the Purchased Assets or the Business. The Seller has not received any notice to the effect that, or otherwise been advised that, the Seller is not in compliance with any such Regulations or Court Orders, and the Seller does not reasonably anticipate that any existing circumstances are likely to result in violations of any of the foregoing.
- (b) <u>Healthcare and Kickback Laws</u>. To the Seller's Knowledge, Seller and the Principal Stockholder have complied in all material respects with all applicable Healthcare Laws, Kickback Laws and any orders related thereto promulgated by any Governmental Body ("<u>Governmental Order</u>"). Seller and the Principal Stockholder are not relying on any exemption from or deferral of any Healthcare Laws, Kickback Laws, Governmental Order or governmental

authorization related thereto that would not be available to Purchaser after the Closing. No physicians or other health care professionals with either an ownership or compensation relationship with the Seller or the Principal Stockholder are provided compensation or paid any ownership remuneration in any manner that, directly or indirectly, takes into account the value or volume of referrals or other business generated by that physician or other health care professional for Seller or the Principal Stockholder. In addition, none of the physicians or physician groups that perform medical director, professional interpretation, professional supervision, or other personal services on behalf of the Seller or the Principal Stockholder including, without limitation, ********, either refer patients to Seller or the Principal Stockholder or are in a position to refer patients to them.

3.16 No Brokers. Neither the Seller nor any of the Seller's Representatives has entered into or will enter into any contract, agreement, arrangement or understanding with any broker, finder or similar agent or any Person which will result in the obligation of Parent, DIS, Purchaser, Seller or any of their respective Affiliates to pay any finder's fee, brokerage fees or commission or similar payment in connection with the transactions contemplated by this Agreement.

3.17 Intellectual Property.

(a) General. Section 3.17(a) of the Disclosure Schedule sets forth with respect to the Intellectual Property Rights owned by the Seller that are applicable to the Business: (i) for each patent and patent application with a claim or potential claim applicable to the Business, the patent number or application serial number for each jurisdiction in which the patent or application has been filed, the date filed or issued, and the present status thereof; (ii) for each registered trademark, trade name or service mark, the application serial number or registration number, for each country, province and state, and the class of goods covered; (iii) for any URL or domain name, the registration date, any renewal date and name of registry; (iv) for each mask work, the date of first commercial exploitation and if registered, the registration number and date of registration, for each among country, province and state; (v) for the likeness and images of Hilie King, the date of consent to use thereof by the Seller; and (vi) for each registered copyrighted work, the number and date of registration for each by country, province and state in which a copyright application has been registered. In addition, Section 3.17(a) of the Disclosure Schedule includes a list of all Software incorporated in, provided with or otherwise necessary to use, support and maintain the Business. True and correct copies of all applications filed and registrations (including all pending applications and application related documents) related to the Intellectual Property Rights listed on Section 3.17(a) of the Disclosure Schedule have been provided or made available to the Purchaser. Finally, Section 3.17(a) of the Disclosure Schedule lists all in-licenses of Technology or Intellectual Property Rights applicable to the products and services of the Business, other than standard, off-the-shelf Software commercially available on standard terms from third-party vendors.

(b) <u>Sufficiency</u>. The Intellectual Property Rights and Technology owned by the Seller, or licensed by the Seller for use in the Business constitute all Intellectual Property Rights and Technology necessary for the conduct of the Business as presently conducted, including the design, manufacture, license and sale of all products and services currently under development or in production.

- (c) <u>Royalties and Licenses</u>. Except pursuant to the licenses listed in <u>Section 3.17(a)</u> of the Disclosure Schedule, the Seller has no obligation to compensate or account to any Person for the use of any of the Intellectual Property Rights or Technology applicable to the Business.
- (d) <u>Ownership</u>. The Seller (i) owns all right, title and interest in and to the Intellectual Property Rights and Technology purported to be owned by the Seller applicable to the Business, including the Intellectual Property Rights and Technology listed on <u>Section 3.17(a)</u> of the Disclosure Schedule, free and clear of any liens, claims or encumbrances and (ii) has a valid and enforceable right or license to use all other Intellectual Property Rights and Technology used in the conduct of the Business, and all such licensed Intellectual Property Rights and rights to use Technology will not cease to be valid and enforceable rights on behalf of the Business by reason of the execution, delivery and performance of this Agreement or the Ancillary Agreements or the consummation of the transactions contemplated hereby or thereby. Without limiting the foregoing, the Intellectual Property Rights and Technology owned by the Seller have been: (i) developed by employees of the Seller within the scope of their employment; (ii) developed by independent contractors who have assigned their rights to the Seller pursuant to enforceable written agreements or (iii) otherwise acquired by the Seller from a third party who has assigned all Intellectual Property Rights and ownership of all Technology it has developed on the Seller's behalf to the Seller.
- (e) Absence of Claims; Non-infringement. No proceedings, claims or actions have been instituted or are pending against the Seller, or, to the Knowledge of the Seller, are threatened, that challenge the right of the Seller with respect to the use or ownership of the Intellectual Property Rights or Technology of the Seller applicable to the Business. Without limiting the foregoing, no interference, opposition, reissue, reexamination or other proceeding is or has been pending or, to the Knowledge of the Seller, threatened, in which the scope, validity, or enforceability of any of the Seller's Intellectual Property Rights related to the Business is being, has been or could reasonably be expected to be contested or challenged. To the Knowledge of the Seller, neither Seller's past nor present use of Intellectual Property Rights or Technology owned by the Seller applicable to the Business infringes upon or misappropriates, breaches or otherwise conflicts with the rights of any other Person. The Seller has not received any notice alleging, and otherwise has no Knowledge of (i) the invalidity of, or any limitation on Seller's right to use, any of the Intellectual Property Rights or Technology owned by the Seller applicable to the Business or of (ii) the alleged infringement, misappropriation or breach of any Intellectual Property Rights of others by the Seller applicable to the Business. The Intellectual Property Rights and Technology owned by the Seller applicable to the Business are not subject to any outstanding judgment, decree, order, writ, award, injunction or determination of an arbitrator or court or other Governmental Body affecting the rights of the Seller with respect thereto. To the Knowledge of the Seller, no Person has interfered with, infringed upon or misappropriated any of the Seller's Intellectual Property Rights, or is currently doing so.
- (f) <u>Licenses to Third Parties</u>. <u>Section 3.17(f)</u> of the Disclosure Schedule lists all Contracts pursuant to which any Person has been granted any license under, or otherwise has received or acquired any right (whether or not currently exercisable) or interest in, any Intellectual

Property Rights or Technology of the Seller applicable to the Business. The Seller is not bound by, and no Intellectual Property Rights owned by the Seller applicable to the Business is subject to, any Contract containing any covenant or other provision that in any way limits or restricts the ability of the Seller to use, exploit, assert or enforce any of its Intellectual Property Rights anywhere in the world on behalf of the Business. Without limiting the foregoing, the Seller has not granted any exclusive licenses to the Intellectual Property Rights or Technology owned by the Seller and applicable to the Business.

(g) Protection of Intellectual Property Rights. All of the registrations and pending applications to Governmental Bodies with respect to the Intellectual Property Rights owned by the Seller and applicable to the Business have been timely and duly filed, prosecution for such applications has been attended to, all maintenance and related fees have been paid, and the Seller has taken all other actions required to maintain their validity and effectiveness. The Seller has taken all steps reasonably necessary or appropriate (including, entering into written confidentiality and nondisclosure agreements with officers, directors, subcontractors, employees, licensees and customers) to safeguard and maintain the secrecy and confidentiality of trade secrets that are material to the Business. Without limiting the foregoing, (i) there has been no misappropriation of any trade secrets or other confidential Intellectual Property Rights or Technology used in connection with the Business by any Person; (ii) to the Knowledge of the Seller, no employee, independent contractor or agent of the Seller has misappropriated any trade secrets of any other Person in the course of performance as an employee, independent contractor or agent of the Business and (iii) to the Knowledge of Seller, no employee, independent contractor or agent of the Business and into the Seller is in default or breach of any term of any employment agreement, nondisclosure agreement, assignment of invention agreement or similar agreement or Contract relating in any way to the protection, ownership, development, use or transfer of the Intellectual Property Rights and Technology applicable to the Business. No funding, facilities, or personnel of any Governmental Body or educational institution were used, directly or indirectly, to develop or create, in whole or in part, any of the Seller's Intellectual Property Rights or Technology applicable to the Business. The Seller has not made any submission or suggestion to, and is not subject to any agreement with, any stan

3.18 Employee Plans.

- (a) <u>Disclosure of Employee Plans</u>. <u>Section 3.18</u> of the Disclosure Schedule contains a complete list of all Employee Plans.
- (b) <u>Pension Plans</u>. Neither the Seller nor any ERISA Affiliate of the Seller has sponsored, maintained, participated in or contributed to, and does not now sponsor, maintain, participate in or contribute to (i) any plan subject to Title IV of ERISA or (ii) any plan subject to the minimum funding requirements of Section 412 of the Code. Neither the Seller nor any ERISA Affiliate of the Seller has any unsatisfied liability under Title IV of ERISA or Section 412 of the Code. Each Employee Plan intended to be "qualified" within the meaning of Section 401(a) of the Code is so qualified, its trust is exempt from taxation under Section 501(a) of the Code and it has received a current and valid determination letter (or opinion letter if applicable) from the Internal Revenue Service.

- (c) <u>Multiemployer Plans</u>. No Employee Plan is a "multiemployer plan" as defined in Sections 3(37) or 4001(a)(3) of ERISA ("<u>Multiemployer Plan</u>"), and neither the Seller nor any of its ERISA Affiliates sponsors or has previously sponsored, maintained, contributed to or incurred an obligation to contribute to any Multiemployer Plan.
- (d) <u>Welfare Plans</u>. Each Employee Plan which is a "group health plan," as defined in Section 607(1) of ERISA, has been operated in material compliance with provisions of Parts 6 and 7 of Title I, Subtitle B of ERISA and Section 4980B of the Code and any similar applicable state laws ("<u>COBRA</u>") at all times. Neither the Seller nor any of its ERISA Affiliates sponsors or has previously sponsored or maintained any plan which provides, or has any liability to provide, life insurance, medical, severance or other employee welfare benefits after retirement to any former Business Employee, except as required by COBRA. None of the Employee Plans is a "voluntary employees' beneficiary association," as defined in Section 501(c)(9) of the Code.
- (e) <u>No Liability</u>. Neither the execution nor delivery of this Agreement or the Ancillary Agreements by the Seller, nor the consummation of the transactions contemplated hereby or thereby, will result in any Liability to Parent, DIS or Purchaser under or with respect to any Employee Plan.
- (f) Compliance with Laws. (i) The Seller has not engaged in a prohibited transaction under Section 406 of ERISA or 4975 of the Code; (ii) neither the Seller, any ERISA Affiliate nor, to the Knowledge of the Seller, any third-party fiduciary, has breached its fiduciary responsibility under Part 4 of Title I of ERISA with respect to any Employee Plan, which could result in a material liability to the Seller; (iii) each Employee Plan has been maintained and operated in material compliance with its terms and applicable Regulations, including where applicable ERISA and the Code; (iv) neither the Seller nor its ERISA Affiliates have any material liability for any penalty or Tax under Sections 4971, 4972, 4975, 4976, 4979 or 4980 of the Code or Section 502 of ERISA; (v) other than claims for benefits in the Ordinary Course, there is no material claim pending, or, to the Knowledge of the Seller, threatened involving any Employee Plan; and (vi) no Employee Plan is subject to an ongoing audit or other administrative proceeding of the Internal Revenue Service (the "IRS"), the Department of Labor or any other Governmental Body or has applied for administrative relief under any voluntary compliance program of the IRS, the Department of Labor or any other Governmental Body.
- (g) <u>Deductibility of Payments</u>. There is no Contract, agreement, plan or arrangement covering any Business Employee that individually or collectively requires the payment of any amount (i) that is not deductible under Section 162(a)(1) or 404 of the Code or (ii) that is an "excess parachute payment" pursuant to Section 280G of the Code. Neither the execution and delivery of this Agreement or the Ancillary Agreements by the Seller nor the consummation of the transactions contemplated hereby or thereby will result in the acceleration, increase or creation of any rights of any Business Employee to benefits, whether or not under any Employee Plan (including, without limitation, the acceleration or creation of any rights under any severance, parachute or change in control agreement).

3.19 <u>Transactions with Certain Persons</u>. No officer, director, employee or other Affiliate of the Seller or any member of any such Person's immediate family is presently a party to any transaction relating to the Business, including, without limitation, any Contract (a) providing for the furnishing of services by; (b) providing for the rental of real or personal property from; or (c) otherwise requiring payments to (other than for services as officers, directors or employees of the Seller) any such Person or corporation, partnership, trust or other entity in which any such Person has an interest as a shareholder, officer, director, trustee or partner.

3.20 Tax Matters.

- (a) <u>Filing of Tax Returns</u>. The Seller has timely filed (taking into account any extensions of time for such filings that have been properly and timely requested by the Seller) all Tax Returns that were required to be filed. All such Tax Returns are complete and accurate in all material respects. All Taxes owed by the Seller (whether or not shown on any Tax Return) have been paid. Except as set forth on <u>Section 3.20</u> of the Disclosure Schedule, the Seller is not currently the beneficiary of any extension of time within which to file any Tax Return. To the Knowledge of the Seller, no claim has ever been made by an authority in a jurisdiction in which the Seller does not file Tax Returns that it is or may be subject to taxation by that jurisdiction.
- (b) <u>Audit History; Liens; etc.</u> No deficiencies for Taxes have been claimed, proposed or assessed by any taxing or other Governmental Body against the Seller. There are no pending or, to the Knowledge of the Seller, threatened audits, investigations, disputes or claims for or relating to any additional liability for Taxes with respect to the Seller. There are no matters under discussion with any Governmental Body with respect to Taxes that, in the reasonable judgment of the Seller, are likely to result in an additional liability for Taxes with respect to the Seller. There are no liens, pledges, charges, claims, restrictions on transfer, mortgages, security interests or other encumbrances of any sort (collectively, "<u>Liens</u>") on any Purchased Assets for Taxes (other than for current Taxes not yet due and payable). The Seller has not waived any statute of limitations in respect of Taxes or agreed to any extension of time with respect to a Tax assessment or deficiency. The Seller has previously delivered to the Purchaser true and correct copies of (i) all Tax Returns for the three previous tax years ending prior to the Closing Date and (ii) any examination reports and statements of deficiencies assessed against or agreed to by the Seller within the past five years. To the Knowledge of the Seller, there is no basis for the assertion of any claim for any liabilities for unpaid Taxes for which the Purchaser would become liable as a result of the transactions contemplated by this Agreement or that would result in any Lien on any of the Purchased Assets.

(c) <u>Tax Elections</u>. All material elections with respect to Taxes affecting the Business or the Purchased Assets as of the date hereof, to the extent such elections are not shown on or in the Tax Returns of the Seller that have been delivered to the Purchaser by the Seller prior to the date hereof, are set forth on <u>Section 3.20</u> of the Disclosure Schedule. No new elections with respect to Taxes, or changes in current elections with respect to Taxes, affecting the Business or the

Purchased Assets shall be made after the date of this Agreement without Parent's prior written consent unless otherwise required by law. The Seller (i) has not agreed, and is not required, to make any adjustment under Section 481(a) of the Code by reason of a change in accounting method or otherwise; (ii) has not made an election, and is not required, to treat any Purchased Asset as owned by another Person pursuant to the provisions of former Section 168(f) of the Code or as tax-exempt bond financed property or tax-exempt use property within the meaning of Section 168 of the Code and (iii) has not made any of the foregoing elections and is not required to apply any of the foregoing rules under any comparable state or local Tax provision.

- (d) <u>Prior Affiliated Groups; Taxes of Other Persons</u>. The Seller has never been a member of an affiliated group of corporations within the meaning of Section 1504 of the Code and has no liability for the Taxes of any other Person (i) under Treasury Regulation Section 1.1502-6 (or any similar provision of state, local or foreign law); (ii) as a transferee or successor; (iii) by contract or (iv) otherwise.
- (e) <u>Tax Sharing Agreements</u>. There are no, and at the Closing Date there will be no, tax-sharing agreements or similar arrangements with respect to or involving the Business or the Purchased Assets, and, after the Closing Date, the Seller shall not be bound by any such tax-sharing agreements or similar arrangements or have any liability thereunder for amounts due in respect of periods prior to the Closing Date.
- (f) <u>Withholding</u>. The Seller has withheld and paid (and timely paid over any withheld amounts to the appropriate Taxing authority) all federal and state income taxes, Federal Insurance Contribution Act, Federal Unemployment Tax Act and other Taxes required to have been withheld and paid in connection with amounts paid or owing to any employee, independent contractor, creditor, shareholder or other third party.
- 3.21 <u>Liabilities from Business Combination</u>. The Seller does not have Liabilities due or to become due with respect to the Business arising from the merger or other combination of Ultrascan, Inc. and Ultrascan of Macon, Inc. that would have a Material Adverse Effect on the Purchased Assets or the Business.
- 3.22 <u>Insurance</u>. Section 3.22 of the Disclosure Schedule contains a complete and accurate list of all policies or binders of fire, liability, title, worker's compensation, product liability and other forms of insurance maintained since January 1, 2004 by the Seller on the Business, the Purchased Assets or the Seller's employees that engage in the Business. All insurance coverage applicable to the Business, the Purchased Assets and the Seller's employees that engage in the Business is in full force and effect, insures the Seller in reasonably sufficient amounts against all risks usually insured against by Persons operating similar businesses or properties of similar size in the localities where such businesses or properties are located, provides coverage as may be required by applicable Regulation and by any and all Contracts to which the Seller is a party and has been issued by insurers of recognized responsibility. There is no Default under any such coverage nor has there been any failure to give notice or present any claim under any such coverage in a due and timely fashion. There are no outstanding unpaid premiums except in the Ordinary Course of Business and no notice of cancellation or non-renewal of any such coverage has been received.

There are no provisions in such insurance policies for retroactive or retrospective premium adjustments. Except as set forth on Section 3.22 of the Disclosure Schedule, all products liability, general liability and workers' compensation insurance policies maintained by the Seller relating to the Business have been occurrence policies and not claims-made policies. There are no outstanding performance bonds covering or issued for the benefit of the Seller relating to the Business. There are no facts upon which an insurer might be justified in reducing coverage or increasing premiums on existing policies or binders relating to the Business. No insurer has advised the Seller that it intends to reduce coverage, increase premiums or fail to renew any existing policy or binder relating to the Business.

3.23 Compliance With Environmental Laws. The operations of the Business and each Operating Site comply and have complied in each case in all material respects with applicable Environmental Laws. To the Knowledge of the Seller, no Hazardous Substances are present on, in or under any Operating Site, regardless of how the substance or Hazardous Substances came to rest there, either in violation of any applicable Environmental Law or of a type or in a location, concentration or condition that has resulted, or that is reasonably likely to result, in any Liability to the Seller or any of Parent, DIS or Purchaser under any Environmental Law, Environmental Claim or Environmental Exposure Claim, including any obligation to investigate, clean, remediate or take other similar action pursuant to any Environmental Law. There is no ongoing, and, since January 1, 2000, has not been any, investigation, clean up, remediation or other similar action pursuant to Environmental Law related to the actual or potential presence of any Hazardous Substances on any Operating Site. To the Knowledge of the Seller, no underground storage tanks, PCBs or asbestos-containing materials are located on, in or under any Operating Site. The Seller is not subject to any pending, and has no Knowledge of any threatened or any reasonable basis for, Environmental Claims, against the Seller or related to the Business. No Encumbrances have been, or are, imposed on the Business or any of the Purchased Assets under any Environmental Law. The Seller has obtained all Permits, and has made all reports and notifications, required under any Environmental Law in connection with the Purchased Assets and the operation of the Business, all such Permits are in good standing and the Seller is in material compliance with the terms and conditions of such Permits, and all such Permits, and any corrective action taken, or citations and notices of violations related thereto, are listed on Section 3.23 of the Disclosure Schedule. To the Knowledge of the Seller, no products

3.24 <u>Performance of Services</u>. All services that have been performed by or on behalf of the Seller that relate to the Business were performed in conformity with the terms and requirements of all applicable warranties and other Contracts and with all applicable Regulations. Parent, DIS and Purchaser will not incur or otherwise become subject to any Liability arising from any services performed by or on behalf of the Seller in connection with the Business on or prior to the Closing Date. There is no Action pending or, to the Knowledge of the Seller, being threatened against the Seller relating to any services performed by or on behalf of the Seller in connection with the Business which will or is reasonably likely to result in a Liability, and, to the Knowledge of the Seller, no event has occurred, and no condition or circumstance exists, that could (with or without notice or lapse of time) give rise to or serve as a basis for any such Action.

- 3.25 <u>Customers and Distributors</u>. Section 3.25 of the Disclosure Schedule accurately identifies, and provides an accurate and complete breakdown of the revenue from, all of the customers of the Seller related to the Purchased Assets for the calendar years 2005 and 2006 (up to the date of this Agreement). The Seller has not received any written notice or other written communication indicating that any customer or other Person identified or required to be identified in Section 3.25 of the Disclosure Schedule intends to cease dealing with the Seller or otherwise reduce the volume of business transacted by such Person with the Seller below historical levels and, to the Knowledge of the Seller, no such customer or other Person intends to do so.
- 3.26 <u>Sufficiency of Purchased Assets</u>. The Purchased Assets will constitute, as of the Closing Date, substantially all of the properties, rights, interests and other tangible and intangible assets necessary to enable the Purchaser to (a) own and use the Purchased Assets in the manner in which the Purchased Assets have been used prior to the date hereof, are currently being used and are planned to be used by the Seller to perform its obligations under a Contract to be assumed by the Purchaser and (b) conduct the Business in the manner in which the Business has been conducted prior to the date hereof, is currently being conducted and is currently proposed by the Seller to be conducted. Except as set forth in Section 3.26 of the Disclosure Schedule, no licenses or consents from, or payments to, any other Person are or will be necessary for the Purchaser to use any of the Purchased Assets in substantially the manner in which the Seller has used such Purchased Assets. Except as provided in this Agreement, no restrictions will exist on the Purchaser's right to sell, resell, license or sublicense any of the Purchased Assets or engage in the Business, nor will any such restrictions be imposed on the Purchaser as a consequence of the transactions contemplated by this Agreement or the Ancillary Agreements.
- 3.27 <u>WARN Act</u>. The Seller is not planning or contemplating, and has not made or taken, any decisions or actions concerning the Business that would require the service of notice under the Worker Adjustment and Retraining Act of 1988 or any similar Regulations.
- 3.28 <u>Foreign Corrupt Practices Act</u>. With respect to the Business, neither the Seller nor any predecessor, nor, to the Knowledge of the Seller, any agent, employee or other Person associated with or acting on behalf of the Seller or any predecessor has, directly or indirectly, used any corporate funds for unlawful contributions, gifts, entertainment or other unlawful expenses relating to political activity, made any unlawful payment to foreign or domestic government officials or employees or to foreign or domestic political parties or campaigns from corporate funds, violated any provision of the Foreign Corrupt Practices Act of 1977, as amended, or made any bribe, rebate, payoff, influence payment, kickback or other similar unlawful payment.
- 3.29 Other Agreements. The Seller has not entered into any other agreement or arrangement relating to the sale or other disposition of any of the Purchased Assets other than with respect to sales of Inventory in the Ordinary Course of Business.

- 3.30 <u>Bulk Transfer Laws</u>. The Seller represents (a) that the Purchased Assets represent a majority if not all of the Seller's fixed assets and equipment as of the date hereof and (b) that there are no current or past creditors of the Seller to which any Bulk Transfer Law requires the delivery of notice or from whom any form of consent is required in conjunction with undertaking the transactions contemplated by this Agreement and the Ancillary Agreements other than those notices given or consents that have been obtained.
- 3.31 <u>Redemption of Seller Shares by Seller</u>. None of the information (including information regarding this Agreement, the transactions contemplated hereby, and information concerning the Parent, DIS and the Purchaser) provided, or the representations or warranties made, by the Seller to the Seller's shareholders in connection with the redemption of all outstanding shares of the Seller (other than the Seller's shares held by the Principal Stockholder) contained any untrue statement of a material fact or omitted to state any material fact.
- 3.32 <u>Material Misstatements or Omissions</u>. No representations or warranties by the Seller in this Agreement, nor any document, exhibit, statement, certificate or schedule heretofore or hereafter furnished to Parent, DIS or Purchaser pursuant hereto, or in connection with the transactions contemplated hereby, including without limitation the Disclosure Schedule, contains or will contain any untrue statement of a material fact, or omits or will omit to state any material fact necessary to make the statements or facts contained therein not misleading.

ARTICLE IV. REPRESENTATIONS AND WARRANTIES OF THE PURCHASER

As an inducement to each of the Seller and Principal Stockholder to enter into this Agreement, Parent, DIS and Purchaser hereby severally, and not jointly, represent and warrant to the Seller and the Principal Stockholder, as follows:

- 4.1 <u>Organization</u>. Each of Parent, DIS and Purchaser is duly organized, validly existing and in good standing under the laws of the State of Delaware.
- 4.2 <u>Authorization</u>. Each of Parent, DIS and Purchaser has all requisite power and authority, and has taken or prior to the Closing will have taken all action necessary, to execute and deliver this Agreement and the Ancillary Agreements to which it is a party, to consummate the transactions contemplated hereby and thereby and to perform its obligations hereunder and thereunder. The execution and delivery of this Agreement and the Ancillary Agreements to which it is a party by Parent, DIS and Purchaser and the consummation by Parent, DIS and Purchaser of the transactions contemplated hereby and thereby have been duly approved. No other proceedings on the part of the Parent, DIS and Purchaser are necessary to authorize this Agreement and the Ancillary Agreements to which any of them is a party and the transactions contemplated hereby and thereby. This Agreement has been duly executed and delivered by each of Parent, DIS and Purchaser and is, and upon execution and delivery the Ancillary Agreements to which any of them is a party will each be, a legal, valid and binding obligation of Parent, DIS and Purchaser, as applicable, enforceable against them in accordance with its terms, except as enforcement may be

limited by applicable bankruptcy, insolvency, reorganization, moratorium and other laws affecting enforcement of creditor's rights generally and except insofar as the availability of equitable remedies may be limited by applicable Regulations.

- 4.3 No Conflict or Violation. Neither the execution, delivery or performance of this Agreement or the Ancillary Agreements nor the consummation of the transactions contemplated hereby or thereby, nor compliance by Parent, DIS or Purchaser with any of the provisions hereof or thereof, will (a) violate or conflict with any provision of their respective Certificates of Incorporation or Bylaws, (b) violate, conflict with, or result in or constitute a Default under, or result in the termination of, or accelerate the performance required by, or result in a right of termination or acceleration under, any of the terms, conditions or provisions of any agreement, contract, obligation, promise or undertaking that is legally binding on any of Parent, DIS or Purchaser, or (c) violate, conflict with, contravene or give any Person the right to exercise any remedy or obtain any relief under any Regulation or Court Order.
- 4.4 <u>Consents and Approvals</u>. No Consent is required to be made or obtained by Parent, DIS or Purchaser or any of their respective Affiliates in connection with the execution, delivery and performance by them of this Agreement and the Ancillary Agreements and the consummation of the transactions contemplated hereby and thereby.
- 4.5 <u>SEC Filings</u>. Parent has timely filed all forms, reports, and documents required to be filed by Parent with the Securities and Exchange Commission (the "<u>SEC</u>"), and any other federal, state or local securities regulator and NASDAQ (collectively, the "<u>Parent Securities Filings</u>"). The Parent Securities Filings at the time they were filed (or if amended or superseded by a filing prior to the date of this Agreement, then on the date of such filing) complied in all material respects with the requirements of (i) the Securities Act or the Exchange Act, as the case may be, and the rules and regulations of the SEC thereunder applicable to such Parent Securities Filings; and (ii) the rules and regulations of NASDAQ. The Parent Securities Filings did not at the time they were filed (or if amended or superseded by a filing prior to the date of this Agreement, then on the date of such filing) contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary in order to make the statements therein, in the light of the circumstances under which they were made, not misleading.
- 4.6 <u>No Brokers</u>. Neither Parent, DIS or Purchaser nor any of their respective Representatives has entered into or will enter into any contract, agreement, arrangement or understanding with any broker, finder or similar agent or any Person which will result in the obligation of Parent, DIS, Purchaser, Seller or any of their respective Affiliates to pay any finder's fee, brokerage fees or commission or similar payment in connection with the transactions contemplated by this Agreement.

ARTICLE V. COVENANTS AND AGREEMENTS

5.1 Omitted.

5.2 Access and Investigation. The Seller shall ensure that, at all times after the execution of this Agreement and prior to the Closing: (a) the Seller and its Representatives provide the Purchaser and its Representatives with reasonable access during normal business hours to the Seller's Representatives, personnel and assets and to all existing books, records, Tax Returns, work papers and other documents and information relating to the Business and the Purchased Assets; (b) the Seller and its Representatives provide the Purchaser and its Representatives with such copies of existing books, records, Tax Returns, work papers and other documents and information relating to the Business and the Purchaser as the Purchaser may reasonably request and (c) the Seller and its Representatives compile and provide the Purchaser and its Representatives with such additional financial, operating and other data and information relating to the Business and the Purchased Assets as the Purchaser may reasonably request. Any such access and availability shall include, without limitation, access for the Purchaser and its Representatives, at the Purchaser's sole cost and expense, to conduct any site assessment or inspection in respect of environmental matters at the Operating Site, and physical inspections of the Purchased Assets and each Facility, as the Purchaser may reasonably request.

5.3 Stock Registration.

- (a) <u>Registration upon Offering</u>. If Parent shall determine to register any of its securities either for its own account or the account of a stockholder, other than a registration relating solely to employee benefit plans, a registration relating to the offer and sale of debt securities, a registration relating to a corporate reorganization or other Rule 145 transaction, or a registration on any registration form that does not permit secondary sales, Parent will:
 - (i) promptly give written notice of the proposed registration to the Seller; and
- (ii) use its commercially reasonable efforts to include in such registration (and any related qualification under blue sky laws or other compliance), and in any underwriting involved therein, all shares of its Common Stock that may be issued pursuant to the Earnout Payments ("Earnout Shares") that are not freely tradable under Rule 144 at such time.
- (b) <u>Underwriting</u>. If the registration contemplated in Section 5.3(a) above is an underwritten offering, the Seller (together with the Parent and other holders of securities of the Parent with registration rights to participate therein distributing their securities through such underwriting) shall enter into an underwriting agreement in customary form with the representative of the underwriter or underwriters selected by Parent.

Notwithstanding any other provision of this Section 5.3(b), if the underwriters advise the Parent in writing that marketing factors require a limitation on the number of shares to be underwritten, the underwriters may exclude all Earnout Shares or limit the number of Earnout Shares to be included in, the registration and underwriting. The number of shares of securities that are entitled to be included in the registration and underwriting shall be allocated, as follows: (i) first, to the Parent for securities being sold for its own account, (ii) second, to the Seller to include the Earnout Shares and (iii) third, to the other securities holders requesting to include their shares in such registration statement.

If the Seller does not agree to the terms of any such underwriting, the Seller shall be excluded therefrom. The Earnout Shares so excluded shall also be withdrawn from such registration.

- (c) <u>Right to Terminate Registration</u>. Parent shall have the right to terminate or withdraw any registration initiated by it under this Section 5.3 prior to the effectiveness of such registration.
- 5.4 Filings and Consents. The Seller shall ensure that: (a) the Seller and its Representatives cooperate with the Purchaser and the Purchaser's Representatives, and comply with the requests of the Purchaser with respect to, and make, give and obtain on a timely basis, all filings, notices and Consents required to be made, given and obtained in order to consummate the transactions contemplated by this Agreement and the Ancillary Agreements, including without limitation any such filings, notices or Consents required or otherwise reasonably necessary to transfer or assign all Permits currently held by the Seller related to the Business and (b) at all times after the execution of this Agreement and prior to the Closing, the Seller and its Representatives cooperate with the Purchaser and with the Purchaser's Representatives, and prepare and make available such documents and take such other actions as the Purchaser may request, in connection with any filing, notice or Consent that the Purchaser is required or elects to make, give or obtain.
- 5.5 <u>Reasonable Efforts</u>. At all times following the execution of this Agreement and prior to the Closing, the Seller shall use its commercially reasonable efforts to cause the conditions set forth in Article VI of this Agreement to be satisfied on a timely basis and the Purchaser shall use its commercially reasonable efforts to cause the conditions set forth in Article VII of this Agreement to be satisfied on a timely basis.

5.6 Certain Tax Matters.

- (a) The Purchaser and the Seller agree to furnish or cause to be furnished to the other, upon request, as promptly as practicable, such information and assistance relating to the Purchased Assets, including, without limitation, access to books and records, as is reasonably necessary for the filing of all Tax Returns by the Purchaser or the Seller, the making of any election relating to Taxes, the preparation for any audit by any taxing authority, and the prosecution or defense of any claim, suit or proceeding relating to any Tax. The Purchaser and the Seller shall cooperate with each other in the conduct of any audit or other proceeding relating to Taxes involving the Purchased Assets.
- (b) Subject to Section 5.6(c) below, Seller will be responsible for the preparation and filing of all Tax Returns of Seller (including Tax Returns required to be filed after the Closing Date) to the extent such Tax Returns include or relate to Seller's operation of the Business or Seller's use or ownership of the Purchased Assets on or prior to the Closing Date. Seller's Tax Returns to the extent they relate to the Business or Purchased Assets shall be true, complete and correct and prepared in accordance with applicable law. Seller will be responsible for and make all payments of Taxes shown to be due on such Tax Returns to the extent they relate to the Purchased Assets or the Business on or prior to the Closing Date.

- (c) In the case of any real or personal property taxes (or other similar Taxes) attributable to the Purchased Assets for which Taxes are reported on a Tax Return covering a period commencing before the Closing and ending thereafter (a "<u>Straddle Period Tax</u>"), any such Straddle Period Taxes shall be prorated between the Purchaser and Seller on a per diem basis. The party required by law to pay any such Straddle Period Tax (the "<u>Paying Party</u>") shall file the Tax Return related to such Straddle Period Tax within the time period prescribed by law and shall timely pay such Straddle Period Tax. To the extent any such payment exceeds the obligation of the Paying Party hereunder, the Paying Party shall provide the other party (the "<u>Non-Paying Party</u>") with notice of payment and a copy of such return establishing such Straddle Period Tax, and within 10 days of receipt of such notice of payment, the Non-Paying Party shall reimburse the Paying Party for the Non-Paying Party's share of such Straddle Period Taxes. Either the Purchaser or he Seller shall promptly notify the other party in writing upon receipt of notice of any pending or threatened United States federal, state, local or foreign Tax audits or assessments relating to the income, properties or operations of the Seller that reasonably may be expected to relate to the Purchased Assets.
- (d) Any payments made to any party pursuant to Article IX shall constitute an adjustment to the Purchase Price for Tax purposes and shall be treated as such by the Seller and Purchaser on their Tax Returns to the extent permitted by applicable Regulations.
- (e) Seller will be responsible for and make all payments of Taxes related to the merger or other business combination of Ultrascan, Inc. and Ultrascan of Macon, Inc.

5.7 Employee Matters.

- (a) Prior to the Closing Date, the Purchaser will offer to employ all of the Seller's employees who are engaged in the Business as of the date hereof, except for certain employees to be determined by the Purchaser following Closing. Such offers of employment (the "Employee Offers") shall (i) be contingent upon the occurrence of the Closing and such employees satisfying normal employment conditions of Purchaser and (ii) terminate automatically in the event that the Closing does not occur in accordance with this Agreement. Those Employees who accept employment with the Purchaser are the "Hired Employees."
- (b) Except as required by any applicable law or as may be agreed to by the Purchaser and the Seller, as of the Closing Date, the Purchaser shall not assume any severance obligations or any Employee Plans, including equity compensation plans, for the Hired Employees. For the avoidance of doubt, the Seller shall retain sponsorship of, and all Liabilities under, the Employee Plans. The Purchaser shall have no Liabilities with respect to the Employee Plans, and shall not be responsible for any Liabilities with respect to such Employee Plans, and the Seller shall cause each Employee Plan to be amended as necessary to ensure that the Purchaser has no Liabilities with respect to the Employee Plans following the Closing Date. The Seller shall retain sole responsibility for all Liabilities relating to, or in connection with, the employment, or termination of employment, of each Hired Employee of the Seller on or prior to the Closing Date.

- (c) On the Closing Date, subject to applicable Regulations, the Seller shall terminate the employment of each Hired Employee; *provided*, *however*, that the Seller shall not be required hereunder to terminate the employment of any employee (i) whom the Purchaser does not intend to hire because such person fails to satisfy any normal employment conditions of the Purchaser or (ii) for whom the Purchaser has retracted an Employee Offer. For the avoidance of doubt, the Purchaser shall not be obligated to hire or employ any employee of the Seller engaged in the Business who does not accept an Employee Offer. On or prior to the Closing, the Seller shall pay all amounts accrued for vacation for the Hired Employees to the extent such Hired Employees have not validly elected to have their vacation accruals assumed by Purchaser.
- (d) Seller shall be responsible for any and all liability under the WARN Act or under any state, local or foreign Regulation concerning layoffs or the closing or relocation of worksites or the like which arises out of or results from any termination of employment by the Seller on or before the Closing Date.
- (e) The Purchaser and the Seller each covenant and agree, as soon as administratively feasible following the Closing, as follows: (i) to facilitate a distribution of the account balance of each Hired Employee in the Seller's 401(k) plan upon the election of such Hired Employee and (ii) to make reasonable efforts to take any and all actions necessary to accomplish the distribution and rollover to the Purchaser's 401(k) plan, to the extent allowed by Regulation, of amounts distributed from the Seller's 401(k) plan with respect to the Hired Employees. The Purchaser agrees to accept rollovers of distributions from the Seller's 401(k) plan with respect to the Employees, to the extent allowed by Regulation and the Purchaser's 401(k) plan.
- (f) The Purchaser and the Seller acknowledge and agree that all provisions contained in this Section 5.7 with respect to the Business Employees are included for the sole benefit of the Purchaser and the Seller, and that nothing herein, whether express or implied, shall create any third party beneficiary or other rights (i) in any other Person, including, without limitation, any Business Employees, former Business Employees, any participant in any Employee Plan, or any dependent or beneficiary thereof, or (ii) to continued employment with the Purchaser or the Seller, or any of their respective Affiliates.
- (g) The Purchaser shall prepare and furnish to each of the Hired Employees a Form W-2 that shall reflect all wages and compensation paid to such employee for the entire calendar year in which the Closing Date occurs. Seller shall furnish to the Purchaser the Forms W-4 and W-5 of each such Hired Employee for the portion of the calendar year up to and including the Closing Date. The Purchaser shall send to the appropriate Social Security Administration office a duly completed Form W-3 and accompanying copies of the duly completed Forms W-2. It is the intent of the parties hereunder that the obligations of The Purchaser and Seller under this Section 5.7(g) shall be carried out in accordance with Section 5 of Revenue Procedure 2004-53.
- 5.8 <u>Cessation of Use of Names and Web Site</u>. After the Closing, the Seller shall cease using the name "Ultrascan", any other trade names used by the Business and the "www.ultrascaninc.com" website.

5.9 Non-Competition/Non-Solicitation.

- (a) In consideration for the amount paid for the purchase of the goodwill of the Business and the execution of the employment agreements attached hereto as Exhibit B, from and after the Closing Date until the ***** anniversary of the Closing Date, the Seller shall not, and shall not permit any of its stockholders (including the Principal Stockholder), directors, officers, and Subsidiaries, directly or indirectly, to (i) offer or provide in ****** any products or services that are substantially similar to the products and services provided as of the Closing by the Business in ******* or (ii) grant any license to any of the intellectual property listed on Section 3.17(a) of the Disclosure Schedule to any Person that provides products or services that are substantially similar to the products and services provided as of the Closing by the Business. Purchaser and the Principal Stockholder shall enter into a separate agreement set forth on Exhibit A-1 that further obligates the Principal Stockholder to comply with the covenants set forth in this Section 5.9 regarding his inability to compete with the Purchaser and the Business (the "Protective Covenant Agreement"). Purchaser shall enter into an agreement, attached hereto as Exhibit D, with Seller's stockholders (including the Principal Stockholder), directors, and officers, providing that none of them will, directly or indirectly, compete with the Business as described above in exchange for the consideration duly paid. In connection with the termination of key employee Matthew G. Molchan, the Seller shall deliver at Closing the Molchan Settlement Agreement which is attached hereto as Exhibit K.
- (b) From and after the Closing Date until the fifth (5th) anniversary of the Closing Date, the Seller shall not, and shall not permit any of its stockholders (including the Principal Stockholder), directors, officers and Subsidiaries, directly or indirectly, to solicit, encourage or induce any Person employed as of the Closing Date by the Purchaser or any of its Subsidiaries to leave the employment of the Purchaser or any of its Subsidiaries. From and after the Closing Date until the fifth (5th) anniversary of the Closing Date, the Purchaser shall not, and shall not permit any of its Subsidiaries, directly or indirectly, to solicit, encourage or induce any Person employed as of the Closing Date by the Seller or any of its Subsidiaries to leave the employment of the Seller or any of its Subsidiaries.
- (c) Notwithstanding any other provision of this Agreement, it is understood and agreed that remedies at law would by themselves be inadequate in the case of any breach of the covenants contained in this Section 5.9, and the parties shall be entitled to equitable relief in respect thereof, including the remedy of specific performance, with respect to any breach of such covenants, without the need to show any proof of damages.
- 5.10 Estoppel Certificates. The Seller shall deliver from the landlord under each Lease, a certificate stating: (a) the Lease with such landlord is in full force and effect; (b) to landlord's knowledge, the Seller is not in default thereunder; (c) annexed to such certificate is a correct and complete copy of such Lease; and (d) such other matters as the Purchaser shall reasonably require.
- 5.11 <u>Delivery of Title Commitment and Survey</u>. As soon as reasonably practicable, but in no event later than five (5) days after the date hereof, the Seller shall obtain and

cause to be delivered to the Purchaser (at the Seller's expense) with respect to each material Leased Real Property: (a) a current title insurance commitment (together with copies of all documents referenced therein) issued by a title company acceptable to the Purchaser; and (b) a new or updated ALTA survey to be prepared by a licensed surveyor or engineer acceptable to the Purchaser.

- 5.12 <u>Delivery of Title to Leased Equipment</u>. As soon as reasonably practicable, the Seller shall obtain and cause to be delivered to the Purchaser, with respect to each material Leased Equipment and the Vehicles, any and all documents required to transfer such leases to the Purchaser.
- 5.13 *******. As soon as reasonably practicable and, in any event, no later than *** days from the Closing, the Seller shall enter in to the ***** with ***** or such other ***** affiliated entity.
- 5.14 ******. As soon as reasonably practicable and, in any event, no later than **** days from the Closing, the Seller shall obtain the ***** to assign the July 1, 2002 Leasing Agreement by and between ****** and the Seller to the Purchaser.
- 5.15 <u>Tax Disclosure Agreement</u>. The Seller shall deliver to Purchaser a voluntary tax disclosure agreement within one hundred and eighty (180) days from the Closing.
- 5.16 <u>Further Assurances</u>. The Purchaser, the Seller's Subsidiaries and the Principal Stockholder shall cooperate reasonably with each other in connection with any steps required to be taken as part of their respective obligations under this Agreement, and shall (a) furnish upon request to each other such further information, (b) execute and deliver to each other such other documents, and (c) do such other acts and things, all as the other party may reasonably request for the purpose of carrying out the intent of this Agreement and the transactions contemplated hereby.

ARTICLE VI. CONDITIONS PRECEDENT TO OBLIGATIONS OF THE PURCHASER

- 6.1 <u>Conditions</u>. The obligations of the Purchaser to consummate the transactions provided for hereby are subject, in the discretion of the Purchaser, to the satisfaction, on or prior to the Closing Date, of each of the following conditions, any of which may be waived by the Purchaser:
- (a) the representations and warranties in Article III shall be true and correct in all material respects when made and at and as of the Closing Date as if such representations and warranties were made at such time (except that those representations and warranties which are made as of a specific date shall be true and correct only as of such date);
- (b) Seller and Principal Stockholder shall have performed and satisfied in all material respects all agreements and covenants required hereby to be performed or satisfied by them prior to or at the Closing Date;
- (c) all Consents, filings, registrations and notifications necessary to permit the consummation of the transactions contemplated by this Agreement and the Ancillary Agreements shall have been obtained or made, including the Requisite Stockholder Approval of such transactions;

- (d) no Court Order, Action, investigation or proceeding shall have been instituted or threatened which makes the transactions contemplated by this Agreement or the Ancillary Agreements illegal or otherwise prohibited or that otherwise adversely affects the right or ability of the Purchaser to own, operate or control the Business or the Purchased Assets or seeks damages in connection therewith;
- (e) following the date of this Agreement, there shall not have occurred any event, change or condition that, individually or in the aggregate, has had or would reasonably be expected to have a Material Adverse Effect;
- (f) Seller and the Principal Stockholder shall have delivered the documents required to be delivered by it pursuant to Section 8.1(a) of this Agreement, in form and content satisfactory to the Purchaser, and the Ancillary Agreements shall be in full force and effect;
- (g) no regulatory change shall have been instituted or threatened which makes the transactions contemplated by this Agreement or the Ancillary Agreements illegal or otherwise prohibited or that otherwise adversely affects the right or ability of the Purchaser to own, operate or control the Business or the Purchased Assets or seeks damages in connection therewith;
- (h) Ultrascan, Inc. and Ultrascan of Macon, Inc. shall have been merged or otherwise combined into the Seller as permitted under Georgia law;
 - (i) no declaration of non-coverage by any third party payor shall have been made or threatened;
- (j) all Consents, approvals and actions of, filings with and notices to any governmental or regulatory authority necessary to permit Seller and the Principal Stockholder to perform their obligations under this Agreement and the Ancillary Agreements and to consummate the transactions contemplated hereby and thereby shall have been duly obtained, made or given, and all terminations or expirations of waiting periods imposed by any governmental or regulatory Authority necessary for the consummation of the transactions contemplated by this Agreement and the Ancillary Agreements, shall have occurred;
 - (k) the Seller shall have received the Requisite Stockholder Approval; and
 - (1) the Ancillary Agreements contemplated hereunder, shall have been delivered in form and content satisfactory to the Purchaser.

ARTICLE VII. CONDITIONS PRECEDENT TO OBLIGATIONS OF THE SELLER

- 7.1 <u>Conditions</u>. The obligations of the Seller to consummate the transactions provided for hereby are subject, in the discretion of the Seller, to the satisfaction, on or prior to the Closing Date, of each of the following conditions, any of which may be waived by the Seller:
- (a) the representations and warranties in Article IV shall be true and correct in all material respects when made and at and as of the Closing Date as if such representations and warranties were made at such time (except that those representations and warranties which are made as of a specific date shall be true and correct only as of such date);
- (b) Parent, DIS and Purchaser shall have performed and satisfied in all material respects all agreements and covenants required hereby to be performed or satisfied by them prior to or at the Closing Date;
 - (c) the Seller shall have received the Requisite Stockholder Approval;
- (d) no Court Order, Action, investigation or proceeding shall have been instituted or threatened which makes the transactions contemplated by this Agreement or the Ancillary Agreements illegal or otherwise prohibited; and
- (e) Parent, DIS and Purchaser shall have delivered the documents required to be delivered by them pursuant to Section 8.1(b), and the Ancillary Agreements shall be in full force and effect.

ARTICLE VIII. CLOSING

- 8.1 Deliveries at Closing. On the Closing Date:
- (a) <u>Deliveries by the Seller to the Purchaser</u>. The Seller and Principal Stockholder shall execute (where applicable) and deliver, or cause to be delivered, to the Purchaser:
 - (i) the Ancillary Agreements to which it or he is a party;
- (ii) a certificate executed by the Secretary of the Seller, dated as of the Closing Date, certifying the following matters: (A) resolutions adopted by the Seller's board of directors relating to the transactions contemplated by this Agreement and the Ancillary Agreements; and (B) the resolutions of the shareholders of the Seller as in effect as of the date thereof;
- (iii) a certificate executed by a duly authorized officer of the Seller, dated as of the Closing Date, certifying that each of the conditions set forth in Sections 6.1(a), (b), and (c) has been satisfied or waived in all respects;
 - (iv) the non-competition and non-solicitation agreements contemplated by Section 5.9 of this Agreement and attached hereto;
 - (v) the Consent of Hilie King attached hereto as Exhibit L;

- (vi) the Molchan Settlement Agreement;
- (vii) the consent of 1048 Industrial Court LLC to assign the lease for the premises currently occupied by Seller;
- (viii) such other documents, conveying instruments and items as the Purchaser may reasonably request in connection with the transactions contemplated by this Agreement and the Ancillary Agreements, including evidence of termination of existing UCC financing statements or pay-off letters related to such UCC financing statements; and
- (ix) all certificates, agreements and other documents as are customary, or reasonably requested by a title company, in connection with the transfers of Real Property, including without limitation, owner's affidavits, non-foreign status certificates and transfer tax declarations.
- (b) <u>Deliveries by the Purchaser to the Seller</u>. Parent, DIS and Purchaser shall execute (where applicable) and deliver, or cause to be delivered, to the Seller:
 - (i) the Ancillary Agreements to which it is a party;
- (ii) a certificate executed by the Secretary of the Purchaser, dated as of the Closing Date, certifying the following matters: (A) resolutions adopted by the Purchaser's board of directors relating to the transactions contemplated by this Agreement and the Ancillary Agreements;
- (iii) a certificate executed by an officer of the Purchaser, dated as of the Closing Date, certifying that each of the conditions set forth in Sections 7.1(a) and (b) has been satisfied or waived in all respects; and
- (iv) such other documents and items as the Seller may reasonably request in connection with the transactions contemplated by this Agreement and the Ancillary Agreements.
- (c) <u>Closing Payment</u>. The Purchaser shall deliver the Closing Payment and establish the Closing Escrow Fund in accordance with the provisions of Article II of this Agreement.

ARTICLE IX. INDEMNIFICATION

9.1 Survival of Representations and Warranties and Indemnification Obligations

(a) All statements contained in Articles III and IV of this Agreement, the Disclosure Schedule or in any certificate delivered by or on behalf of the parties pursuant to Section 8.1 of this Agreement shall be deemed to be representations and warranties by the parties

hereunder. The representations and warranties of the parties contained herein shall survive the Closing Date until the ****** day after the Closing Date; provided, however, that the representations and warranties of the Seller and the Principal Stockholder contained in Sections 3.1, 3.2, 3.3, 3.12, 3.13, 3.14, 3.15, 3.17, 3.18, 3.20, 3.22, 3.23 and 3.27 of this Agreement and the representations and warranties of Parent, DIS and Purchaser contained in Sections 4.1 and 4.2 of this Agreement shall survive until sixty (60) days after the expiration of the applicable statutes of limitation (including any extensions thereof) with respect thereto (each such applicable date, the "Expiration Date"). The right to indemnification or other remedy based on the representations, warranties, covenants and agreements herein will not be affected by any investigation conducted with respect to, or any Knowledge acquired (or capable of being acquired) at any time, whether before or after the execution and delivery of this Agreement or the Closing, with respect to the accuracy or inaccuracy of or compliance with, any such representation, warranty, covenant or agreement. Without limiting the foregoing, nothing disclosed on the Disclosure Schedule shall in any way affect the Purchaser's right to indemnification under Sections 9.2(a)(ii) through (vii) hereof. If written notice of a claim meeting the requirements of Section 9.3 below has been given prior to the Expiration Date by a party in whose favor such representations and warranties were made, then the relevant representations and warranties shall survive as to such claim, until the claim has been finally resolved.

(b) The obligations of the Seller and the Principal Stockholder to indemnify the Purchaser Indemnified Parties pursuant to Section 9.2(a)(i) of this Agreement and the obligations of Parent, DIS and Purchaser to indemnify the Seller Indemnified Parties pursuant to Section 9.2(b)(i) of this Agreement shall terminate on the Expiration Date; *provided*, *however*, that the obligations of Parent, DIS and Purchaser, on the one hand, or the Seller and the Principal Stockholder, on the other hand, as the case may be, to indemnify and hold harmless any Indemnified Party shall not terminate with respect to any claim as to which such Indemnified Party shall have, before the Expiration Date, delivered a written notice to the Purchaser or the Seller, as the case may be, in accordance with Section 9.3(a) of this Agreement ("Notice of Claim").

9.2 Indemnification.

(a) Subsequent to the Closing, the Seller and the Principal Stockholder shall jointly and severally indemnify the Parent, DIS, Purchaser, and each of their respective Affiliates, officers, directors, employees, stockholders, partners and agents, as the case may be ("Purchaser Indemnified Parties"), against, and hold each Purchaser Indemnified Party harmless from, any damage, claim, loss, cost, liability or expense, including without limitation, interest, penalties, reasonable attorneys' fees and expenses of investigation, lost profits, damages, diminution in value, response action, removal action or remedial action (collectively "Damages") incurred by such Purchaser Indemnified Party, that are incident to, arise out of, in connection with or related to, whether directly or indirectly: (i) any breach of any representation or warranty of the Seller contained in this Agreement, the Disclosure Schedule or in any certificate delivered by or on behalf of the Seller pursuant to Section 8.1 of this Agreement; (ii) any breach or non-performance by the Seller of any of its covenants or agreements contained in this Agreement; (iii) the operation of the Business prior to the Closing; (iv) Pre-Closing Environmental Matters; (v) Pre-Closing Healthcare

Matters; (vi) any noncompliance by Seller with any Bulk Sales Laws or fraudulent transfer law in connection with the transactions contemplated under this Agreement; (vii) any Liabilities of the Seller, or any member of any consolidated, affiliated, combined or unitary group of corporations of which the Seller is or has been a member, for Taxes and any Liabilities for Taxes attributable to the Purchased Assets or the operation of the Business for any Pre-Closing Tax Period; (viii) any Liabilities of the Seller other than the Assumed Liabilities or (ix) any Liability to which the Purchaser or any of the other Purchaser Indemnified Parties may become subject that arises from or relates to any failure to comply with any bulk transfer law or similar Regulations in connection with any of the transactions contemplated by this Agreement and the Ancillary Agreements.

- (b) Subsequent to the Closing, Parent shall indemnify the Seller and its respective Affiliates, officers, directors, employees, stockholders, partners and agents, as the case may be, including, without limitation, the Principal Stockholder ("Seller Indemnified Parties"), against, and hold each of the Seller Indemnified Parties harmless from, any Damages incurred by such Seller Indemnified Party that are incident to, arise out of, in connection with or related to, whether directly or indirectly: (i) any breach of any representation or warranty of Parent, DIS or Purchaser contained in this Agreement or in any certificate delivered by or on behalf of any of them pursuant to Section 8.1 of this Agreement; (ii) any breach or non-performance by the Purchaser of its covenants or agreements contained in this Agreement; (iii) any Assumed Liability; or (iv) the operation of the Business at and after the Closing.
- (c) The term "<u>Damages</u>" as used in this Section 9.2 is not limited to matters asserted by third parties against Seller Indemnified Parties or Purchaser Indemnified Parties, but includes Damages incurred or sustained by such Persons in the absence of third-party claims, and payments by the Indemnified Party shall not be a condition precedent to recovery.

9.3 Notice of Claims.

- (a) As used herein, the term "Claim" shall mean a claim for indemnification under this Article IX, "Indemnified Party." shall mean any Parent Indemnified Party or Seller Indemnified Party making a claim for indemnification pursuant to Sections 9.2(a) and 9.2(b) respectively, and "Indemnifying Party" shall mean the Closing Escrow Fund, in the case of a claim brought pursuant to Sections 9.2(a), or Parent, in the case of a claim brought pursuant to Section 9.2(b). Subject to the terms of this Agreement, the Indemnified Party shall deliver a Notice of Claim to the Indemnifying Party (with a copy to the Escrow Agent) promptly after such Indemnified Party becomes aware of the existence of any potential claim by such Indemnified Party for indemnification from the Indemnifying Party under this Article IX; provided, however, no Notice of Claim shall be required in connection with any dispute relating to the determination of Working Capital, which disputes shall be governed by the procedures set forth in Section 2.6.
- (b) Each Notice of Claim by an Indemnified Party given pursuant to Section 9.3(a) shall contain the following information: (i) that such Indemnified Party has directly or indirectly incurred, paid or properly accrued, or sustained or, in good faith, believes it is reasonably likely that it shall have to directly or indirectly incur or sustain, Damages in an aggregate stated amount arising from such Claim (which amount may be the amount of damages claimed by a third

party in an action brought against such Indemnified Party based on alleged facts, which if true, would give rise to liability for Damages indemnifiable to such Indemnified Party under this Article IX); and (ii) a brief description, in reasonable detail (to the extent reasonably available to such Indemnified Party), of the facts, circumstances or events giving rise to the alleged Damages based on such Indemnified Party's good faith belief thereof, including the identity and address of any third-party claimant (to the extent reasonably available to such Indemnified Party) and copies of any formal demand or complaint, the amount of Damages, the date each such item was incurred, paid or properly accrued, or sustained, or the basis for such anticipated liability, and the specific nature of the breach to which such item is related.

(c) An Indemnified Party may submit a Notice of Claim at any time during the period commencing with the Closing Date and ending on the Expiration Date, but shall not be permitted to bring a Notice of Claim at any time after the Expiration Date (and any delivery or attempted delivery of a Notice of Claim after such time shall be void and of no force or effect). Notwithstanding anything contained herein to the contrary, any Claims for Damages specified in any Notice of Claim delivered to an Indemnifying Party prior to expiration of the Expiration Date shall remain outstanding until such Claims for Damages have been resolved or satisfied, notwithstanding the passage of the Expiration Date. Until the Expiration Date, no delay on the part of an Indemnified Party in giving the Indemnifying Party a Notice of Claim shall relieve the Indemnifying Party from any of its obligations under this Article IX unless (and then only to the extent that) the Indemnifying Party is materially prejudiced thereby.

9.4 Resolution of Notice of Claim. Each Notice of Claim given by an Indemnified Party shall be resolved as follows:

(a) If, within thirty (30) days after a Notice of Claim is received by the Indemnifying Party, the Indemnifying Party does not contest such Notice of Claim in writing to the Indemnified Party delivering such Notice of Claim, the Indemnifying Party shall be conclusively deemed to have consented to the recovery by the Indemnified Party of the full amount of Damages specified in the Notice of Claim in accordance with this Article IX.

(b) If the Indemnifying Party gives the Indemnified Party delivering a Notice of Claim written notice contesting all or any portion of such Notice of Claim (a "Contested Claim") (with a copy to the Escrow Agent) within the thirty (30) day period specified in Section 9.4(a), then such Contested Claim shall be resolved by either (i) a written settlement agreement executed by Parent and Seller (a copy of which shall be furnished to the Escrow Agent, if applicable) or (ii) in the absence of such a written settlement agreement within thirty (30) days following receipt by the Indemnified Party of the written notice from the Indemnifying Party, by arbitration between Parent and Seller in accordance with the terms and provisions of Section 9.4. In the event that an Indemnified Party shall prevail in any such arbitration, the Indemnified Party shall be entitled to recover, in addition to any other rights they may have, the amount of Damages awarded in such arbitration. The prevailing party's recovery shall be limited to the forfeiture of that portion of the Closing Escrow Fund having an aggregate value equal to such awarded Damages. In the case of claims brought pursuant to Section 9.2(b), such recovery shall be by payment of additional cash equal to the value of such awarded Damages to Seller.

(c) Arbitration.

(i) Contested Claims under Section 9.4(b)(ii) shall be submitted for arbitration unless the amount of the Damages is at issue in pending litigation with a third party, in which event arbitration shall not be commenced until such amount is ascertained or both parties agree to arbitration; and in either such event the matter shall be settled by arbitration conducted by one arbitrator mutually agreeable to Parent and Seller. The arbitrator shall set a limited time period and establish procedures designed to reduce the cost and time for discovery while allowing the parties an opportunity, adequate in the sole judgment of the arbitrator to discover relevant information from the opposing parties about the subject matter of the dispute. The arbitrator shall rule upon motions to compel or limit discovery and shall have the authority to impose sanctions, including attorneys' fees and costs, to the extent as a competent court of law or equity, should the arbitrator determine that discovery was sought without substantial justification or that discovery was refused or objected to without substantial justification. The decision of the arbitrator as to the validity and amount of any Contested Claim shall be binding and conclusive upon the parties to this Agreement. Such decision shall be written and shall be supported by written findings of fact and conclusions, which shall set forth the award, judgment, decree or order awarded by the arbitrator.

(ii) Judgment upon any award rendered by the arbitrator may be entered in any court having jurisdiction. Any such arbitration shall be held in the County of San Diego, under the rules then in effect of JAMS or the American Arbitration Association. The arbitrator shall determine how all expenses relating to the arbitration shall be paid, including without limitation, payment of the prevailing party's expenses, the fees of the arbitrator and the administrative fee of JAMS or the American Arbitration Association.

9.5 Release of Closing Escrow Fund.

(a) Subject to the provisions of Section 9.1, the Closing Escrow Amount then held by the Escrow Agent shall be released by the Escrow Agent to the Seller at 5:00 p.m., local time at Parent's headquarters, on the ***** day following the Closing (the "Release Date"); provided, however, that the escrow period shall not terminate with respect to any amount which, in the reasonable judgment of Parent, is necessary to satisfy any unsatisfied claims specified in any Notice of Claim theretofore delivered to the Escrow Agent and, in the case of claims made pursuant to Section 9.2(a), the Principal Stockholder, prior to the Release Date with respect to facts and circumstances existing prior to the Release Date and unresolved prior to the Release Date ("Unresolved Claims"). On the Release Date, the Escrow Agent shall deliver the entire remaining portion of the Closing Escrow Fund (other than amounts set forth in any such pending Notices of Claim to satisfy any Unresolved Claims), to the Seller; provided, however, that in the event the ******** is not executed within **** days following the Closing, the Escrow Agent shall release \$******* from the Closing Escrow Amount to the Purchaser and/or Parent including any interest thereon minus any unpaid fees due and owing to the Escrow Agent on the Release Date. As soon as any Unresolved Claims have been resolved, the Escrow Agent shall deliver the remaining portion of the Closing Escrow Amount, if any, no longer required to satisfy such previously Unresolved Claims in the manner set forth in the preceding sentence.

(b) The ****** Escrow Amount shall be released by the Escrow Agent to the Seller at 5:00 p.m., local time at Parent's headquarters, upon the execution of the ****** by ****** or ***** and the Seller; provided, however, that in the event the ****** are not executed within ***** days following the Closing, the Escrow Agent shall release the ***** Escrow Amount to the Purchaser and/or Parent including any interest thereon minus any unpaid fees due and owing to the Escrow Agent.

9.6 Third-Party Claims. In the event Parent becomes aware of a third party claim (a "Third Party Claim") which Parent reasonably believes may result in a demand for indemnification pursuant to this Article IX Parent shall notify Seller of such claim, and Seller shall be entitled on behalf of the Indemnifying Parties, at its expense, to participate in, but not to determine or conduct, the defense of such Third Party Claim. Parent shall have the right in its sole discretion to conduct the defense of, and to settle, any such Third Party Claim; provided, however, that except with the prior written consent of such settlement by Seller (which consent shall not be unreasonably withheld or delayed), no settlement of any such Third Party Claim with third party claimants shall be determinative of the amount of Damages relating to such matter. In the event that the Seller has consented to any such settlement, the Indemnifying Parties shall have no power or authority to object under any provision of this Article IX to the amount of any Third Party Claim by Parent against the Closing Escrow Fund or special Escrow Fund with respect to such settlement.

9.7 Limitation on Indemnity; Remedies; Maximum Liability.

- (a) Notwithstanding the foregoing, an Indemnifying Party shall not be obligated to indemnify any Indemnified Party under Sections 9.2(a) or 9.2(b) of this Agreement, as the case may be, unless and until the aggregate of all Damages suffered by Purchaser Indemnified Parties or Seller Indemnified Parties, as the case may be, hereunder exceeds \$***** (the "Threshold Amount"), whereupon, provided the other requirements of this Article IX have been complied with, the amount of all such Damages, and all subsequent Damages, shall become due and payable.
- (b) Prior to the Release Date, Purchaser Indemnified Parties shall have the right to offset the amount of any claims for indemnification against the cash held in the Closing Escrow Fund (subject to the terms and conditions of the Escrow Agreement). Subsequent to the Release Date, Purchaser Indemnified Parties shall have the right to demand and receive from the Seller the Purchase Price paid to the Seller as of the date such Damages are awarded, and to offset the amount of any claims for indemnification against the Earnout Payments (if any) payable to Seller. Claims for indemnification against the Purchase Price, Earnout Payments (if any) and Closing Escrow Fund shall be the sole and exclusive remedy for the Purchaser Indemnified Parties with respect to claims relating to any of the matters set forth in Section 9.2(a), except in the case of claims of common law fraud, intentional misrepresentation or criminal activity.
- (c) Except in the case of common law fraud, intentional misrepresentation or criminal activity, the aggregate indemnification obligations of the Seller and Principal

Stockholder pursuant to Section 9.2(a) of this Agreement shall be limited to a maximum amount equal to (x) the aggregate Cash Consideration paid to Seller, plus (y) the aggregate Earnout Payments (if any) paid to Seller. Except in the case of common law fraud, intentional misrepresentation or criminal activity, the aggregate indemnification obligations of Parent, DIS and Purchaser pursuant to Section 9.2(b) of this Agreement shall be limited to a maximum amount equal to (x) the Cash Consideration, plus (y) the aggregate principal amount of the Assumed Liabilities.

ARTICLE X. TERMINATION

10.1 Termination.

- (a) Except as provided in Section 10.1(b), this Agreement may be terminated at any time prior to the Closing:
 - (i) by mutual agreement of Parent and Seller;
 - (ii) by Parent or Seller, if the Closing Date shall not have occurred by April 30, 2007;
- (iii) by Parent or Seller, if the Requisite Stockholder Approval has not been obtained by written consent or at a meeting of Seller's stockholders duly called and held in accordance with the Seller's Articles of Incorporation, or any postponement or adjournment thereof;
- (iv) by Parent or Seller, if (i) a court of competent jurisdiction or other Governmental Authority shall have issued a nonappealable final order, decree or ruling or taken any other action, in each case having the effect of permanently restraining, enjoining or otherwise prohibiting any material transaction contemplated by this Agreement, or (ii) any statute, rule, regulation or order is enacted, promulgated or issued by any Governmental Authority that would make consummation of any material transaction contemplated by this Agreement illegal;
- (v) by Parent, if (i) it is not in material breach of its obligations under this Agreement and (ii) there has been a breach of any representation, warranty, covenant or agreement of Seller or the Principal Stockholder contained in this Agreement such that the conditions set forth in Section 8.1(a) would not be satisfied at the time of such breach and such breach has not been cured within ten (10) business days after written notice thereof to the Seller or the Principal Stockholder; *provided*, *however*, that no cure period shall be required for a breach which by its nature cannot be cured;
- (vi) by Parent, if the Board of Directors of Seller shall withhold, withdraw, change or otherwise modify in a manner adverse to Parent its unanimous recommendation that the stockholders of Seller adopt this Agreement and approve the transactions contemplated herein, or if the Seller shall fail to permit such unanimous recommendation to be included in any communications to Seller's stockholders; or

(vii) by Seller, if (i) neither Seller nor the Principal Stockholder is in material breach of its obligations under this Agreement and (ii) there has been a breach of any representation, warranty, covenant or agreement contained in this Agreement such that the conditions set forth in Section 8.1(b) would not be satisfied and such breach has not been cured within ten (10) business days after written notice thereof to Parent; *provided, however*, that no cure period shall be required for a breach which by its nature cannot be cured.

(b) Notwithstanding the above, a party shall not be allowed to exercise any right of termination pursuant to Section 10.1(a) if the event giving rise to the termination right shall be due to such party's breach of one or more of its representations or warranties in this Agreement or the failure of such party to perform or observe in any material respect any of the covenants or agreements to be performed or observed by such party pursuant to this Agreement. In the event this Agreement is terminated in accordance with Section 10.1(a), no party shall have any further liability hereunder, except for willful breach of this Agreement.

ARTICLE XI. MISCELLANEOUS

11.1 <u>Assignment</u>. Neither this Agreement nor any of the rights or obligations hereunder may be assigned by the Seller without the prior written consent of the Purchaser. Prior to the Closing, the Purchaser may assign its rights hereunder to an Affiliate. After the Closing, the Purchaser may assign its rights hereunder to a successor to the equity interests of the Purchaser or all or substantially all of the assets of the Purchaser. Subject to the foregoing, this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective permitted successors and assigns, and no other Person shall have any right, benefit or obligation hereunder.

11.2 Notices. Unless otherwise provided herein, any notice, request, instruction or other document to be given hereunder by any party to the other shall be in writing and delivered in person or by courier, telegraphed, telexed or by facsimile transmission, sent by a nationally recognized overnight delivery service or mailed by registered or certified mail, postage prepaid, return receipt requested (such mailed notice to be effective on the date of such receipt is acknowledged), as follows:

If to Parent, DIS or Purchaser:

Digirad Corporation 13950 Stowe Drive Poway, California Attention: Chief Executive Officer Fax: (858) 726-1700

With a copy (which shall not constitute notice) to:

Wilson Sonsini Goodrich & Rosati, Professional Corporation 12235 El Camino Real, Suite 200 San Diego, California 92130 Attention: Martin J. Waters, Esq.

Fax: (858) 350-2399

If to the Seller or Selling Stockholder:

Ultrascan, Inc. 1048 Industrial Court, Suite E Suwanee, Georgia 30024 Attention: Chief Executive Officer

Fax: (770) 813-0326

With a copy (which shall not constitute notice) to Seller's Counsel:

Troutman Sanders LLP. 600 Peachtree St., N.E., Suite 5200 Atlanta, Georgia 30308 Attention: Thomas W. Baker, Esq.

Fax: (404) 962-6505

Any party may, from time to time, designate any other address to which any such notice to it or such party shall be sent. Any such notice shall be deemed to have been delivered upon receipt.

11.3 <u>Choice of Law</u>. This Agreement shall be governed in all respects by the laws of the State of Delaware without regard to conflict of law principles that would result in the application of any law other than the law of the State of Delaware.

11.4 Entire Agreement; Amendments and Waivers; Interpretation. This Agreement, the Ancillary Agreements and all exhibits and schedules hereto and thereto and the Confidentiality Agreement constitute the entire agreement among the parties pertaining to the subject matter hereof and supersede all prior agreements, understandings, negotiations and discussions, whether oral or written, of the parties. No supplement, modification or waiver of this Agreement shall be binding unless executed in writing by the party to be bound thereby. No waiver of any of the provisions of this Agreement shall be deemed or shall constitute a waiver of any other provision hereof (whether or not similar), nor shall such waiver constitute a continuing waiver unless otherwise expressly provided. No failure or delay on the part of any party hereto to exercise any right or remedy under this Agreement shall operate as a waiver of such right or remedy, and no single or partial exercise of any such right or remedy shall preclude any other or further exercise thereof. No party shall be deemed to have waived any claim arising out of this Agreement, or any right or remedy under this Agreement, unless the waiver of such claim, right or remedy is expressly set forth in a written instrument duly executed and delivered on behalf of such party. The parties have participated jointly in the negotiation and drafting of this Agreement. In the event an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as if drafted jointly by the parties and no presumption or burden or proof shall arise favoring or disfavoring any party by virtue of the authorship of any of the provisions of this Agreement. For all

purposes of and under this Agreement, (a) the word "including" shall be deemed to be immediately followed by the words "without limitation;" (b) words (including defined terms) in the singular shall be deemed to include the plural and vice versa; (c) words of one gender shall be deemed to include the other gender as the context requires and (d) the terms "hereof," "herein," "hereto," "herewith" and any other words of similar import shall, unless otherwise stated, be construed to refer to this Agreement as a whole (including all of the schedules and exhibits to this Agreement) and not to any particular term or provision of this Agreement, unless otherwise specified.

- 11.5 <u>Counterparts</u>. This Agreement may be executed in one or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.
- 11.6 <u>Severability</u>. In the event that any one or more of the provisions contained in this Agreement or in any other instrument referred to herein, shall, for any reason, be held to be invalid, illegal or unenforceable in any respect, such invalidity, illegality or unenforceability shall not affect any other provision of this Agreement or any other such instrument.
- 11.7 <u>Headings</u>. The headings of the Articles and Sections herein are inserted for convenience of reference only and are not intended to be a part of or to affect the meaning or interpretation of this Agreement.
- 11.8 Expenses. Except as otherwise specifically provided in this Agreement, each party will pay its own expenses incident to this Agreement and the transactions contemplated hereby, including legal and accounting fees and disbursements.
- 11.9 <u>Schedules</u>. The Schedules and Exhibits referenced in this Agreement are a material part hereof and shall be treated as if fully incorporated into the body of the Agreement.
- 11.10 <u>Publicity</u>. Except as required by law, neither the Seller nor the Purchaser shall issue any press release or make any public statement regarding the transactions contemplated hereby without the prior written approval of the other party.
- 11.11 No Third-Party Beneficiaries. This Agreement is for the sole benefit of the parties hereto (and their permitted successors and assigns), and nothing herein expressed or implied shall give, or be construed to give, to any Person, other than the parties hereto and such permitted successors and assigns, any legal or equitable rights hereunder.
- 11.12 Remedies Cumulative; Specific Performance. The rights and remedies of the parties hereto shall be cumulative (and not alternative). Each party acknowledges and agrees that the other party would be damaged irreparably in the event any of the provisions of this Agreement was not performed in accordance with its specific terms or was otherwise breached. Accordingly, each party agrees that the other party shall be entitled to an injunction or injunctions to prevent breaches of the provisions of this Agreement and to enforce specifically this Agreement and the terms and provisions hereof in any action or proceeding in addition to any other remedy to which it may be entitled, at law or in equity.

11.13 Further Assurances.

- (a) Upon the terms and subject to the conditions contained herein, after the Closing the parties agree (i) to use all reasonable efforts to take, or cause to be taken, all actions and to do, or cause to be done, all things necessary, proper or advisable to consummate and make effective the transactions contemplated by this Agreement and the Ancillary Agreements; (ii) to execute any documents, instruments or conveyances of any kind which may be reasonably necessary or advisable to carry out any of the transactions contemplated hereunder and thereunder; and (iii) to cooperate with each other in connection with the foregoing.
- (b) After the Closing, each party agrees that it will cooperate with and make available to the other party, during normal business hours, all books and records, information and employees (without substantial disruption of employment) retained and remaining in existence after the Closing which are necessary or useful in connection with any financial statement audit, tax inquiry, audit, investigation or dispute, any litigation or investigation or any other matter requiring any such books and records, information or employees for any reasonable business purpose and will take reasonable measures to cause it Representatives, including its accountants, to do the same.
- (c) If, for any reason after the Closing, any payment is received by the Seller, to the extent that any portion of said payment is in satisfaction of an obligation that is, in accordance with the terms of this Agreement, properly payable to the Purchaser, then the Seller shall promptly reimburse the Purchaser for or promptly remit to the Purchaser, as the case may be, such payment.

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed on their respective behalf all as of the day and year first above written.

ULTRASCAN, INC.

a Georgia corporation

/s/ William "Mickey" King, Jr.

By: William "Mickey" King, Jr. Title: President and CEO

WILLIAM "MICKEY" KING, JR.

/s/ William "Mickey" King, Jr.

DIGIRAD ULTRASCAN SOLUTIONS, INC.

a Delaware corporation

/s/ Todd Clyde

By: Todd Clyde
Title: CFO, President

DIGIRAD IMAGING SOLUTIONS, INC.

a Delaware corporation

/s/ Todd Clyde

By: Todd Clyde Title: CFO

DIGIRAD CORPORATION

a Delaware corporation

/s/ Mark Casner

By: Mark Casner

Title: Chief Executive Officer

SIGNATURE PAGE TO ASSET PURCHASE AGREEMENT

CERTIFICATION OF PRINCIPAL EXECUTIVE OFFICER PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, Mark L. Casner, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of Digirad Corporation;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting;
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

May 7, 2007

/s/ MARK L. CASNER

Mark L. Casner President and Chief Executive Officer (Principal Executive Officer)

CERTIFICATION OF PRINCIPAL FINANCIAL OFFICER PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, Todd P. Clyde, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of Digirad Corporation;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting;
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

May 7, 2007

/s/ TODD P. CLYDE

Todd P. Clyde Chief Financial Officer (Principal Financial Officer)

CERTIFICATION OF PRINCIPAL EXECUTIVE OFFICER PURSUANT TO 18 U.S.C. SECTION 1350 (SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002)

In connection with the accompanying Quarterly Report on Form 10-Q of Digirad Corporation for the period ended March 31, 2007, I, Mark L. Casner, Chief Executive Officer of Digirad Corporation, hereby certify pursuant to 18 U.S.C. § 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that to the best of my knowledge:

- (1) such Quarterly Report on Form 10-Q of Digirad Corporation for the period ended March 31, 2007, fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934, as amended; and
- (2) the information contained in such Quarterly Report on Form 10-Q of Digirad Corporation for the period ended March 31, 2007, fairly presents, in all material respects, the financial condition and results of operations of Digirad Corporation at the dates and for the periods indicated.

This certification has not been, and shall not be deemed, "filed" with the Securities and Exchange Commission.

May 7, 2007

/s/ MARK L. CASNER

Mark L. Casner

President and Chief Executive Officer
(Principal Executive Officer)

A signed copy of this written statement required by Section 906 has been provided to Digirad Corporation and will be retained by Digirad Corporation and furnished to the Securities and Exchange Commission or its staff upon request.

CERTIFICATION OF PRINCIPAL FINANCIAL OFFICER PURSUANT TO 18 U.S.C. SECTION 1350 (SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002)

In connection with the accompanying Quarterly Report on Form 10-Q of Digirad Corporation for the period ended March 31, 2007, I, Todd P. Clyde, Chief Financial Officer of Digirad Corporation, hereby certify pursuant to 18 U.S.C. § 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that to the best of my knowledge:

- (1) such Quarterly Report on Form 10-Q of Digirad Corporation for the period ended March 31, 2007, fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934, as amended; and
- (2) the information contained in such Quarterly Report on Form 10-Q of Digirad Corporation for the period ended March 31, 2007, fairly presents, in all material respects, the financial condition and results of operations of Digirad Corporation at the dates and for the periods indicated.

This certification has not been, and shall not be deemed, "filed" with the Securities and Exchange Commission.

May 7, 2007

/s/ TODD P. CLYDE

Todd P. Clyde Chief Financial Officer (Principal Financial Officer)

A signed copy of this written statement required by Section 906 has been provided to Digirad Corporation and will be retained by Digirad Corporation and furnished to the Securities and Exchange Commission or its staff upon request.